Copyright

by

Julia Casey McCreary

2012

The Treatise Committee for Julia Casey McCreary certifies that this is the approved version of the following treatise:

Teachers' Perceptions of the Hiring Process in Texas Public Schools: Information Richness, Position Fit, and Intentions to Remain in the Classroom

Committee:

Ruben Olivarez, Supervisor

Marilyn Kameen

Sherry Field

Susan Holley

Kerry Moll

Teachers' Perceptions of the Hiring Process in Texas Public Schools: Information Richness, Position Fit, and Intentions to Remain in the Classroom

by

Julia Casey McCreary, B.S. Ed.; M.S.

Treatise

Presented to the Faculty of the Graduate School of

The University of Texas at Austin

in Partial Fulfillment

of the Requirements

for the Degree of

Doctor of Education

The University of Texas at Austin May 2012

Dedication

This paper is dedicated to my sister Keith McCreary Donnelly (1968–1998) whose tenacity and joie de vivre continue to inspire me. I know that she would be proud.

Acknowledgements

The Cooperative Superintendency Program at the University of Texas at Austin

My journey began in 2005 when Mr. David D. Anderson, Dr. Paula Moeller, and Dr. Bobby Ott nominated me to serve as a candidate for the prestigious University of Texas Cooperative Superintendency Program (CSP). I greatly appreciate their faith in me. I was further honored when Dr. Nolan Estes invited me to join the CSP Cohort XVI with nine others. Dr. Estes provided profound leadership during my nascent time in the program, as did Dr. Ruben Olivarez during my last years as a member of the program.

Thanks to the collaborative efforts of my Cohort XVI colleagues (referred to as "The Family" by many of our professors), we made it through the 60 plus hours of course work while working full time in our various careers. Although it was exhausting, we maintained our sense of humor and established valuable friendships, both professionally and personally. Kudos to my fellow cohort members for whom I have the upmost respect, including Jonas Chartock, Dr. Barbara Christenson, Dr. TJ Dilworth, Mark Gesch, Michael Greenwalt, Dr. Ann O'Doherty, Gene Solis, Chris Ulcek, and Dr. Michelle Ungrait.

A special thank you to Dr. Ruben Olivarez for agreeing to chair my committee and to help me navigate towards the completion of this paper. I appreciate my other committee members who were so generous with their time on my behalf including Dr. Sherry Field, Dr. Susan Holley, Dr. Marilyn Kameen, and Dr. Kerry Moll. As a committee member, work colleague, and liaison with my grant program, Dr. Holley deserves triple accolades for providing support through multiple avenues. She is a dear friend, colleague, and mentor. I greatly appreciate both the Graduate School Advisor, Dr. Martha Ovando and Graduate School Coordinator, Ms. Hortensia Palomares, whose open-door policies welcomed me to visit or contact them at any time. Their wealth of information about the path toward graduation was invaluable.

Another note of thanks to CSP graduate Dr. Esthela Rodriguez whose wisdom and advice I appreciate, and who introduced me to Jessica Doss, my editor extraordinaire.

Texas Association of School Administrators (TASA)

I could not have finished the program without the continued offers of reassurance from TASA Executive Director, Dr. Johnny Veselka (also a CSP graduate), that during my tenure at TASA my efforts to complete the program coursework and the treatise would be supported. Dr. Veselka, as expected, was true to his word. I have the upmost respect for him as an individual and leader in the field of school administration. In addition, TASA Associate Executive Director, Amy Beneski offered moral support whenever I felt overwhelmed, and provided work flexibility whenever University deadlines were on the horizon. Thank you to Albert Rivas, Jennifer Garrido, and all my other TASA colleagues who inquired as to my progress and offered words of encouragement along the way. I want to give an additional nod to TASA superintendents Dr. Michael Hinojosa, and Dr. Reece Blincoe for their continuous support and motivation.

Center for Research, Evaluation, and Advancement of Teacher Education (CREATE)

I extend gratitude to Dr. Bill Reaves for recruiting, and then guiding me through the initial stages as a research scholar with the CREATE Texas Public School Research Network, and for introducing me to CREATE Executive Director Dr. Mona Wineburg. A special note of thanks to Sherri Lowry for the never-ending questions she answered and the constant encouragement that she offered; to Dr. Dale Johnson who never hesitated in sharing textbooks from his personal library or in providing valuable statistical guidance with a twinkle in his eye. My scholarship with CREATE, also provided me an opportunity to work with Dr. Ed. Fuller and I appreciate his expertise and interest in the focus of this paper.

Research by Dr. Susan Moore-Johnson and Dr. Edward Liu caught my eye and the eyes of my CREATE colleagues. Several research projects, including this one were initiated after they agreed to permit us to use the teacher and principal surveys they had developed. Many thanks to Dr. Johnson and Dr. Liu for sharing their work, thus expanding the research related to teacher-hiring systems.

Family and Friends

The most important acknowledgement goes to my parents, David and Binnie Hoffman and Lou and Patsy McCreary who collectively taught me the value of hard work and the importance of education. Their unwavering faith and pride gave me the drive to persevere towards the ultimate goal of graduation. I love them all so very much. To my sisters Shay and Meghan, thanks for the sisterly love!

Thank you to my Dallas friends, the "Ladybugs" (Karen Fields Evans, Lisa Long, and Shaleen Margolese), for their unwavering support throughout the completion of my Master's Degree program and this Doctoral program. Thank you to my Austin friends (Kathy Brown, Laura Novy, Andy Oliver, ToniKay Osgood, Jamie Patterson, and Charlotte Wallis) for helping me to maintain at least a small degree of sanity during my overbooked work and school schedules for the last six years. And a last shout out of gratitude goes to the Legislative Ladies Chapter of the Ladies Auxiliary Club (you know who you are)!

Teachers' Perceptions of the Hiring Process in Texas Public Schools: Information Richness, Position Fit, and Intentions to Remain in the Classroom

Julia Casey McCreary, Ed.D. The University of Texas at Austin, 2012

Supervisor: Ruben Olivarez

Abstract

The United States Department of Education issued a blueprint in 2010 outlining intended changes for the Elementary and Secondary Education Act. Included was a focus on selecting *effective* teachers per new criteria. Information about teacher selection and assignment systems as related to teacher retention is beneficial to school district leadership as they prepare for the new federal expectations. This study extends previous research by Liu and Johnson (2006) that examined the experiences of newly hired teachers and introduced the construct of an information-rich hiring process, in which a district hiring process provides both the teacher applicant and district employer with sufficient interaction and adequate information-exchange to make informed decisions leading to a position fit of teacher to teaching position and campus. In addition to the use of an information-rich hiring process as a hypothetical construct, other theories incorporated in this research include: realistic job preview theory; human resource management theory; person-job-fit, person-organization-fit, and person-group-fit

theories; and two-sided matching theory. The problem addressed in this study: The selection and assignment of teachers is often done in complex systems leading to poor matches that culminate in job dissatisfaction and teachers' intentions to leave the classroom. The purpose of this study was to examine how newly hired teachers perceived their hiring experience and their *fit* with their campus and classroom assignments in order to determine if these perceptions predicted their intentions to remain in the classroom. This study used a nonexperimental approach with an ex-post facto design and a quantitative methodology to examine associations between variables. Participants in the study included 1,430 newly hired teachers at 92 campuses located across 13 Texas school districts who were administered an electronic survey instrument.

Keywords: teacher hiring, effective teachers, job fit, job satisfaction

Table of Contents

List of Tablesxvi
Chapter One
Contexts that Affect the Hiring Environment1
A Brief Review of Teacher Selection and Assignment Systems4
Policy and organizational factors that affect teacher-hiring practices
Teacher recruitment
How teachers are selected and assigned to campus positions7
Obstacles to urban school hiring systems
Teachers' intentions to remain in the classroom10
Conceptual Framework
Statement of the Problem
Purpose of the Study14
Research Questions
Methodology16
Significance17
Definition of Terms
Delimitations
Limitations
Assumptions
Chapter Summary21

Organization of the Study	21
Chapter Two: Literature Review	23
Methodology of the Literature Review	25
Policy and Organizational Factors that Affect Teacher Hiring Practices	26
Teacher policy reform	27
Affects of decentralized and centralized hiring environments on teacher selection	29
Hiring delays as obstacles	33
Recruitment of Teachers	34
Supply and demand	34
Teacher recruitment theories	
21st century teacher recruitment methods needed	37
Teacher Selection and Assignment Systems	
Avoiding Selection bias	
Purpose and objectives of teacher selection	42
Selection as a process, not an event	43
Procedural stages of selection	44
Foundation of the selection process	44
Selection tools	45
Hiring tools in the screening stage	46
Hiring tools in the selection or employment stage	48
Hiring theories and frameworks	49

Human resource management theory	
Person-job fit, person-organization fit, person-group fit theory	50
Two-sided matching	
Information-rich hiring process	53
Principals' role in selecting effective teachers	
Selection for 21st century learning	56
Principals' preferences	57
Urban Schools and Obstacles to an Information-Rich Hiring I	Process60
Urban recruitment challenges	60
Late hiring timelines and other hurdles	61
Teachers of urban campuses	61
Obstacles to staffing effective teachers at urban schoo	ls62
Unequal distribution of novice teachers	64
Selection of teachers from traditional versus alternative paths to certification	65
Strategies for improvement of the urban teacher work:	force66
Teacher Turnover: Costly	67
Teachers' Intentions to Remain in the Classroom	69
Good matches lead to job satisfaction and intentions to remain	69
Positive work outcomes lead to job satisfaction and intentions to remain	70
Conclusions	72

Chapter Summary	76
Chapter Three: Method	78
Conceptual Framework, Research Questions, and Null Hypotheses	
Research Design	79
Variables	80
Description of Population and Sample	80
Procedures for Data Collection	83
Instrument	85
Data Analysis	
Chapter Summary	90
Chapter Four: Analysis of Data	91
Regression Analysis	
Results for hypothesis 1	92
Results for hypothesis 2	93
Descriptive Statistics	94
Chapter Five: Discussion, Conclusions, and Recommendations for Future Research	100
Summary of Results	100
Discussion and Conclusions	101
Limitations	103
Significance and Recommendations for Practice	104
Recommendations for Further Inquiry	105

Appendix A: InTASC Core Teaching Standards	108
Appendix B: Teacher Survey	125
Appendix C: IRB Approval for CREATE/TASA Employment Study	160
Appendix D: CREATE Permission Letter	161
Appendix E: Exemption Status Approval	162
References	165

List of Tables

Table 1: TEA Snapshot of School District Community Type	.83
Table 2: Logistic Regression: Information-Rich Practices asPredictor of Teachers' Intentions to Remain in the Classroom	93
Table 3: Logistic Regression: Position Fit as Predictor ofTeachers' Intentions to Remain in the Classroom	94
Table 4: Future Intentions of Classroom Teachers	.95
Table 5: Degree of Information-Rich Hiring Practices	.96
Table 6: Degree of Position Fit	99

Chapter One

There is a common assumption that teachers are interchangeable parts, referred to as the *widget effect* by Weisberg, Sexton, Mulhern, and Keeling (2009). This perspective views every teacher in a uniform manner without giving consideration to the fact that teachers as individuals have unique attributes, skills, experiences, and work expectations that may or may not make them a good fit for a particular teaching assignment. However, Weisberg et al. determined that the selection and assignment of teachers who support the academic achievement of their students involves more critical decisionmaking than simply ensuring that an open position is filled with any available teacher. The process of hiring teachers who are a good fit with classroom and campus assignments is influenced by numerous contexts. This introduction includes the various contexts that affect the hiring environments of Texas teachers. This study examined recently hired Texas teachers' perceptions of their hiring experiences and their satisfaction with their classroom and campus assignments in an attempt to identify relationships between teachers' hiring experiences, satisfaction with their employment decisions, and their subsequent attitudes towards remaining in the classroom.

Contexts that Affect the Hiring Environment

The context of federal policy centered on high stakes accountability brings into sharp focus the pivotal role that a teacher plays in the system. The current requirements of the *No Child Left Behind Act of 2001* (NCLB) call for a *highly qualified* teacher in every classroom and dictate the certifications needed for each teaching position (2002). However, the United States Department of Education (USDE) recently issued a blueprint outlining changes intended for the next authorization of the *Elementary and Secondary Education Act of 1965* (ESEA) (2010). The blueprint indicates a new approach that will shift the focus from highly qualified to *highly effective* teachers, foreshadowing a potential increase in pressure for school district hiring systems because teacher and principal evaluations would be directly linked to the academic achievement of their students as demonstrated by growth on standardized test scores. Another goal outlined in the USDE blueprint is geared toward ensuring a more equitable distribution of effective teachers at high-need schools, a chronic problem in public school selection and assignment systems. Although the blueprint offers one avenue of future support with mention of proposed innovation grants that could be used to improve efficiency of hiring systems through the establishment of earlier hiring timelines (2010, p. 16), other strategies that address staffing inequities are not included; thus it remains unclear how the chronic problem of inequitable distribution will be specifically addressed.

The context of state level policy designates responsibility in teacher selection and assignment systems. Texas Education Code (TEC) (1995a) mandates that a *campus planning and site-based decision-making committee* assist the principal in decisions regarding staffing patterns that support the goals of the *campus improvement plan*. TEC (1995b) also authorizes the principal to approve teacher assignments by selecting from a pool of applicants who meet district and campus hiring criteria. The enactment of these statutes demonstrates a movement towards decentralizing the teacher hiring process in Texas public school environments. Although the state gives much hiring responsibility to principals, there is a lack of professional development available to inform the decision

2

making of principals hiring teachers deemed effective. This may be an issue at the state level in part because there is no federal definition or professional consensus of what constitutes an "effective" teacher at this time. At the state level in Texas there are minimum standards outlined in the Texas Administrative Code (TAC)(1999) for teachers of different grade levels and subject areas, which is at least a starting point for principals and others making hiring decisions.

Another important context of the teacher-hiring environment in Texas is the dismal state of school finance at federal and state levels. According to a report by Anderson (2010) the federal economic crisis continues with more cuts anticipated for fiscal year 2011. In addition, the USDE blueprint (2010) indicates the system for distributing federal grant monies to public schools will change from formula to competitive grant structures, a move anticipated to increase financial instability at the state level. A last impairment to the state of school finance was experienced when the 82nd Texas Legislature convened in 2011, as the state faced a \$27 billion shortfall that lead to \$5.3 billion in cuts to education allocations that support teacher salaries and other instructional resources (House Bill 1, 2011).

In summary, while policy demands school systems to hire effective teachers and increases expectations for student achievement, the funding and resources needed to attract and retain quality teachers is decreasing. These conflicting actions are particularly troublesome for high-need schools that already experience staffing hardships. Superintendents, school board members, human resource personnel, and principals face

3

challenges in making decisions related to investments in staffing and in maintaining programs despite increasing budget constraints.

The following section briefly reviews previous findings related to teacher selection and assignment systems and introduces the conceptual framework involving *information-rich hiring processes*; satisfactory *position fit* of teacher to assignment, and teacher to campus: and associations between the these two constructs and teachers' intentions to remain in the classroom.

A Brief Review of Teacher Selection and Assignment Systems

Historically, policies that evolved through statute, such as the NCLB Act, codify the important role a teacher plays in student achievement, thus making relevant the examination of how teachers are selected. In addition to policy factors, other relevant components of the teacher selection landscape include recruitment, hiring tools, hiring theories, and system challenges. This review also addresses the role that the teacherhiring process may have on teachers' decisions to remain in the classroom.

Policy and organizational factors that affect teacher-hiring practices. A brief review of policy and organizational factors in the current system demonstrates that the selection and assignment of teachers is an important responsibility for school systems as organizations (Bolton, 1969). The National Commission on Teaching and America's Future (NCTAF) advocated that a teacher's content knowledge and pedagogical skills should be the determining factors of quality (1996). Despite the advocacy of NCTAF there are no uniform guidelines to help a principal or other teacher employer assess a teacher's content knowledge. NCLB policy (2002) focuses on improving teacher quality

at the local level with a call for a highly qualified teacher for every student. NCLB mandates that *teachers of record* who provide direct instruction in core academic areas must: (a) hold at least a bachelor's degree, (b) be fully state-certified to teach, and (c) demonstrate competency in their core subject area (2002). The USDE blueprint (2010) indicates the current teacher certification requirements will be maintained but with more flexibility to allow for a focus on highly effective teachers, per new criteria, which has not yet been explicitly defined. The blueprint also indicates that an equitable distribution of effective teachers to high-need schools is a priority.

Indicated by the literature, policy that steers environments towards either decentralized or centralized practices affects hiring systems. Many researchers found that teacher selection and assignment systems move back and forth between each end of a spectrum of centralized and decentralized practices (Cooper & Fusarelli, 2004; Hanson, 2003; Hannaway & Stanislawski, 2005; Heck, 2004; Schwartz, 2010), but that many districts often implement a mixture of centralized and decentralized practices (Hannaway & Stanislawski, 2005; Hanson, 2003; Liu, 2002; Wise, Darling-Hammond, Barnett, Berliner, Haller, Praskac, & Schlechty, 1987). Young and Miller-Smith (2006) looked at districts that had legislatively prescribed, site-based councils as part of the teacher hiring process (similar to Texas statutory requirements) but did not find significant benefits resulting from the participation of site-based councils. Other researchers determined that conclusive findings linking decentralized hiring practices to student achievement continue to be elusive (Hannaway, & Stanislawski, 2005; Hanson, 2003), although others

5

found benefits to decentralized practices for nonpoor schools (Galiani, Gertler, & Schargrodsky, 2008).

Also affecting teacher-hiring environments, reviews of school district policies indicated that district selection and assignment systems that permit late hiring timelines experience an attrition of stronger teacher candidates, resulting in a supply pool of less qualified teachers (Castetter & Young, 2000; Darling-Hammond; 2001; Levin & Quinn, 2003; Levin et al., 2005; Liu & Johnson, 2006; The New Teacher Project, 2007; Young, 2008). These same scholars found that school principals in these districts often rush to fill vacancies before the school year begins, leaving little time for the hiring process.

Teacher recruitment. Many scholars showed that recruitment is an essential part of the teacher selection process and the first step toward attracting quality individuals to the teaching field, a school district, or a campus (Castetter & Young, 2008; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008). Guarino, Santibañez, and Daley (2006) applied an *economic labor theory of supply and demand* to the teaching market and found that districts could manipulate frameworks to focus on quality versus quantity to attract the best of teacher applicants who have attributes (pre-established during recruitment) associated with student achievement. Researchers found that teachers who have certain qualities, such as strong verbal ability, could positively influence lower achieving students as measured by composite achievement (Coleman, 1966) and standardized test scores (Ferguson, 1991). Another recruitment technique based on *realistic job preview theory* ensures teaching applicants have an accurate understanding of an assignment before accepting a job offer and therefore are more likely to experience an information-rich hiring process and job satisfaction (Castetter & Young, 2000; Clement, 2008; Hays & Behrstock, 2009; Liu & Johnson, 2006; Seyforth, 2002).

Other studies examined school district recruitment systems, and the researchers deemed systems outdated and in need of catching up to 21st Century practices in terms of acknowledgement of changes to the teacher supply pool (candidates more mobile and less likely to make teaching a life-long career) and in need of more aggressive approaches to compete with the private sector (Behrstock & Clifford, 2009; Hays & Behrstock, 2009; Hess, 2009; Johnson & Birkeland, 2003; Smith, 2009; Winter & Melloy, 2005; Young, 2008). Other scholars found that recruitment systems need to incorporate strategies for attracting Generation Y candidates, those born between 1977 and 1995 who have the expertise and comfort level with technology needed to prepare students for 21st Century college and workforce environments (Behrstock & Clifford, 2009; McGraner, 2009).

How teachers are selected and assigned to campus positions. The idea that collaborative efforts ensure a greater selection of teachers considered to be a good fit for an assignment led to decentralization of staffing decisions, and campus-based decisionmaking teams replaced central office decision making (Smith, 2009). This movement to campus-based decision making requires that principals and site-based team members be aware of laws and regulations designed to prevent selection bias (Castetter & Young, 2000; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008). These researchers agreed that selection is a process that demands informed decision making on behalf of employers and applicants. Furthermore, hiring decisions involve hiring tools in various stages of the process, such as written transcripts in the screening stage and interviews in the employment stage; as opportunities for campuses to make informed decisions increase, the information richness of the hiring experience increases, and as a result both the applicant and employer(s) benefit (Castetter & Young, 2000; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008).

Many researchers showed that the campus principal in the role of steward is responsible for carefully selecting teacher applicants to ensure that each teacher assigned to a campus position has the qualities and attributes needed to promote the success of the particular student population (The Council of Chief State School Officers (CCSSO), 2008; Smith, 2009; Sorenson & Goldsmith, 2009). Principals play a prominent role in the teacher-selection process; however, they prioritize differently when making hiring decisions and have different preferences for teachers with particular attributes (Harris, Rutledge, Ingle, & Thompson, 2007; Rutledge, Harris, Thompson, & Ingle, 2007; Trimble, 2001). Some researchers recommend that principals be trained to identify and select teachers that can incorporate new education paradigms into their instructional practices and thus equip students with the more advanced 21st century technology skills (Berhstock & Clifford, 2009; Hill, 2009).

Three hiring theories in human resource literature apply to teacher selection and assignment systems: *human resource management theory; person-job-fit, person-organization-fit, person-group-fit theories*; and *two-sided matching theory*. Each of these theories involves a two-way exchange of information as well as choices made by individuals on both sides of the hiring process (applicant and employer). During a study on the perspectives of recently hired teachers, Liu and Johnson (2006) introduced another

two-way theory that describes an *information-rich hiring process* as one in which teacher applicants and those making the hiring decisions have maximum opportunity to exchange information and interact. On the opposite end of the hiring spectrum, Liu and Johnson posited that an *information-poor hiring process* lacks opportunities for both sides to exchange information, and is often associated with a teacher's ultimate dissatisfaction with an assigned position. The two researchers introduced the term *position fit* as a satisfactory fit between teacher and classroom assignment as well as teacher and campus assignment (Liu & Johnson, 2006).

Obstacles to urban school hiring systems. Many scholars conducted research that focused on the unique challenges faced by urban districts often found to have hard-to-staff campuses (Levin & Quinn, 2003; Levin et al., 2005; Liu, 2007; Strauss, Bowes, Marks, & Plesko, 2000). Boyd et al. (2005, 2006) and Darling-Hammond and Prince (2007) found that obstacles urban schools face during teacher selection and assignment are related to recruitment systems that offer less attractive enticements than their neighboring suburban districts. Other obstacles to urban school hiring systems result from thick layers of bureaucracy that lead to the loss of quality applicants because of late hiring timelines and allow novice teachers to be assigned to the most difficult teaching assignments. Because of these types of obstacles, urban schools have a disadvantage in hiring the most effective of teachers, and as a result students who most need quality teachers may not have access to them (Claycomb, 2000; Darling-Hammond & Prince, 2007; Jacob, 2007; Lankford, Loeb, & Wyckoff, 2002).

9

Solutions recommended by scholars to enrich urban hiring systems include strategies such as: (a) establishing more streamlined district processes, (b) forming partnerships between districts and neighboring universities, (c) considering alternate pathways to certification, and (d) providing targeted training by teacher preparation programs to individuals likely to work in urban school environments (Claycomb, 2000; Darling-Hammond & Prince, 2007; Jacob, 2007; McGraner, 2009). The New Teacher Project (2003) recommended four broad actions to improve urban hiring. The solutions include: (a) require teachers to give earlier vacancy notifications, (b) expedite interdistrict transfer processes, (c) promote earlier and more predictive budget processes, especially for the harder to staff schools, and (d) streamline human resource department processes and increase the role in hiring at the campus level.

Teachers' intentions to remain in the classroom. Research related to teacher turnover concludes that high turnover is expensive, and it is worthwhile to examine methods or procedures that are proactive in preventing attrition (Behrstock & Clifford, 2009; Guarino, Santibañez, & Daley, 2006; Miller & Chait, 2008; National Commission on Teaching and America's Future [NCTAF], 2007; Perrachione, Rosser, & Peterson, 2008). Scholars also examined methods that may positively impact teachers' intentions to remain in the classroom because campuses with high turnover rates often experience a loss of high quality teachers, especially detrimental to schools that are already struggling (Behrstock & Clifford, 2009; Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Papa & Baxter, 2008; Perrachione et al., 2008). Some theories related to the prevention of high teacher turnover identify that compatibility between employee and job is an important factor in an employee's job satisfaction (Bretz & Judge, 1994; Dugonni & Ilgen, 1981; Liu, 2002; Liu & Johnson, 2006; McGraner, 2009; O'Reilly, Chatman, & Caldwell, 1991; Perrachione et al., 2008), and job satisfaction is a major influence on teachers' decisions to remain in the classroom (Cohn, 1992; Johnson & Birkeland, 2003; Goodlad, 1984; Lee & Mowday, 1987; Lortie, 1975; Meek, 1998; Murnane, 1991; Perrachine et al., 2008). Hiring practices and systems play a role in matching employee to job, and the more thorough the teacher selection and assignment system, the more likely teachers will experience job satisfaction and thus remain in their positions (Bretz & Judge, 1994; Dugonni & Ilgen, 1981; Liu, 2002; Liu & Johnson, 2006; McGraner, 2009; Miller & Chait, 2008; NCTAF, 2007).

Another factor that leads to job satisfaction and intention to remain includes an employee's ability to cope with the demands of his or her job; some scholars found the use of *realistic job preview* to positively influence satisfaction and retention by improving teachers' job-coping capacity (Dugonni & Ilgen, 1981; Liu & Johnson, 2006; McGraner, 2009). Other scholars found links between person-organization fit (wherein the employee and organization he or she is working for are compatible) and positive work outcomes such as retention (Bowman, 2005; Bretz & Judge, 1994; Cable & Judge, 1996; Chatman, 1991; Judge, Higgins, & Cable, 2000; Liu, 2002; McGraner, 2009; O'Reilly et al., 1991; Rutledge et al., 2007). Likewise, person-job fit (in which there is compatibility between employee and his or her specific job assignment) positively affects work outcomes such as performance, satisfaction, and retention (Bowman, 2005; Bretz & outcomes and his or her specific job assignment) positively affects work

Judge, 1994; Cable & Judge, 1996; Chatman, 1991; Kristoff, 1996; McGraner, 2009; O'Reilly et al., 1991; Rutledge et al., 2007). A final established factor in job satisfaction and positive work outcomes includes an information-rich hiring process, wherein elements of preview and fit theories allow for extended exchanges of information between employee (teacher) and employer (Dugonni & Ilgen, 1981; Judge et al., 2000; Liu & Johnson, 2006; McGraner, 2009).

Conceptual Framework

This study incorporated Liu and Johnson's framework (2006) of an ideal teacherhiring environment as one that offers information-rich exchanges between an applicant and employer(s). This study extends use of that framework to examine relationships that may exist between information-rich hiring processes, position fit, and teachers' subsequent intentions to remain in the classroom.

When defining the nature of theory, Kerlinger stated, "A theory is a set of interrelated constructs (concepts), definitions, and propositions that present a systematic view of phenomena by specifying relationships among variables, with the purpose of explaining and predicting the phenomena" (2000, p. 11). The conceptual framework used for this study systemically involves two interrelated hypothetical constructs.

The first hypothetical construct is that an information-rich hiring process could predict teachers' intentions to remain in the classroom. Because this process allows for a mutual exchange of information between a teacher applicant and a principal (or others involved in campus-level hiring decisions), post-hire surprises at either end are minimized. An information-rich hiring process could lead to a campus staffed with teachers who could devote the majority of their time to academic responsibilities without the distraction of dissatisfaction related to their expectations. An information-rich hiring process may predict teachers' intentions to remain in the classroom.

The second hypothetical construct is that position fit may help to predict teachers' intentions to remain in the classroom. Teachers who experience compatibility with their teaching assignment (job) and campus (organization) may be more likely to remain in their teaching positions, leading to *job stability*, a longer period of time serving in the same or similar job assignment. A good position fit could provide stability that allows teachers the time and experience to develop and implement teaching strategies specific to the needs of their student population. As a result, a good position fit could increase teachers' perceptions of effectiveness with their students, increase job satisfaction, and therefore, may ultimately predict teachers' intentions to remain in the classroom.

Statement of the Problem

The problem addressed in this study: The selection and assignment of teachers is often done in complex systems leading to poor matches that culminate in job dissatisfaction and teachers' intentions to leave the classroom. According to the USDE blueprint goals, "School districts must also put in place policies to help ensure that principals are able to select and build a strong team of teachers with a shared vision and that teachers are choosing to be part of a school team" (2010, p. 16). This goal shows that the selection of teachers continues to involve critical decision making for stakeholders and supports well-planned district policy for selection and assignment systems. These systems should be able to (a) recruit, hire, and assign effective teachers,

(b) match teacher applicant with compatible teaching assignment, and (c) lay the foundation for teacher-job stability and improved teacher retention. Previous research demonstrates that an information-rich hiring process supports the appropriate pairing of teacher applicant to teaching assignment, but an information-poor hiring process, as Liu and Johnson show (2006), allows too little interaction and exchange of information between applicant and employer to ensure that the teacher will be a good fit for the assignment. Other research indicates that position fit is an important contributor to job satisfaction, and the more compatible the fit, the more likely an employee is to remain in his or her position (Bowman, 2005; Bretz & Judge, 1994; DeArmond, Shaw, & Wright, 2009; Judge et al., 2000; Liu & Johnson, 2006; Rutledge et al., 2007). Additional studies yield findings that a high rate of teacher turnover is costly to districts (NCTAF, 2007) and detrimental to the school environment (Behrstock & Clifford, 2009). Combined with increasing budget cuts to education (House Bill 1, 2011) and increasing federal policy demands (USDE, 2010), school districts are in need of effective, efficient hiring systems and practices that support the retention of highly effective teachers.

This study attempted to address the problem by examining if an information-rich hiring process and position fit outcomes predict teachers' intentions to remain in the classroom.

Purpose of the Study

The purpose of this study was to examine: (a) how newly hired teachers at Texas campuses perceived the information richness of their hiring experiences, (b) newly hired teachers' perceived fit with their assigned positions, and (c) newly hired teachers'

intentions to remain classroom teachers. This study intended to determine if relationships exist between an information-rich hiring process, position fit, and teachers' intentions to remain in the classroom. The attempt of this study was to provide new and timely information regarding the possible contributions an information-rich hiring process and position fit can make toward establishing a hiring environment that supports the retention of classroom teachers.

Literature from 1965 through 2010 revealed that much is at stake when hiring decisions are made about teachers (Behrtsock & Clifford, 2009; Bolton, 1965; Ferguson, 1991; Hill 2009; NCTAF, 1996; Sanders & Horn 1998; The Coleman Report, 1966; USDE, 2010; Wright, Horn, & Sanders, 1997), yet studies also found that school district hiring practices are inadequate, inefficient, and outdated (Ballou, 1996; Ballou & Podgursky, 1995, 1998; Darling-Hammond, 2001; Darling-Hammond & Prince 2007; Levin & Quinn, 2003; Levin et al., 2005; Liu, 2002; Liu & Johnson, 2006; Pfluam & Abramson, 1990; The New Teacher Project, 2007). Research studies show that particular obstacles exist for hard-to-staff urban schools during the selection process, indicating that students at these schools may not receive the benefit of a stabilized workforce of teachers who are capable of creating environments conducive to academic achievement (Boyd, Lankford, Loeb, & Wyckoff, 2005, 2006; Jacob, 2007; Levin, & Quinn, 2003; Levin et al., 2005; Liu, 2007; Papa & Baxter, 2008; Strauss et al., 2000). This study attempted to contribute to the literature on teacher selection and assignment systems by assessing whether the degree of information richness and position fit that a teacher experiences

15

when hired can be used as a predictor of the teacher's intention to remain in the classroom, thus providing increased campus stability.

Research Questions

While considering the hypothetical constructs described in the Conceptual Framework section of this chapter, two research questions emerged:

- 1. Does an information-rich hiring process predict teachers' intentions to remain in the classroom?
- 2. Does position fit predict teachers' intentions to remain in the classroom?

Methodology

This was a non-experimental study that used a quantitative methodology and a postpostitivist theoretical paradigm to examine associations between variables. This study design may also be referred to as *ex-post facto* because no experimental manipulations were performed. Kerlinger and Lee defined nonexperimental research as follows:

Nonexperimental research is a systematic empirical inquiry in which the scientist does not have direct control of independent variables because their manifestations have already occurred or because they are inherently not manipulable. Inferences about relations among variables are made, without direct interventions, from

The data this study analyzed was derived from a survey, previously administered to 1,430 of newly hired teachers at 92 campuses in 13 purposively selected Texas school districts. According to Mertens "If a purposeful sampling procedure will be used, the researcher

concomitant variation of independent and dependent variables. (2000, p. 558)

needs to provide sufficient details about the people in the study to communicate to the reader their important characteristics" (2005, p. 176). A detailed description of the population and sample was included in the methodology chapter in order to communicate the reasoning behind this purposively selected sample.

In this study, an electronic survey is the research instrument. A generalized linear model, logistic regression, is used to test the null hypotheses. For more detailed descriptions of the methodology, measures, design, and procedures, refer to Chapter Three.

Significance

New standards and changing expectations, as defined by the USDE in the ESEA blueprint, require administrators to modify their hiring systems in order to effectively recruit, select, and retain teachers who are deemed effective per this new criteria (USDE, 2010). These expectations exceed maintenance of the current teacher workforce and will necessitate significant changes. Simultaneously teacher and principal evaluations may become tied to academic performance of their students on standardized assessments through the proposed reauthorization of the ESEA. As these changes to federal education policy are being developed, finalized, and implemented, stakeholders stand to benefit from the contributions made by this and previous studies.

The findings in this study yield timely information accessible to law and policy makers as they develop statutes and rules to align with the new federal goals. This study also benefits school district leadership as it prepares for the new policy trends anticipated with the reauthorization of the ESEA, as well as school administrators as they work to increase teacher retention in order to improve campus stability and minimize teacher replacement costs. Knowledge of any links between teacher selection and assignment systems, hiring tools and methods, and teachers' intentions to remain in the classroom could influence district and campus administrators as they prepare to the meet the new federal expectations.

In addition, another aim of this study was to expand the understanding of the responsibility that human resource has in school leadership. The Interstate School Licensure Consortium Educational Leadership Policy Standards discussed the area of human resources and posed a question for policy makers, education leaders, and organizations, "How does a district or school evaluate the skills and dispositions of a candidate to improve student performance?" (CCSSO, 2008, p. 5). Within the same framework, school leaders are directed to "develop the instructional and leadership capacity of staff" and to "obtain, allocate, align, and efficiently utilize human, fiscal, and technological resources" (p. 14). The questions examined in this study align with the questions posed in the CCSSO document and provide new information for leadership to consider when selecting and assigning teachers.

Furthermore, the findings of this study can contribute significantly to the knowledge base of university superintendent-, principal-, and teacher-preparation programs, thus preparing future school district leadership to improve teacher selection and assignment systems, as well as preparing teacher candidates to successfully navigate current hiring environments and make informed decisions that will lead to job satisfaction and job stability.

Definition of Terms

Information-Rich Hiring Processes: refers to the degree to which district and campus hiring processes provide sufficient information and opportunities for both the job candidate and employer to gain knowledge of possible compatibility between applicant, position, and campus environment.

• Information-Rich Hiring Process = Teacher Survey Questions $\sum (12a-12j)$

Position Fit: refers to teacher and administrator perceptions of the compatibility of the teacher's professional skills and attributes to their current classroom assignment (job fit) and campus assignment (organization fit).

• Position Fit = Teacher Survey Questions $\sum (13a-13e \text{ and } 14a-14f)$

Intention to Remain in the Classroom: refers to a teacher's intention to remain a classroom teacher, as indicated by his or her response on the teacher survey.

Intent = Teacher Survey Question 10: I will (10a) most likely remain a classroom teacher, (10b) most likely leave classroom teaching, but stay in education (10c) most likely leave classroom teaching.

Delimitations

It was not the intent of this research to conduct a case study to collect detailed qualitative data; rather it was to examine from a distance the potential relationships between an information-rich hiring process, position fit, and teachers' intentions to remain classroom teachers at Texas public school campuses. It was also not the intent of this study to examine all variables included in the teacher survey instrument, but rather to conduct a quality study with a defined focus on the survey responses specifically related to potential predictive relationships between an information-rich hiring process, position fit, and teachers' intentions to remain in the classroom.

Limitations

This study had limitations that require acknowledgment but are not serious enough to threaten the validity of the conclusions. The first limitation was that the data are not generalizable to all populations because the sample was not random. However, "In some types of research, the researcher emphasizes the total context in which the research took place to enable readers to make judgments as to the *transferability* of the study's results to their own situation" (Mertens, 2005, p. 4). Details are provided in Chapter Three that emphasize the total context of the sample. Readers have the necessary information to determine the transferability of the findings presented in this study.

The second limitation was that there is not a professional consensus on the definition of "effective" or "quality" teacher.

The third limitation was the reliance on self-reported data through the administration of the electronic survey instrument.

The fourth limitation was that *ex-post facto* methodology does not allow for control over a situation that has already occurred. According to Leedy and Ormrod an *ex-post facto* design is "an approach in which one looks at conditions that have already

occurred and then collects data to investigate a possible relationship between these conditions and subsequent characteristics or behaviors" (2005, p. 8).

The fifth limitation was that because this study was limited to after-the-fact data, without the ability to manipulate treatment, cause and effect could not be assumed.

Assumptions

In this research the assumption was made that teachers responded honestly to the questions on the electronic survey instrument. "Surveys rely on individuals' self reports of their knowledge, attitudes, or behaviors," explains Mertons, "Thus the validity of the information is contingent on the honesty of the respondent" (Mertons, 2005, p. 167). For more detailed information about the survey instrument, refer to Chapter Three of this study. In addition, although not produced through a random sample, school district leadership and university preparation program stakeholders should benefit from the findings of this study.

Chapter Summary

Chapter One introduced the contexts that affect hiring environments of Texas schools, including the USDE blueprint of upcoming changes to the ESEA Act, and included an examination of factors involved in teacher selection and assignment. A review of literature and previous studies created the conceptual framework to examine how the hiring environment and subsequent teacher-position-organization fit may predict the likelihood that teachers will remain classroom teachers. This chapter included a statement of the problem, research questions, explanation of the methodology, and

significance of the research, as well as a definition of terms and statements of delimitations, limitations, and assumptions of this study.

Organization of the Study

This research study is organized into five chapters, with additional appendices and references included at the conclusion of the study.

Chapter One provides an introduction and summary of the literature related to teacher selection, assignment systems, and the relationship those systems have with teachers' intentions to remain in the classroom. In addition, Chapter One provides the problem, the purpose of the study, and the research questions and hypothetical constructs guiding the research. Included is a description of the methodology, operational definitions, and a discussion of delimitations, limitations, assumptions, and organization.

Chapter Two provides a literature review of research pertaining to the selection and assignment of teachers; the importance of teacher hiring systems that employ an information-rich hiring process; the importance of ensuring position fit of teachers with their classroom and campus assignments; and the relationship of these elements as potential predictors of teachers' intentions to remain in the classroom.

Chapter Three outlines the quantitative methodology used to design and conduct this research study.

Chapter Four provides an analysis of the data and findings of the teacher survey.

Chapter Five presents findings, conclusions, and implications of the study results. This chapter also proposes areas for further research, as suggested by the results of this study.

Chapter Two: Literature Review

In 1969, DL Bolton conducted a study that established a perspective still applicable in 2011:

Each teacher represents a potential gain or loss to the school system in terms of goal accomplishments. Therefore, the teacher selection process provides an opportunity for an educational administrator to make a major contribution to the improvement of a school system. In addition, it affords an example of how the decision process itself can be studied systematically and results generalized to many administrative tasks. (1969, p. 329)

Bolton's study is relevant 40 years after it was first published. The paramount role a teacher plays in student achievement was codified through the enactment of NCLB (2002) and the USDE blueprint report (2010) that outlines proposed changes for the next authorization of the ESEA. The blueprint indicates that a teacher's role in school systems will be further solidified in the future and establishes an environment that values decisions surrounding the selection and assignment of teachers (USDE, 2010). The blueprint also suggests that strategies will be expected that help systems provide a more equitable distribution of quality teachers at hard-to-staff campuses. While the terms *effective* and *quality* are often used interchangeably when describing the ideal teacher candidate, education stakeholders continue to search for means to ensure that all students have access to teachers who provide advantageous learning environments. Though many studies have reviewed components of the teacher selection process separately, few have

examined the system as a whole. Liu and Johnson (2006) conducted a holistic review of district selection systems when they examined post-hire outcomes from the perspective of recently hired teachers. Their research laid the foundation for this study —a holistic review of teacher hiring environments and newly hired teachers' intentions to remain in the classroom. The literature did provide evidence that urban schools have persistent challenges and recommended continued scrutiny of urban schools' selection and assignment processes. Continued holistic study of teacher selection and assignment systems may yield findings that can aid districts in finding and keeping quality/effective teachers and in preparing to meet the expectations anticipated from the reauthorization of the ESEA.

This literature review provides an overview of the teacher selection and assignment systems in public school districts and identifies attributes of hiring environments that ensure the very best teachers are selected for the students who most need them, including students at high-need, hard-to-staff urban campuses.

According to Mertens (2005), a literature review "serves to explain the topic of research and to build a rationale for the topic that is studied" (p. 88). The topic of this review was teacher selection and assignment systems as they relate to teacher retention and the hiring challenges faced by urban schools. The purpose of this topic was to determine if teachers' perceptions of their recent hiring experiences and their perceived fit with their assigned position and campus are predictive factors in teachers' intentions to remain in the classroom. This review is relevant because federal and state policies continue to increase expectations for student achievement and accountability standards

for teachers, principals, and other district leadership. This review is timely also because the reauthorization of the ESEA is expected in the near future, and the USDE blueprint (2010) indicates a movement toward using student achievement growth on standardized assessments as part of teacher and principal evaluations and also indicates the expectation for a more equal distribution of effective teachers at urban schools. These relevant factors are connected to school leadership's ultimate goal of providing students with a stable campus workforce of teachers who have been selected specifically for the particular assignment and campus to which they were hired.

As suggested by Harlen and Schlapp (1998), this chapter contains key sections of empirical work; these include an introduction, methodology, main body, and conclusions. The main body of review is organized by category and includes: an examination of policy and organizational factors that affect the selection of teachers; an inspection of teacher recruitment as a critical element of the teacher selection process; a scrutiny of system components, including the role of school principals in teacher selection; an appraisal of challenges in teacher selection systems often incurred by urban schools; and factors that influence teachers' intentions to remain in the classroom. Implications for further research are included throughout the main body. This chapter ends with conclusions drawn from review of the literature.

Methodology of the Literature Review

Mertens (2005) identified three items important to the development of a search strategy: (a) preliminary sources, (b) primary research journals, and (c) personal networks. This literature review includes five major databases consulted as preliminary sources: (a) EBSCO, (b) Elsevier, (c) JSTOR, (d) Wiley InterScience, and (e) WilsonWeb. From these databases, peer-reviewed primary research journals and papers were identified and then additional resources were located by examining reference lists of the most relevant research articles. To glean more information and locate further resources, networking was done with professional colleagues who hold interest in the selection of effective teachers. Key search terms included: (a) teacher hiring, (b) effective teachers, (c) teacher retention, (d) job fit, and (e) job satisfaction. Although the majority of studies selected were completed after federal policy, with the enactment of the NCLB Act, acknowledged the selection of teachers as a high-stakes endeavor, research prior to this time was included in order to incorporate seminal findings still applicable today.

Policy and Organizational Factors that Affect Teacher Hiring Practices

With the multiple policy and organizational factors affecting how public school districts hire teachers, it is not surprising that many researchers have inferred that districts don't hire the best teachers. Some researchers have presented evidence that districts hire teacher applicants who lack strength in particular personal and professional attributes (Ballou, 1996; Ballou & Podgursky, 1995 & 1998; Pfluam & Abramson, 1990), while others suggested that difficulties in hiring the most qualified teachers is in part due to a lack of efficiency in district hiring systems (Darling-Hammond, 2001; Levin & Quinn, 2003; Levin et al., 2005; Liu, 2002; Liu & Johnson, 2006; The New Teacher Project, 2007). Despite these inefficiencies, teacher-related policy continues to develop as

standards for students escalate, and the need for hiring and retaining teachers who can help prepare students to meet increased expectations is becoming more essential.

Teacher policy reform. The U.S. Department of Health, Education, and Welfare commissioned a report that documents teachers as key to educational policy reform; *The Coleman Report* (1966) established that the quality of teachers is a factor in improving test scores of economically disadvantaged students—a factor controllable by policy makers and school district leadership. Hanushek, as cited in Heck, explained that an important outcome of the Coleman Report is that it "directed the attention of researchers and policymakers toward student achievement as the primary means to assess the effectiveness of educational efforts" (2004, p.132).

Another important study by Bolton (1969), established that the teacher selection process provides opportunity for administrators to contribute to the overall improvement of school systems. Ferguson (1991) made a contribution to the literature when he found that teachers of African American and Hispanic students in Texas schools had lower verbal scores on standardized tests, and the performance of these students improved when their teachers had particular attributes that included strong verbal skills, more years of teaching experience, and a master's degree.

The connection between good teachers and student achievement continued to appear in the educational horizon, evident when the NCTAF (1996) declared the goal of having a *high quality* teacher for every student by 2006. The report upheld that quality be determined from the professionalism of the teacher, which includes the teacher's

27

content knowledge and pedagogy skills. NCTAF also called on policy makers to provide environments that support these goals.

The enactment of the NCLB *Qualifications for Teachers and Professionals* (2002), established a new focus in policy that set minimum requirements of highly qualified teachers, defined by a bachelor's degree, subject matter competency, and state licensure. Although not included as a component in NCLB data calculations, another data-centered teacher evaluation method indicated that teachers are the most important factor in student learning. This *value-added* approach to evaluation calculates teacher contribution to student test scores over a specified time period. Limitations exist as the data calculations can only be made for teachers of core content areas whose students take standardized assessments (Wright et al., 1997; Sanders & Horn, 1998).

With the onset of the Obama administration and a new set of policy actors at the federal level, a new approach shifted the focus from highly qualified teachers (a focus on qualifications that describe teacher quality) to highly effective teachers (a focus on student achievement outcomes) as seen in the USDE blueprint (2010) that outlines federal intentions for the reauthorization of the ESEA. The report indicates that student achievement growth on standardized assessments will be a major component in the determination of whether a teacher is highly effective, and the report affirms the identification process will also include: a state definition of effective teachers and principals; linkage of teacher and principal evaluations to student test scores; and incorporation of a federal definition of highly effective at the state level. In addition, the blueprint advises that particular focus be placed on the equitable distribution of effective

teachers at high-need schools, although few details are provided as to how this will be accomplished.

With the anticipated reauthorization of the ESEA in the near future, school districts have a brief window of time to review teacher selection and assignment systems to ensure they are prepared to incorporate the new policy mandates shown in the USDE blueprint to be highly prescriptive and centralized in nature.

Affects of decentralized and centralized hiring environments on teacher selection. As shown throughout educational policy literature, the movement between centralized and decentralized policy in school systems is ongoing and illustrated as a *continuum* by Hanson (2003), a *spasmodic effort* by Cooper, Fusarelli, and Randall (2004), a *trend* by Heck (2004), a *flip flop* in school reform by Hannaway and Stanislawski (2005), and as *cyclical in nature* by Schwartz (2010).

Hannaway and Stanislawski (2005), and Heck (2004) considered the radical reform that occurred during Chicago's effort to incorporate site-based management in the 1980's—in which the governance of education moved from the purview of bureaucracy to the purview of the community—to be a mechanism to remove power from state lawmakers and give it to the discretion of the local community. Heck also showed that such site-based governance may be examined through a cultural lens because policy makers' personal values become embedded into policy that ultimately affects district systems (2004).

Micropolitics as an aspect of organizational change in schools provided another lens to examine decentralization, as Björk and Blase (2009) found in their qualitative longitudinal case study of the mid-sized suburban Drayton County Public School District. The study results provide evidence that district management (central office) can both promote and deter decentralized policies mandated by state legislature and local school boards. Björk and Blase recommended that future studies investigate the role of middle management acting as a liaison between the superintendent and campus principals during trends toward decentralization (2009).

When scrutinizing continuums of decision-making and centralization of authority, scholars surmised that districts are rarely on one extreme end of the continuum. Instead, it was found that a more common practice for districts is to implement a mix of centralized and decentralized practices (Hannaway & Stanislawski, 2005; Hanson, 2003; Liu, 2002; Wise et al., 1987). Liu (2002) found that even when district practices lean more heavily toward decentralization, there is evidence that district leadership doesn't use this status to their advantage. The evidence suggests that districts are not utilizing the opportunity to employ decentralized practices so as to maximize the exchange of information between the teacher applicant and the campus or district staff involved in hiring the applicant.

Due to the lack of empirical research on how teachers are hired and the degree to which centralized or decentralized practices are deployed, Liu (2002) conducted a pilot study in New Jersey that measured the hiring experiences of new teachers. The pilot results indicate that three types of hiring are used among districts: (a) highly centralized, (b) highly decentralized, and (c) a mixture of moderately centralized and decentralized. A larger follow-up study conducted by Liu and Johnson (2006) studied the hiring experiences of new teachers in California, Florida, Massachusetts, and Michigan and found different results. The results of the larger study showed that most teachers are hired through a decentralized process, with the principal as the most highly involved participant in the selection process (Liu and Johnson, 2006).

During an experimental study, Young and Miller-Smith (2006) investigated the effects of state mandated site-based councils in the screening portion of the teacher selection process. Their study used role theory and a 2x3x2 factorial design to vary the roles of state legislation, principal, teacher, or parent as decision maker, and academic performance of a campus as either high- or low-performing. The population purposively selected for the study consisted of all elementary campuses in Kentucky, a state that has decentralized teacher-hiring practices due to the enactment of legislatively required sitebased councils, and in Ohio, which has no legislatively prescribed site-based councils. Young and Miller-Smith found that state mandated site-based councils do not have substantial effects on screening candidates for elementary teaching positions, nor were there differences found in the screening process between low and high-performing campuses. The two researchers noted that although seminal research by Wise, Darling-Hammond, Barnett, Berliner, Haller, Praskac, & Schlechty (1987) assumed that highperforming schools implement different teacher selection processes than low-performing schools, there has been no empirical research to confirm this assumption. Young and Miller-Smith recommended that future studies examine school achievement levels in concert with comparison of the quality of applicants in teacher supply pools of low- and high-performing campuses.

31

Cooper, Fusarelli, and Randall (2004) ascertained that trends intended to improve teaching included *centralized and tough policy* as established by the NCTAF, which advocates that only teachers who are licensed and accredited through traditional programs be in the classroom versus those who are alternatively certified, and a *decentralized, deregulated policy* trend, which asserts that too much regulation will deter quality teachers from entering the profession, and maintains that staffing decisions be made at the campus level. Scholars also found that mixed messages are often sent by policy makers, who allow site-based management yet limit resource allocations requested by local site-based constituents that are needed to prepare students for the increased standards and expectations issued by the same policy makers (Cooper, et al., 2004; Hannaway & Stanislawski, 2005). An example of this power struggle is forecasted in the USDE blueprint (2010) that shows federal intentions to shift teacher-training funding from formula to competitive grant programs. This proposed shift indicates that federal government will dictate to local government how grant monies must be spent. While the proposed changes to federal grant programs are an indication of a centralized process, a contradictory trend towards decentralization is established as the charter school movement expands under federal purview. As upheld by Schwartz (2010) during a comparison of practices between traditional and charter schools, the movement to expand charters is a prime example of decentralized practices to the extreme degree.

During an examination of San Diego's school reform effort that began in 1998, Hannaway and Stanislawski (2005) discovered that there was value to the movement termed *flip flopping* in the context of a large-scale momentum involving different spectrums of centralization and decentralization. The researchers theorized the *temporal* order of decision shifts should be considered, as initial centralized decision-making paved the way for future decentralized decision-making. Hannaway and Stanislawski's research did not result with conclusive findings regarding the reform and long-term affects on improved student achievement.

Hanson (2003) found that affirming the effects of decentralization on student achievement is a challenging endeavor due to the numerous variables involved. Galiani, Gertler, and Schargrodsky (2008) addressed this challenging endeavor in their study on decentralization in relation to student achievement when they used standardized test scores to analyze the effects of decentralization on educational quality by poverty levels. This international study involved a sample population of students in 99% of public secondary schools in Argentina from 1994–1999 as schools were transferred from central to provincial administration. After the move to decentralized oversight, local provinces became responsible for all budgetary matters, including the selection and assignment of teachers. Study results showed the decentralization movement benefitted students in nonpoor schools, but did not improve student achievement in poor provinces. The researchers suggested that future decentralization reforms incorporate systems of support for poor communities that may not be able to attract more qualified staff who are prepared to take advantage of local decision-making opportunities.

Hiring delays as obstacles. Within the framework of policy and organization, current hiring practices have been found inadequate; succinctly stated by Liu and Johnson in the title of their research (2006, p. 324) the process is "late, rushed, and information

poor." Several research sources evidence that late hiring leads to attrition patterns where stronger candidates drop out of applicant pools leaving only the less qualified teacher applicants available for hire (Castetter & Young, 2000; Darling-Hammond, 2001; Levin & Quinn, 2003; Levin et al., 2005; Liu & Johnson, 2006; The New Teacher Project, 2007; Young, 2008). Levin and Quinn (2003) found that employers in urban school districts value applicants with stronger academic credentials and who are qualified to teach critical shortage areas. However, because of drawn-out hiring processes, district administrators can't hire these stronger candidates who have accepted job offers elsewhere.

Systems and policies that lead to delays in hiring are reviewed next in the context of teacher recruitment, a precursor to the selection and assignment of teachers.

Recruitment of Teachers

Recruitment is intertwined with, essential to, and dependent on the teacher selection process (Castetter & Young, 2008; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008). Young defined recruitment more succinctly as "a human resource function involving the generation of applicant pools" (2008, p. G–8). Guarino et al. (2006) examined empirical research of recruitment and retention that used the conceptual framework of supply and demand economic labor theory. Guarino et al. verified that district recruitment and retention policies, in addition to current labor market conditions, impact individual decisions to enter or remain in teaching positions.

Supply and demand. Guarino et al. (2006) described *supply* in terms of *attractiveness* of options, meaning that individuals will choose to teach if teaching

represents the most attractive of all available options. In light of this theory, these researchers recommended that policies be adopted that make local teaching assignments the most attractive of options so as to entice the most qualified individuals to enter the applicant pool, thereby meeting the *demand* for qualified teachers. Guarino et al. also showed that the demand side of the market is influenced by factors such as the fluctuation of student enrollment numbers, class size ratios, and budgets affected by local, state, and federal economies. Teacher quality, in terms of effectiveness in positively influencing student achievement, is another variable that districts can manipulate in the supply and demand framework (Guarino et al., 2006).

Smith (2009) found that recruitment practices that focus on attracting teacher applicants with preestablished attributes, as opposed to recruitment practices that focus on attracting *more* applicants, ultimately improve the selection process. Levin and Quinn (2003) concur that more aggressive and targeted recruitment strategies lead to a surplus in applicant pools, but warn that without sound selection processes in place, there is no guarantee the best of applicants will be employed from the surplus pool. Likewise, increasing teacher recruitment or retention rates, as revealed by Guarino et al., does not ensure that the teachers entering or remaining in the system are effective in improving student achievement (2006).

Ingersoll (2001) asserted that the root of staffing problems is related to factors, such as job dissatisfaction, that must be reviewed from an organizational perspective that includes a review of supply and demand imbalances that establish a *revolving door* of new teachers replacing teachers who are leaving their jobs. In addition, Young (2008)

recommended improving the workforce by recruiting enough qualified applicants that teachers could be *selected* for a particular position instead of being *assigned* to a particular position. DeArmond, Shaw, and Wright (2009) substantiated the idea that problems related to the supply and demand of *human capital*—in this case teachers—often originate in the way districts approach staffing; and they recommended that further investigation be conducted on the benefits and detriments of district hiring practices in terms of human resource management.

Teacher recruitment theories. Delli and Vera (2003) examined *recruitment-asmarketing theory*, or ways in which recruitment messages influence applicants' perception of job desirability, and revealed that administrators use job interviews as an opportunity to "sell" the school. Likewise, scholars found that the motivational needs of teacher candidates also play a role in the recruitment and selection process. Castetter and Young (2000) and later Young (2008) explained and compared motivational needs: The *objective theory* of motivation illustrates applicants as economic beings, the *subjective theory* establishes applicants as psychological beings with needs to fulfill, and the *work itself theory* shows applicants as rational beings. Castetter and Young ascertained that teachers are least influenced by the objective theory, or economic incentives, and most influenced by subjective theory.

Another recent perspective proposes that the interests of the applicant must be considered throughout the hiring process (Clement, 2008;Young, 2008). Liu and Johnson (2006) surveyed teachers to discern the applicant perspective of the hiring experience, and found that many teachers hiring experience involves a poor exchange of information

that affects teachers' post-hire decisions to remain or leave their teaching positions. Liu and Johnson recommended that future research be conducted to ascertain how to improve information exchange in hiring processes (2006).

Other researchers found that the *realistic job preview theory*, as first established by Dugoni and Ilgen (1981), can be applied to the district perspective because a recruitment message that accurately describes an open teaching position is more likely to lead to job satisfaction after an applicant accepts a job offer and begins employment (Castetter & Young, 2000; Clement, 2008; Hays & Behrstock, 2009; Liu & Johnson, 2006; Seyforth, 2002). To implement realistic job preview theory, Seyforth recommends using a *job model* that presents attractive and unattractive features of an open position, rather than a *job description* that assumes all teaching positions are alike (2002).

21st century teacher recruitment methods needed. Hess (2009) termed the human capital pipeline in education *anachronistic* (p. 116) and found that private sector practices are more evolved and offer more incentives. Hess ascertained that a 21st Century approach to recruiting will not assume that teacher candidates are recent college graduates, nor that candidates have intentions to make teaching a lifetime career (2009). Strauss, Bowes, Marks, and Plesko's (2000) Pennsylvania study confirmed that districts take a more traditional approach to recruitment, however the private sector, claims DeArmond et al. (2009), uses a more aggressive approach. Many researchers corroborated that if districts are not more proactive with assertive and proficient teacher recruitment, qualified individuals will choose careers in private sector labor markets

(Behrstock & Clifford, 2009; Hays & Behrstock, 2009; Johnson & Birkeland, 2003; Smith, 2009; Winter & Malloy, 2005; Young, 2008).

Another component of 21st Century recruitment involves the *Generation Y* supply pool. Behrstock and Clifford chronicled generations of teachers and cite Shaffer's definition of Generation Y as "the cohort of people born between 1977 and 1995" (2009, p. 2). Their research shows that this cohorts' comfort and expertise with technology make it unlikely that they will accept a job in any field that lacks sufficient technology. Thus, Behrstock and Clifford assert that districts must ensure their work environments are equipped with current technology.

Other elements of recruitment that extend through the 21st Century involve finding teachers with particular content expertise. McGraner (2009) addressed this need when she developed key strategies for districts to use when recruiting science, technology, engineering, and mathematics (STEM) teachers whose skills are increasing in demand as policy makers increase expectations for students to demonstrate college and workforce readiness. McGraner found that systematic and streamlined recruitment processes should include strategies that provide (a) opportunities for an information-rich exchange between applicant and school system, and (b) multiple pathways to entice professionals who have had careers in some aspect of STEM outside of education to join the education community.

Teacher Selection and Assignment Systems

Addressing the American Federation of Teachers in 2008, then-Senator Obama made a statement regarding the critical role that teachers play:

Real change is finally giving our kids everything they need to have a fighting chance in today's world. That begins with recognizing that the single most important factor in determining a child's achievement is not the color of their skin or where they come from; its not who their parents are or how much money they have. It's who their teacher is. (Obama)

President Obama's viewpoint of a teacher's role in student achievement shows that school leadership should ensure that district selection and assignment systems do not utilize the widget effect—the assumption that teachers are interchangeable parts (Weisberg, Sexton, Mulhern, & Keeling, 2009).

Smith (2009) found that the onset of school-based learning, professional learning communities, and site-based decision-making teams bring change to human resource administration because critical staffing decisions are shifted to the campus level. Shifting in responsibilities is based on the assumption that as the principal and site-based team make collaborative staffing decisions, it is more likely that the needs of all community members will be met (Sorenson & Goldsmith, 2009). A consensus of research presents evidence that all participants involved in the process of selecting teachers should receive applicable human resource training (Hays & Behrstock, 2009; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008). Training in laws and regulations that govern teacher-hiring processes were also found to be pertinent, and are addressed in the next section.

Avoiding Selection bias. Any and all personnel involved in hiring decisions must be cognizant of laws and regulations that protect individuals and groups by

forbidding discrimination on the basis of *protected class status* (Castetter & Young, 2000; Smith 2009; Young, 2008). Employers should be particularly aware of *Title VII of the Civil Rights Act of 1964*, which protects class groups of race, color, religion, national origin, and gender, and prohibits discrimination on the basis of these legislatively defined group characteristics. *In part* (a legal term used in an employment context) means that even if the discrimination is only tangentially related and not a primary or controlling factor in the employment outcome, it is considered discrimination and can result in legal action. (Young, 2008). Valente and Valente (as cited in Young, 2008) found that districts that have carefully constructed hiring policies are more likely to comply fully with mandated public employment legislation, and those making hiring decisions are able to avoid formal challenges or sanctions.

As described on the U.S. Equal Opportunity Commission website (2010) *Title I of The Americans with Disabilities Act of 1990* (ADA) was passed to protect individuals *otherwise qualified* to perform essential job functions. The ADA added a new component to the employment process—the concept of *essential functions*. This component requires employers to make their hiring determinations with a focus on whether or not the job applicant could perform the essential functions, or job-specific criteria, with or without accommodations. According to one researcher, when interviewing teacher candidates:

Approximately 70% of the interviewer's questions should be competency-based and should focus on tangible instructional skills (e.g. how to begin a lesson), professional knowledge (e.g. copyright laws), classroom behavior (e.g. pacing classroom instruction), and interpersonal skills (e.g. dealing with a difficult parent or a parent in general). The questions should focus on candidate behavior. (Peterson, 2002, p. 59)

Seyforth (2002) confirmed that law will be violated if an employer choses an applicant without a disability because the person can perform peripheral functions of the job that another applicant with a disability can not perform. If an allegation of discrimination is made, the applicant needs only to establish *prima facie* evidence, or at first view evidence of discrimination (Young, 2008). Once a teacher applicant is able to establish prima facie evidence, the burden of proof legally shifts to the school district. To prevent legal challenges from job applicants, Young recommended that districts and school boards keep documentation of flow statistics that reflect "(1) the number of applicants, (2) the number of hires, and (3) the selection ratio as defined by the percentage rates within a particular employee classification" (2008, p. 136).

Another recommendation is that local school boards establish either *equal opportunity* hiring policies or *affirmative action* hiring policies (Castetter & Young, 2000; Young, 2008). Heneman, Judge, and Heneman described equal opportunity policies as *facially neutral*, meaning all applicants and employees are treated "without regard to the protected class characteristics such as race and sex" (as cited in Young, 2008, p. 98), thus allowing employment decisions to be based solely on merit. According to Young, affirmative action policies favor certain protected class groups with a goal of correcting past injustices that the protected groups have incurred (2008). Young emphasized that it is important for school boards to consider internal and external information regarding the composition of the current workforce when establishing the use of either equal opportunity or affirmative action policy. Once a policy is established, it can become a stabilizing factor in district hiring practices and in establishing organizational history, and it can aid school districts in avoiding discrimination-based challenges and sanctions.

Administrators should consider additional germane statutes and regulations when selecting teachers, as enforced by the U.S. Equal Employment Opportunity Commission (EOC) (2010), which include: (a) *The Pregnancy Discrimination Act*, (b) *The Equal Pay Act of 1963*, (c) *The Age Discrimination in Employment Act of 1967*, (d) *Sections 102 and 103 of the Civil Rights Act of 1991*, (e) *The Genetic Information Nondiscrimination Act of 2008*, and *(f) *Section 504 of the Rehabilitation Act of 1973* (*Not directly under the purview of the EOC).

In addition to compliance with legal parameters, the purpose and goals of an efficient selection system influence hiring decisions and are discussed in the next section.

Purpose and objectives of teacher selection. Young recommended that a district selection system be based on findings of professional literature as follows:

Based on this literature, the purpose of selection is to fill vacant positions with personnel who meet the system's desired qualifications, appear likely to succeed in a designated position, will remain in the system for a reasonable period of time, will be effective contributors to the system at large, and will be sufficiently motivated to fulfill their job assignments at present as well as in the future. (2008, p. 131)

42

Seyforth (2002) documented selection as a process with four objectives: (a) ensuring applicants have needed knowledge and skills, (b) helping applicants make informed decisions, (c) creating a sense of commitment to the organization, and (d) providing support for those hired. Selection is a persisting organizational challenge for districts and campuses, and meeting the challenge in a systematic manner is critical in preventing errors (Castetter & Young, 2000; Young, 2008). Delli and Vera (2003) found that a properly conceptualized system that is executed efficiently is beneficial to all.

From an organizational perspective, two types of selection decision making were demonstrated: (a) top down, or administrator, and (b) site-based; districts often use a combination of the two types of decision-making. An example of a mixed approach involves an initial screening of applicants by central office administrators, followed by the principal and site-based team selecting applicants who make it through the initial screening (Hannaway & Stanislawski, 2005; Hanson, 2003; Liu, 2002; Seyforth, 2002; Smith, 2009; Wise et al., 1987; Young, 2008). Some scholars found that a lack of empirical research exists that designates one approach as most effective (Hanson, 2003; Galiani, et al., 2008; Young, 2008; Young, & Miller-Smith, 2006).

Selection as a process, not an event. In the past, researchers viewed selection as an *event* with the interview as the most important feature, but Young (2008) describes selection as a *process*, wherein the interview is only one of many tools. Many other scholars also claim that selection decisions are made throughout the complex hiring process (Castetter & Young, 2000; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008).

Procedural stages of selection. Selection is a two-stage process that involves: (a) screening decisions, usually involving the review of paper credentials, condensing the applicant pool, and determining which applicants advance to the interview stage; and (b) employment decisions, usually involving interviews, reference checks, and subsequent selection (Seyforth, 2002; Smith, 2009; Young, 2008). Sorenson and Goldsmith (2009) report that during the two-stage selection process, considerations should be given to academic criteria; personal characteristics and qualifications; professional characteristics and expertise; relative experience and professional development (p. 107). In addition, Seyforth showed that some of the most essential information sources include the application form, transcript, references, applicant test scores, and interviews (2002).

Foundation of the selection process. Many scholars established that a *job* or *position analysis* should precede screening and employment decisions and should involve a systematic technique for collecting all aspects of a particular teaching assignment in order that this preestablished criteria guide the matching of applicant qualifications to specific assignments (Castetter &Young, 2000; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith 2009; Young, 2008). One demonstrated method for accomplishing this type of analysis was through a *job model* that involves the identification of tangible results sought, a straightforward description of the job environment, and development of priority actions that describe job demands in narrative format (Seyforth, 2002).

Human resource scholars showed that after the development of a job analysis, the next step is the development of *criteria measures*, or *job-specific criteria* that include predetermined measures of job performance and knowledge and skills needed for a

particular teaching assignment (Castetter & Young, 2000; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008); these researchers considered the development of these criteria as the foundation of the selection system that the principal and the site-based group should establish prior to the interview. Sorenson and Goldsmith (2009) affirmed that development of the criteria helped to establish (a) organizational goals, (b) position design, (c) position performance measures, (d) position skills and performance success, and (e) the selection tools to be used.

A helpful resource sanctioned by the CCSSO (2005) for the Interstate New Teacher Assessment and Support Consortium (InTASC) was designed to help administrators hiring *first-year* teachers. The resource contains ten standards and accompanying indicators that reflect the professional consensus of what beginning teachers should know and be able to do. The CCSSO (2011) recently revised the standards to outline what all teachers at all phases of their teaching career show know and be able to do effectively. Appendix A of this study contains the revised standards.

Additional selection tools used as resources during the teacher hiring process are examined in the next section.

Selection tools. Rutledge, Harris, Thompson, and Ingle (2007) grouped the tools used to hire teachers into two categories: (a) screening tools and (b) selection tools. During both stages of the selection process, scholars found *job predictors* or *predictor variables*, tools used to predict the degree of success of the applicant in the teaching position for which they are applying; the job predictors include a mixture of tools such as applicant test scores, interviews, references, and related experience (Castetter and Young, 2000; Young, 2008). These scholars also viewed some tools as indicators of personal or professional attributes of teachers (e.g., a high-test score is inferred to mean that a candidate has the desirable professional attribute of content knowledge and expertise). Unfortunately the literature did not reveal conclusive evidence as to which tools are the most effective.

Rutledge et al. (2007) found that principals preferred three hiring tools: (a) the interview, (b) experience, and (c) letters of recommendation. Young (2008) asserted that districts should preidentify a list of tools to be used in both stages of the selection process, even though the *number* of tools used vary from district to district.

In addition, scholars have presented decision-making models for both stages of the selection process that verify appropriateness of match between applicant and school (Castetter & Young, 2000; Seyforth, 2002; Smith, 2009; Young, 2008). Young illustrated that the *multiple hurdles model*, which requires a minimum level of competency on the job predictors, should be used first, followed by the *compensatory model*, which is centered on a composite score of competencies used to compare all candidates for a position (Young, 2008).

Hiring tools in the screening stage. Job predictors, or selection tools, in the screening stage are objective and usually supplied through written materials such as job applications, transcripts, references, and candidate test scores (Castetter & Young, 2000; Seyforth, 2002; Young, 2008). Scholars have shown that the principal should conduct a preinterview through the examination of these paper credentials to begin the sorting

46

process, and to formulate interview questions (Clement, 2008; Sorenson & Goldsmith, 2009).

The use of technology as a tool was demonstrated with online, commercial teacher selection instruments such as the Haberman Star Teacher Evaluation *PreScreener* and the *Gallup Teacher Insight Assessment*, which are often used by large urban districts to sort initial applicant pools. These tools are marketed as a centralized, timesaving approach that is more objective because of the scripted format and criterionscored results (Metzer & Wu, 2008). As stated on the Haberman Foundation website, the PreScreener categorizes tools into 10 dimensions: (1) persistence; (2) organization and planning; (3) values student learning; (4) ability to connect theory to practice; (5) approach to at-risk students; (6) approach to students; (7) ability to survive in a bureaucracy; (8) explains teacher success; (9) explains student success; and (10) fallibility. Rockoff, Jacob, Kane, and Staiger (2008) analyzed the Haberman tool to investigate whether or not it predicted teacher effectiveness and did find a modest, predictive relationship with the use of the tool as a nontraditional source of information. Rockoff et al. acknowledged that their research on the Haberman instrument is ongoing, and they encouraged district hiring systems to use the Haberman instrument and other nontraditional credentials during the selection process.

Metzer and Wu (2008) examined a precursor to the Gallup TeacherInsight instrument when they reviewed 24 studies of the Teacher Perceiver Interview (TPI) that measure beliefs, attitudes, and values of teachers; they concluded that what the instrument measures and how it relates to teaching effectiveness is ambiguous. However, the two researchers did confirm that the TPI captures attributes that are consistent with principals' preferences. Metzer and Wu recommended policy makers conduct future studies to scrutinize this costly tool that has captured the interest of district administrators.

Hiring tools in the selection or employment stage. Job predictors, or tools, in the second stage are subjective in nature and usually supplied verbally and through observation of behaviors, according to Young (2008). The interview is the most common tool used by employers (Castetter & Young, 2000; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008). Delli and Vera (2003) showed that despite its widespread use as the cornerstone of teacher employment, there is not a strong research base for the interview in this capacity. Delli and Vera found that the research base relied on "paper people" in hypothetical situations, thus making it difficult to apply findings to the face-to-face interview (p. 139). These two researchers recommended that contextual effects that include the structure, format, and influence of policies on outcomes be considered during teacher interviews.

Other hiring tools in the selection stage include job simulations and work samples which occasionally are used in the employment phase as a result of time constraints, often due to late hiring timelines and busy schedules that are not adjusted to allow time for preparation and observation. However, these tools are shown to generate accurate predictions of performance (Castetter & Young, 2000; Young, 2008).

Many scholars showed that the *unstructured interview* is a poor predictor of performance because employers can not uniformly compare the performance of

candidates; researchers revealed the *structured interview* as a more reliable tool in the prevention of incomparability due to the prescribed questions, which are prepared prior to the interview (Castetter & Young, 2000; Delli & Vera, 2003; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008). The structured interview format is either *dyadic*, having only one interviewer, or *panel*, having multiple interviewers, and is often used for sitebased hiring practices (Castetter & Young, 2000; Sorenson & Goldsmith, 2009; Young, 2003). Delli and Vera (2003) noted ambiguous findings in the literature as to whether the panel or dyadic interview was the best approach.

The structured interview questioning techniques are either *situational*, using the theory that intentions predicted behavior, or *behavioral*, using the theory that past behavior is the best predictor of future performance (Clement, 2008; Seyforth, 2002; Young, 2008). Seyforth (2002) suggested additional techniques that include *perceiver questioning techniques*—which are based on values and philosophies, style of interaction, and analysis of a problematic situation—and *critical incident questioning techniques*—which originated in the business sector and require the applicant to explain how she or he would react in certain situations.

Hiring theories and frameworks. Bretz and Judge (1994) inspected the role of human resource systems in job applicant decisions and found that variables in the system influence applicant choices. The two researchers determined that human resource practices are influential in prehire and posthire environments, particularly in the context of person-organization-fit (Bretz & Judge, 1994).

Human resource related theory and concepts are next examined in the context of teacher hiring environments.

Human resource management theory. DeArmond et al. (2009) found human resource management theory to be a critical function that provides districts an opportunity to more closely evaluate teacher applicants, provides support for information-rich exchanges, and leads to a good person-organization fit. The researchers recommended improving the quality of the teacher workforce by using human resource management to *zoom in* to confront ineffectual practices and to search the private sector for alternative approaches, and to *zoom out* to evaluate systems collectively (DeArmond et al., 2009). DeArmond et al. also revealed that districts should continue to look at human resource management in the context of what decisions should be *tight*, determined centrally, or *loose*, determined at the campus level. Behrstock, and Clifford (2009) found that human capital management is needed to attract Generation Y teachers who are unlikely to be enticed to join the public school system if there is indication that policies lag versus lead in reform efforts.

Person-job fit, person-organization fit, person-group fit theory. Rutledge et al. (2007) ascertained that the complexities of teaching involve levels of: person-job fit, such as certification requirements; person-organization fit, such as organizational coherence to contribute to a more effective school; and person-group fit, such as grade level or subject area planning teams. DeArmond et al. (2009) found that person-organization fit, namely teacher-school fit, is particularly important with regard to collegiality and working relationships as a feature of effective schools.

The interview is the tool used most often by administrators when hiring teachers (Castetter & Young, 2000; Harris et al., 2007; Rutledge et al., 2007; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008). Judge, Higgins, and Cable (2000) examined the interview as a tool for assessment of person-organization fit and define it as "the congruence between an attribute of a person and an attribute of an organization" (p. 393); they found distinctions between *actual* congruence, *perceived* congruence from the applicant's perspective, and *subjective* congruence that involves the interviewer's perceptions. Judge et al. established that perceived congruence is more influential on interviewer decisions, but the effect of the interview and interviewer on the job acceptance decisions of applicants was not confirmed.

Bowman (2005) used person-organization fit and person-job fit as a hiring framework in a survey administered to a random sample of principals and superintendents in a large Midwestern state to determine preferences when selecting teachers. Items in the survey instrument were divided into person-job fit concepts (empathy, communication skills, etc.) or person-organization fit concepts (values, beliefs, etc.), with various factors assigned to each concept. Superintendents and principals were asked to assign a rating to identify the extent to which that they would consider each item during a selection interview. Survey results indicate that when a principal conducts an interview with a teacher applicant, emphasis is placed on person-organization fit concepts, whereas when a superintendent interviews an applicant emphasis is placed on person-job fit concepts. Bowman found that principals focus on items related to school culture and vision, which relate to their direct responsibilities to the campus environment, and superintendents prioritize items in relation to how well the applicant interacts with students. Bowman also determined that principals and superintendents assign different levels of importance to various concepts; superintendents give more weight to concepts of input drive and beliefs, which align with their responsibility for the district as a whole.

Another study by Liu and Johnson (2006) used the concepts of personorganization fit and person-job fit in a survey administered to newly hired teachers to ascertain whether the teachers' recent hiring experiences led to their perception that their job assignment was a good fit. The authors posited that matching teachers to positions and to schools is important because no two teaching positions are the same, and because a good fit makes it more likely that teachers will be satisfied and remain in their positions, thus helping the stability of the campus. This study is discussed in more detail later in this chapter under the information-rich hiring process section.

Two-sided matching. Boyd, Lankford, Loeb, and Wyckoff (2006) examined the concept of two-sided matching based on *game-theoretic approach*, which is used in settings such as college attendance in which the matching is two-sided because it is dependent on choices made by the student (whether to apply or to accept an offer if extended) and the potential college (whether to accept the application and invite the student to attend). In a decentralized teacher labor market Boyd et al. substantiated that the concept of two-sided matching means both the administrator and the teacher make separate choices that work collectively to determine whether the open teaching position is considered by both parties to be a good match. Their study led the scholars to find that

52

the choices made by the applicant and administrator often lead to the unequal distribution of quality teachers (Boyd et al., 2006).

Boyd et al. used the matching theory to investigate how individual choices made by teacher candidates and administrators determine the allocation of teachers across jobs, with a specific focus on *initial* job matching. The scholars found that from the teacher perspective, a shorter distance from home to school is an important preference when deciding whether or not to accept a job offer at a particular campus. Boyd et al. determined that from the administrator perspective, teachers with stronger qualifications are preferred, which contradicts Ballou's (1996) findings that administrators do not give preference to applicants with stronger qualifications. Boyd et al. found a limitation to Ballou's data as it does not take into account choices from both sides of the potential match. In other words, even if an administrator wants to offer a job to an applicant with stronger qualifications, the applicant could choose not to work at the campus with the open position, thus leaving less qualified applicants in the supply pool. The researchers demonstrated the matching model to be a useful tool in analyzing the preferences and decision-making processes of both teachers and administrators.

Information-rich hiring process. Liu and Johnson (2006) introduced the *information-rich hiring process* as a conceptual framework that incorporates elements similar to those contained in theories discussed previously. As with the two-sided matching theory, the two researchers posited that information-rich hiring involves a two-sided process that provides an opportunity for teacher applicants and those hiring to "collect rich information about, and form accurate impressions of, one another" (p. 326)

through interactions that give the applicants a realistic job preview and give principals and site-based committee members opportunity to observe the candidate in school settings. Liu and Johnson showed that at the other end of the continuum there is a process referred to as an *information-poor hiring process*, because few opportunities are provided for exchange of information or interaction, which often leads to a teacher's dissatisfaction with the job assignment (Liu & Johnson, 2006).

Liu and Johnson addressed gaps in existing research by examining the hiring process in relation to posthire outcomes, as well reviewing how school systems organize and conduct teacher selection and assignment. Using the hypothesis that an information-rich hiring process facilitates a better match between applicant and position than an information-poor hiring process, the researchers administered an 85-item survey to teachers in order to glean their perspectives of their recent hiring experiences. The survey achieved a response rate of 69%, consisting of 486 newly hired teachers from four purposively selected states which include California, Florida, Massachusetts, and Michigan. Liu and Johnson (2006) acknowledged study limitations: reliance on self-reported data from the teacher surveys; administration of the survey in the spring semester, which allows the possibility that dissatisfied teachers may have already left their assignment and are not represented; and no distinctions between qualified and unqualified teachers were made (e.g. in terms of certification).

The two scholars examined the degree to which district and campus selection practices contribute to the soundness of the match for newly hired teachers in terms of the degree to which the teachers' skills and philosophies are perceived to coincide with the needs of students in their classrooms (person-job fit), as well as those of colleagues and their campuses (person-organization fit). Liu and Johnson scrutinized factors that affect person-job and person-organization fit including the degree of centralized versus decentralized decision-making and the information richness of district selection processes. The two researchers established a construct of an information-rich hiring process to measure the types and qualities of job information routinely gathered, distributed, and applied by both the teacher applicant and the school administrator to arrive at mutually beneficial job selection decisions. Liu and Johnson also examined timelines and sequence of district and campus selection processes in relation to affects on perceptions of fit.

Survey results showed that teacher perceptions are generally strong in relationship to classroom assignment (person-job fit), but less definitive with regard to campus assignments (person-organization fit). Despite the fact that three-fourths of survey respondents were hired in a decentralized process, the scholars found that selection systems did not allow time for site-based interviews or teaching demonstrations, and these untimely employment practices were identified as the most pressing deterrent to an information-rich hiring process (Liu & Johnson, 2006). Liu and Johnson surmised late hiring occurs because of fluctuating enrollment numbers, state budget decisions, collective bargaining agreements, and inefficient hiring systems; they recommended that state policy makers help counter this problem by removing impediments to informationrich hiring environments.

55

Principals' role in selecting effective teachers. The concept of principal as *steward* in the context of urban schools was included in the most recent Interstate School Leaders Licensure Consortium (ISLLC) Educational Leadership Policy Standards (CCSSO, 2008): "An education leader promotes the success of every student by facilitating the development, articulation, implementation, and stewardship of a vision of learning that is shared and supported by all stakeholders" (p. 14). Other scholars affirmed that principals in the role of steward have an ethical obligation to attract and hire the best teachers for their campuses (Castetter & Young, 2000; Harris et al., 2007; Rutledge et al., 2007; Seyforth, 2002; Sorenson & Goldsmith, 2009; Young, 2008).

Principals are exposed to a myriad of perspectives that define what constitutes an effective teacher. Corcoran (2009) established a working definition of teacher quality as the "set of teacher skills, knowledge, personal attributes, and pedagogical abilities that yield desired student outcomes (i.e., the level of teacher productivity or 'effectiveness')" (p. 30).

Although the literature does not present a uniform agreement of which teacher attributes are paramount indicators of effective teaching, Clotfelter, Ladd, and Vigdor (2007) found that a proficient selection process involves the consideration of each teacher characteristic, not in isolation, but in relation to the particular teaching assignment. Harris et al. (2007) identified shortcomings in existing literature that examine principals' selection of teachers in a holistic context.

Selection for 21st century learning. Hill (2009) revealed that new developments in technology call for different skills and require nascent perceptions of how teachers

should be selected and assigned to more innovative roles. Hill recommended that administrators be prepared to identify teachers who are trained and motivated to use instructional technology, particularly in the context of new paradigms such as virtual schools, or other school settings that involve other-than-face-to-face models of instructional delivery (2009). Behrstock and Clifford (2009) determined Generation Y's comfort and expertise with technology to be attributes that are beneficial in equipping students with 21st century skills and recommended that this expertise be factored into selection decisions.

While theories abound on what principals *should* seek when hiring teachers, studies that examine principals' own preferences reveal what principals *do* seek when hiring teachers.

Principals' preferences. Harris et al. (2007), and Rutledge et al. (2007) conducted two mixed-method studies in a mid-sized Florida district where principals have designated hiring authority. These studies examined principals' preferences for teacher attributes and analyzed the tools used during the selection process in relation to implications for teacher quality policy.

Harris et al. (2007) presented evidence that indicate principals give preference to a mixture of personal and professional characteristics when selecting teachers for their campuses; principals did not view attributes in isolation. The researchers found that staffing a mixture of teachers with various attributes is essential when considering person-organization fit and when ensuring that staff is diversified in terms of age, gender, and race.

Harris et al. (2007) confirmed that when identifying teacher applicants believed to have a good school fit, principals are more likely to consider campus demographic characteristics (such as age and experience) than they are to seek applicants who have similar teaching philosophies and styles as the current campus staff. The researchers also found that teaching experience is one of the top three attributes considered by principals in the selection process (Harris et al., 2007). Principals preferred personal attributes that included enthusiasm, strong communication skills, and ability to work well with others. Harris et al. found that some principals demonstrate preference for intelligence in the context of the applicant's academic background reflected in the transcript, yet others are satisfied with a teacher's attainment of a bachelor's degree as demonstration of intelligence (2007). No evidence is documented to support the propensity of principals to hire applicants with academic backgrounds that are similar to their own. Harris et al. (p. 25) found that principals' understanding of a teacher's subject matter expertise is seen in three ways: (a) content knowledge, (b) knowledge of state standards, and (c) knowledge of child development. The scholars identified the interview as the most favored tool of principals, followed by review of experience and recommendations. Harris et al. also established that although centralized policies affect hiring decisions, when principals find that a policy interferes with the unique needs of the students at their campuses, they are less likely to implement that policy with fidelity (2007).

Rutledge et al. (2007) found that preference for the interview as a hiring tool corresponds with principals' preference for personal attributes that can be determined through the interview. The researchers showed that principals deem four types of team

interviews helpful in ensuring person-group fit and person-organization fit: convenience teaming, spectrum teaming, grade-level teaming, and department teaming. Study results indicated that principals' use of screening instruments (such as Gallup's TeacherInsight, a scripted interview protocol) is inconsistent, and the researchers found no evidence as to the amount of weight that principals give to the results. Video or demonstration lessons were not ranked highly, in part due to time constraints and hiring done outside the school year. Rutledge et al. found no evidence that principals prioritize college course work or the resume as the most essential tools in selecting teachers. The researchers showed that less experienced and younger teachers were often seen as good hires by principals because they were more "pliable," "enthusiastic," and "untenured" (p. 20). The principals view these attributes as indication that these teachers are flexible and open to various roles and responsibilities at the classroom and campus levels. Rutledge et al. also determined that principals give preference to teachers that they can recommend personally—teachers they have observed as student or substitute teachers at their campus.

Trimble (2001) interviewed five veteran principals in high-poverty, highperforming schools in Georgia when conducting a study on what principals seek when selecting teachers. Trimble found these principals to seek teachers who have a strong work ethic, people skills, and communication skills (p. 46). The principals in the study asserted that these attributes provide a foundation for effective teaching that could be improved over time with mentoring and other methods of induction.

The three studies examining principal's perspectives and preferences contribute to understanding of the teacher selection and assignment environment in public school districts.

Urban Schools and Obstacles to an Information-Rich Hiring Process

Many scholars focus on the unique challenges urban districts face when hiring teachers, particularly in comparison with neighboring suburban districts (Liu, 2007; Levin & Quinn, 2003; Levin et al., 2005; Strauss et al., 2000). According to the ISLLC Leadership Policy Standards (CCSSO, 2008), the principal's role as instructional leader of the campus involves the attraction and retention of teachers who provide learning environments conducive to student achievement, a particular challenge for urban school principals. Principals staffing urban campuses face unique challenges.

Urban recruitment challenges. Papa and Baxter (2008) scrutinized hiring systems in New York State and confirmed that principals at urban campuses are disadvantaged as a result of less hiring autonomy, less time, and more limitations. The two researchers also found that urban principals have lower expectations than their suburban counterparts for the quantity of qualified applicants available in their applicant supply pool. Boyd et al. (2005, 2006) upheld evidence that urban districts face many staffing challenges that include: teacher preference for working in the areas in which they live (usually suburban environments), less competitive salaries, unfavorable working conditions, and disadvantaged or lower socio-economic student populations. Darling-Hammond (2001) confirmed that a shortage of qualified teachers only exists in the supply market in districts that are unable to offer the most attractive working conditions or

salaries; thus inner cities are more likely to have shortages while wealthier districts face surpluses. Jacob (2007) cited studies that provide evidence that urban school principals may not recognize or value high-quality teachers and posited that this occurs when principals have different objectives in hiring (e.g. seeking teachers to serve as a role models, or lacking information on what constitutes a quality teacher).

Late hiring timelines and other hurdles. Many researchers found that late hiring is more common in urban districts (Claycomb, 2000; Jacob, 2007; Levin et al., 2005; Levin & Quinn, 2003; Papa & Baxter, 2008; The New Teacher Project, 2007). During a study of the Chicago Public School District, The New Teacher Project (2007) found that despite an initially large applicant pool, late hiring timelines in the district led teacher applicants to accept assignments in districts that hire earlier. Jacob (2007), and Levin and Quinn (2003) found that two of the most significant barriers affecting urban districts include bureaucratic hurdles and a shortage of teacher candidates who desire positions at hard-to-staff urban campuses.

Teachers of urban campuses. The New Teacher Project established a definition of an effective teacher in the specific context of urban, high-need schools:

In a high-need school, an effective teacher consistently achieves average annual student growth of more than one (1) year and ensures that at least 80% of students meet grade-level standards or are on track for on-time high school graduation, or meets other appropriate measures of student academic progress. (The New Teacher Project, 2010).

Other definitions of effective teachers are anticipated after federal goals outlined in the USDE blueprint (2010) show that efforts will be made in the reauthorization of the ESEA to remedy the unequal distribution of quality teachers among schools.

Sachs (2004) asserted that instruments such as the Haberman tool are too costly and time consuming for use in identifying teachers that would be successful in urban environments and designed a study that used a new instrument to measure five attributes consistently identified by research as indicators of effective urban teachers: (a) sociocultural awareness, (b) contextual interpersonal skills, (c) self-understanding, (d) risktaking, and (e) perceived efficacy (p. 178). Two additional factors are identified as subcategories: (a) cultural responsiveness and (b) risk to personal safety. However, study results also indicate that none of the identified variables discriminate between highly and less-highly effective teachers. Sachs recommended that future research on teacher effectiveness in urban environments include control groups of ineffective teachers and that teacher preparation programs be revamped to support development of teachers who anticipate working in urban school environments.

Obstacles to staffing effective teachers at urban schools. Lankford, Loeb, and Wyckoff (2002) presented evidence that teachers in urban schools are less qualified, and therefore minority, low-income, and low-achieving students who most need qualified teachers, don't have them. Claycomb (2000) and Jacob (2007) confirmed that urban school students are less likely to have qualified teachers, particularly in critical subject areas such as mathematics and science. However, Jacob also found that *less qualified* didn't mean *less effective*, and documented studies that provide evidence that many

teacher characteristics are not directly tied to student achievement, with experience and high cognitive ability as the exceptions. Further research is needed to identify and define teacher characteristics tied to effective teaching, as compared with characteristics identifying qualified teachers.

The New Teacher Project (2003, pp. 36–42) recommended four broad actions to improve urban hiring, with sub-actions included for each action. The solutions include: (a) requiring teachers to give earlier vacancy notifications, (b) expedite inter-district transfer processes, (c) promote earlier and more predictive budget processes, especially for the harder to staff schools, and (d) streamline human resource department processes and increase the role in hiring at the campus level. Some of these recommended solutions assume that state decisions are subject to the will of the state teacher union. Texas does not have teacher unions that dictate district decisions. Texas does have teacher organizations that can advocate for their constituents, but without the authority given to unions in other states.

Darling-Hammond and Prince (2007) examined reasons that urban campuses who serve students with the most academic need experience shortages of qualified STEM teachers, and they found that shortages intensify as policy makers continue to increase high school graduation standards (e.g., requiring additional math and science credits), therefore making a more competitive market for STEM teachers. Claycomb (2000) and Darling-Hammond and Prince (2007) showed that urban schools that face shortages assign teachers *out of field* or with emergency teaching certificates. Darling-Hammond and Prince also identified working conditions such as insufficient technology structures and inadequate science equipment as reasons principals at high-need campuses have difficulty staffing teachers qualified to teach STEM related courses.

Claycomb and Darling-Hammond and Prince ascertained that some preparation programs do not prepare teachers to work with the diversity of learners in urban populations. Darling-Hammond and Prince documented that teachers often feel they have had inadequate preparation to be effective with English language learners, students in poverty, and other students with special learning needs who are frequently found in urban environments. The researchers also confirmed that teachers who felt ineffective with their students are more likely to leave their positions. Ensuring that teachers have a repertoire of strategies for diverse student populations is critical for successful urban school environments (Darling-Hammond & Prince, 2007).

Many scholars found that teacher attrition is concentrated at schools with high numbers of minority students, indicating students at these schools may systematically receive a lower quality of education because the quality teachers assigned to the campus are more likely to transfer to other campuses with better working conditions (Lankford et al., 2002; Hanushek & Rivkin, 2004; Scafidi, Sjoquist, & Stinebrickner, 2007).

Unequal distribution of novice teachers. Research confirmed that the educational outcomes for students who have novice teachers are not as positive as outcomes for those students who are assigned more experienced teachers (Claycomb, 2000; Clotfelter, Ladd, & Vigdor, 2005; Ferguson, 1991; Sanders & Horn, 1998). Achinstein, Ogawa, and Speiglman (2004) focused on the teaching assignments often given to first-year teachers as they showed that new teachers "can become agents in the

reproduction of social inequality" (p. 594). As the researchers documented, the *Matthew Effect* occurs when the rich (districts with high capital) get richer, and the poor (districts with low capital) get poorer in the distribution of quality teachers. Achinstein et al. further established a system of tracking first-year teachers composed of elements of state educational policies, district and school conditions, and the personal and professional backgrounds of teachers.

Clotfelter, et al. (2005) examined the unequal distribution of novice teachers across and within school districts and suggested that pressures that administrators face from parents on the demand side and that teachers face on the supply side, affect distribution. For example, at the same time that parents of students at urban, high-need campuses demand more experienced teachers for their children, the more experienced teachers demand transfers from the same campuses. Claycomb (2000) revealed that within a five-year period up to half of all new teachers leave urban school settings, costing administrators time and money to restaff their campuses.

Selection of teachers from traditional versus alternative paths to

certification. Boyd, Grossman, Lankford, Loeb, and Wyckoff (2005) conducted research that examined whether different pathways lead to the hiring of teachers who are more effective in improving student achievement. The researchers' intentions for the study were to glean information that can be shared with high-poverty urban schools, which turn to alternative preparation programs to fill their teacher supply shortages. Boyd et al. found that some differences exist, as those using different pathways are better suited to teach particular grades and subject areas. Jacob (2007) also reviewed studies on certification routes and found little evidence that either traditional or alternative pathways are more advantageous to urban systems. However, Claycomb (2000) suggests that individuals who have the potential to be effective urban teachers may never enter the teaching field because they are intimidated by traditional university based programs; these individuals may find alternative programs to be less intimidating.

Strategies for improvement of the urban teacher workforce. Jacob (2007) reviewed policies aimed at improving the quality of teachers in urban districts from the supply side, including policies that involved higher salaries, improved working conditions, and alternative paths into teaching and mentoring. Jacob found that demandoriented strategies focused on the improvement of district hiring practices, including streamlined procedures so that job offers can be made more expediently, improved identification of effective teachers through screening, and implementation of more decentralized practices in which principals and site-based members select teachers to ensure a better match between applicant and campus.

McGraner (2009) addressed the shortage of STEM teachers in high-need, and high-poverty schools and recommended that districts form university partnerships, provide financial incentives, and consider alternative certification pathways to more effectively compete for these teachers. Darling-Hammond and Prince (2007, p. 39) suggested four approaches that address the STEM shortage at high-need schools: (a) creation of a new pipeline specifically for these schools through traditional or nontraditional means; (b) redistribution of existing teacher pool to schools who most need them through use of financial incentives and policy change; (c) provision of intensive professional development, mentoring, and coaching to current math and science teachers at high-need schools; and (d) improved school working conditions.

Scholars also proposed the following to improve the quality of the teacher workforce at high-need urban schools: recruitment of individuals likely to work in urban environments and revision of teacher preparation program content to prepare candidates for teaching at urban schools (Claycomb, 2000; Darling-Hammond and Prince, 2007). It was also shown that urban teachers need systems of induction and that any factors that cause individuals hesitancy for teaching in urban environments should be minimized (Claycomb, 2000).

Teacher Turnover: Costly

Scholars found that teacher turnover is expensive (Behrstock & Clifford, 2009; Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Perrachione et al., 2008). The expense is monetary in terms of lost investment for schools and taxpayers, and nonmonetary in terms of loss of institutional memory and lowered morale for remaining teachers (Behrstock & Clifford, 2009). High levels of turnover can result in (a) higher costs for recruiting, hiring, and training replacement teachers, (b) students that have less experienced and less effective teachers, and (c) unstable workforces that prevent the development of relationships and coherent instructional programs that support student achievement (Miller & Chait, 2008).

Scholars also agree that campuses with high turnover rates often experience a loss of high quality teachers (Behrstock & Clifford, 2009; Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Papa & Baxter, 2008; Perrachione et al., 2008). Urban

schools with large numbers of lower socio-economic students and urban schools that are deemed low-performing are at a greater disadvantage because they have high rates of turnover and difficulty attracting new teachers to replace those who have left their positions (Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Papa & Baxter, 2008). According to Miller and Chait, "Challenging working conditions, poor human resource practices, and individual preferences represent major obstacles to staffing highpoverty schools with effective teachers" (2008, p. 8). A recent study by Hanushek and Rivkin (2010) shows a different perspective. The researchers found that teacher turnover in a high poverty Texas elementary school was not necessarily detrimental to student learning because the teachers leaving were deemed less effective per mathematics valueadded data analysis than the teachers remaining. The researchers asserted that the teachers who left were often replaced with novice teachers who also were likely to be less effective. However, the study did not address the disruption and low morale that can occur with high turnover on a campus or look at the impact of turnover on student achievement across the entire campus.

NCTAF (2007) conducted a study of teacher turnover costs in five districts in Illinois, Wisconsin, North Carolina and New Mexico. Results of the study indicate that both small and large districts experienced substantial costs in recruiting, hiring, and training replacement teachers. The cost per teacher ranged from \$4,366 in a rural district to \$17,872 in an urban district. The results also show the highest turnover rates in minority, high-poverty and low-performing schools. The study revealed that these struggling schools have to spend their funds on teacher-replacement related costs, leaving little funding for induction programs, specialists, and other avenues of support that affect student achievement.

Teachers' Intentions to Remain in the Classroom

The detrimental affects and steep costs caused by teacher turnover demonstrate a research need for the best hiring practices and assignment systems to support job satisfaction and teachers' intentions to remain in the classroom.

Good matches lead to job satisfaction and intentions to remain. Liu and Johnson (2006) showed that better matches and closer position fit lead to improved schools, teacher satisfaction, and teacher retention. Many researchers found that a good match between employee and job is an important factor in an employee's job satisfaction (Bretz & Judge, 1994; Dugonni & Ilgen, 1981; Liu, 2002; Liu & Johnson, 2006; McGraner, 2009; O'Reilly et al., 1991; Perrachione et al., 2008).

Job satisfaction is a particularly powerful influence on teachers' decisions to remain in their positions (Cohn, 1992; Johnson & Birkeland, 2003; Goodlad, 1984; Lee & Mowday, 1987; Lortie, 1975; Meek, 1998; Murnane, 1991; Perrachine et al., 2008). Teachers reported that a positive work experience strongly influences their job satisfaction, and intrinsic versus extrinsic motivators positively influence their job satisfaction and decisions to remain in the classroom (Cohn, 1992; Johnson & Birkeland, 2003; Lortie, 1975; Meek, 1998; Perrachine et al., 2008). Scholars found that teachers' decisions to stay are influenced by their sense of success in the classroom (an intrinsic motivator); therefore, a poor position match could lead to a compromised sense of

effectiveness, a lessened sense of success, and greater odds that a teacher will leave the teaching assignment or school (Johnson & Birkeland, 2003).

Hiring practices play a major role in matching employee to job. NCTAF study results showed that coherent human resource practices can reduce the negative effects of teacher turnover, saying "With accurate turnover and cost data, school leaders could better manage their human resources to achieve a higher return on their teaching investments" (2007, p. 3). Bretz, and Judge (1994) found that human resource practices convey information that affect decision-making processes and that applicant prehire perceptions of fit are consistent with posthire perceptions and subsequent retention. Other studies showed that a decentralized hiring system provides an opportunity for districts to make good matches between teacher and job because the candidate and employer(s) can exchange more and better information (Liu, 2002; Liu & Johnson, 2006).

Positive work outcomes lead to job satisfaction and intentions to remain.

Some scholars have used realistic job preview theory to consider the exchange of information between both job applicant and employer as an opportunity for each party to provide the other with comprehensive and accurate information—with the hypothesis that realistic job preview influences satisfaction and retention by improving an employees' ability to cope with the demands of the job (Dugonni & Ilgen, 1981; Liu & Johnson, 2006; McGraner, 2009). One study found that use of realistic job preview to reduce turnover was helpful but did not replace the need for good working conditions (Dugonni & Ilgen, 1981). Another study showed that teachers who have only a vague picture of the

campus and job demands prior to accepting a position are often surprised at aspects of their positions, and those surprises often lead to feelings of dissatisfaction and ineffectiveness, and eventually, intentions to leave (Liu & Johnson, 2006).

Scholars have found links between person-organization fit and work outcomes (Bowman, 2005; Bretz & Judge, 1994; Cable & Judge, 1996; Chatman, 1991; Judge et al., 2000; Liu, 2002; McGraner, 2009; O'Reilly et al., 1991; Rutledge et al., 2007). Cable and Judge (1996) found that employees who perceive a high level of person-organization fit are less likely to quit and more likely to recommend their organizations as a good place to work. Similarly, Chatman (1991) found that new employees whose values most closely matched those of their organizations felt the most satisfied, expressed intentions to remain, and actually do remain. Another study has important implications for schools because the results show that the best method to establish person-organization fit is through the interview, in which both the applicant and the employer have the opportunity evaluate congruence in values (Judge et al., 2000). Judge et al. found that the interview has greater effects on applicants if it is conducted over a lengthy period of time, includes multiple persons, and is conducted at the site of the organization (i.e., school campus).

Many study findings evidence that person-job fit positively affects work outcomes such as improved performance, increased satisfaction, and decreased turnover (Bowman, 2005; Bretz & Judge, 1994; Cable & Judge, 1996; Chatman, 1991; Kristoff, 1996; McGraner, 2009; O'Reilly et al., 1991; Rutledge et al., 2007). Bowman's study (2005) found that using elements of both person-organization fit and person-job fit benefits districts when the teacher selection is made at many levels; at the district level person-job fit is most helpful, and at the campus level person-organization fit is most helpful. Rutledge et al. found person-job fit to be most pertinent in identifying teachers with technical skills in content areas in order to meet teacher certification requirements (2007). They also identified person-group fit to affect work outcomes in teaching environments because of the involvement of team, grade-level, or subject area groups of teachers. Rutledge et al. demonstrated that teacher hiring involves aspects of person-organization fit, person-job fit, and person-group fit in that teaching involves interaction with numerous stakeholders at different levels of complexity.

An information-rich hiring process involves elements of preview and fit theories that allow for extended exchanges of information, resulting in positive work outcomes and job satisfaction (Dugonni & Ilgen, 1981; Judge et al., 2000; Liu & Johnson, 2006; McGraner, 2009). McGraner (2009) found that information-rich hiring strategies provide teacher applicants with adequate information about the district, campus, and community, thus allowing for informed decision making, increased job satisfaction, and improved retention—unlike information-poor processes, through which inadequate information exchange leads to incompatible matching of teacher to assignment, poor job satisfaction, and increased attrition.

Conclusions

A good literature review "allows the author not only to summarize the existing literature but also to synthesize it in a way that permits a new perspective" (Boote & Beile, 2005, p. 4). Reviewing the literature on teacher selection and assignment systems allowed the identification of major themes related to teacher hiring processes, contributors to positive work outcomes and the pivotal role that individual preferences play within hiring processes. The foundation for these emergent themes is the changing state of public education. The selection of quality teachers is nonnegotiable in past federal policy through the NCLB Act, and predicted to be nonnegotiable in upcoming federal educational policy reforms, through the reauthorization of the ESEA. In addition, examination of factors faced by hard-to-staff schools, often located in urban districts, is pertinent to the proposed goals of the ESEA that seek a more equitable distribution of effective teachers. Also, the \$5.3 billion in state budget cuts for education from both foundation and special program funds establishes a new fiscal reality in which resources are scarce and the investment in the selection and retention of teachers becomes an important budget consideration (House Bill 1, 2011).

1. Inefficient, outdated hiring systems actually impede the hiring of effective teachers. Late, rushed hiring timelines: (1) drain the applicant pool of more qualified, experienced, needed teacher applicants who accept positions elsewhere rather than wait for positions in districts that hire later in the school year (most common in urban districts); and (2) prevent necessary employer-applicant interaction and exchange of information that enables applicant teachers and district employers to make informed, appropriate hiring and assignment decisions (Claycomb, 2000; Darling-Hammond & Prince, 2007; Jacob, 2007; Lankford, Loeb, & Wyckoff, 2002). The research review revealed that the public education system's practices are outdated and unable to compete with current, assertive, effective recruiting efforts of the private sector (Behrstock &

Clifford, 2009; Hays & Behrstock, 2009; Hess, 2009; Johnson & Birkeland, 2003; Smith, 2009; Winter & Melloy, 2005; Young, 2008).

2. Turnover of effective teachers is detrimental to school campuses and districts. Teacher attrition from their assignments to other districts or to other professions leaves school campuses and districts in need of finding replacement teachers (Behrstock & Clifford, 2009; Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Papa & Baxter, 2008; Perrachione et al., 2008). The NCTAF (2007) presented a study indicating that both small and large districts experience substantial costs in recruiting, hiring, and training replacement teachers. In this 2007 study, the hiring cost per teacher ranged from \$4,366 in a rural district to \$17,872 in an urban district. Recent budget cuts to education as per the 82nd Texas Legislature's House Bill 1 (2011) leave the public education system with \$5.3 billion less to support teacher salaries and other instructional resources. Particularly high-poverty, low-performing schools cannot afford extra teacherreplacement costs (Guarino et al., 2006; Miller & Chait, 2008; NCTAF, 2007; Papa & Baxter, 2008). Districts are in dire need of making lasting investments in recruitment, selection, and assignment of effective teachers.

3. Personal values and preferences of both teacher applicants and school district employers lead to either compatible or incompatible matches of teacher to assignment. The literature review revealed that the selection process is rife with decision-making opportunities for applicants, administrators, and site-based committee members during the screening and employment phases, with tools such as the interview used to support decisions (Castetter & Young, 2000; Liu & Johnson, 2006; Seyforth, 2002; Smith, 2009; Sorenson & Goldsmith, 2009; Young, 2008). The individual choices made by teachers and administrators work in a collective manner to determine if the applicant is a good fit for the teaching assignment and campus (Boyd, et al., 2006; Liu & Johnson, 2006). These decisions lead to either good or poor fit of person to organization, person to job, and person to group (Bretz & Judge, 1994; Dugonni & Ilgen, 1981; Liu, 2002; Liu & Johnson, 2006; McGraner, 2009; O'Reilly, Chatman, & Caldwell, 1991; Perrachione et al., 2008).

4. Position fit leads to job satisfaction. Job satisfaction leads to teachers' intentions to stay and improved retention. The literature review affirmed that good job matches lead to job satisfaction, and that job satisfaction is a powerful influence on teachers' decisions to remain in the classroom (Cohn, 1992; Johnson & Birkeland, 2003; Goodlad, 1984; Lee & Mowday, 1987; Lortie, 1975; Meek, 1998; Murnane, 1991; Perrachine et al., 2008). Factors such as teacher selection and assignment systems, job fit and job preview theories each play a role in contributing to the fit of teachers to their to their assignments, campuses, and districts, and a compatible fit positively influences teachers' feelings of job satisfaction. Job satisfaction is a major contributor to teachers' intentions to remain in the classroom (Bretz & Judge, 1994; Dugonni & Ilgen, 1981; Liu, 2002; Liu & Johnson, 2006; McGraner, 2009; Miller & Chait, 2008; NCTAF, 2007).

5. Information-rich hiring processes support position fit and job satisfaction. An information-rich hiring process, a conceptual framework developed by Liu and Johnson (2006), incorporates adequate interaction and information-exchange between applicant and employer in the hiring process for both parties to make informed decisions. This

hiring process supports compatibility of teacher and position (position fit), supports job satisfaction because teachers have a clear pre-hire understanding of their role and expectations and are therefore able to succeed, and supports improved teacher retention.

These themes generate the foundation of this research study, which attempted to determine if the information richness of teachers' hiring experiences and perceptions of position fit play a role in predicting teachers' subsequent intentions to remain classroom teachers. Adequate information exchange (through information-rich hiring processes) and positive position fit of teacher to assignment should aid districts in hiring and assigning quality/effective teachers, result in improved teacher satisfaction and intentions to remain, increase retention of quality/effective teachers, and ultimately assist districts in meeting federal expectations.

Chapter Summary

The landscape of teacher selection and assignment in public school districts contains elements that include: policy factors that affect teacher hiring practices; teacher recruitment strategies necessary for quality selection systems; components of selection that include purpose, stages, tools, theories, and hiring preferences; challenges to selection systems in school districts, particularly at urban campuses; detrimental effects of, reasons for, and methods of reducing teacher attrition; and factors that influence teachers' decisions to remain in the classroom.

This chapter contains an extensive review of available literature pertaining to the elements that create the landscape of teacher selection and assignment systems.

Furthermore, in this chapter, major themes that emerged through the literature review were analyzed.

Chapter Three: Method

This nonexperimental study used a quantitative methodology and a postpositivist theoretical paradigm to examine associations between variables. The study design may also be referred to as *ex-post facto* because no experimental manipulations were performed and the independent variables were already in place at the time of the study.

The purpose of this study was to examine how newly hired teachers at Texas campuses (a) perceived the information richness of their hiring experiences, and (b) perceived the fit with their classroom and campus assignments, and to use this information to determine if there are any associations between these two perceptions and the teachers' intentions to remain in the classroom.

Conceptual Framework, Research Questions, and Null Hypotheses

Chapter One of this study describes a conceptual framework that involves hypothetical constructs based on hiring theories used in teacher selection and assignment systems: realistic job preview theory; human resource management theory; person-job fit, person-organization fit, and person-group fit theories, and; two-sided matching theory. This study was based on a conceptual construct first introduced by Liu and Johnson (2006) that incorporates elements of these teacher-hiring theories as part of an information-rich hiring process. An information-rich hiring process provides for a sufficient exchange of information between teacher applicant and employer(s) so that both parties gain knowledge of potential match in job skills and school culture, and ultimately to ensure that the teacher hired is a good position fit, or match with the teaching assignment and campus. Two research questions emerged from the hypothetical constructs:

- 1. Does an information-rich hiring process predict teachers' intentions to remain in the classroom?
- 2. Does position fit predict teachers' intentions to remain in the classroom? Further, the following two null hypotheses were tested:
 - Ho: An information-rich hiring process does not predict teachers' intentions to remain in the classroom.
 - Ho: Position fit does not predict teachers' intentions to remain in the classroom.

Research Design

According to Leedy and Ormrod an ex-post facto design "can provide an alternative means by which a researcher can investigate the extent to which specific independent variables may possibly affect the dependent variables of interest" (2005, p. 232). In this ex-post facto study, a nonexperimental approach was appropriate because I examined associations between events that have already occurred, including the administration of a survey instrument.

Kerlinger and Lee denote value in nonexperimental research because experimental inquiry is not appropriate for many of the research problems in education. Therefore, an advantage to this nonexperimental study was that it allowed for "controlled inquiry" into a research problem in education (2000, p. 569).

According to Kerlinger and Lee, nonexperimental research has three major weaknesses: (a) the inability to manipulate independent variables, (b) the lack of power to randomize, and (c) the risk of improper interpretation (2000, p. 568). Therefore, limitations of this study are that cause and effect cannot be assumed because the researcher is limited to after-the-fact data, without the ability to manipulate treatment.

Variables

Two continuous independent variables (not experimentally manipulated) were identified as an information-rich hiring process and position fit. The dependent variable (categorical) was identified as teachers' intentions to remain in the classroom.

The independent variable of an information-rich hiring process was represented by responses to Teacher Survey Questions $\sum (12a-12j)$. The independent variable of position fit was represented by responses to Teacher Survey Questions $\sum (13a-13e)$ and 14a-14f). Individual teacher response was the unit of analysis that represented the degree of information richness and position fit experienced by those recently hired and assigned to campuses.

The dependent variable was represented as a binary variable by responses to Teacher Survey Question 10: (1a) anticipates remaining in the classroom, or (2a) does not anticipate remaining in the classroom. See Appendix B for the complete list of survey questions.

Description of Population and Sample

According to Mertens, "An operational definition of the sample in the postpositivist paradigm is called the *experimentally accessible population*, defined as the list of people who fit the conceptual definition" (2005, p. 309). The population in this

study consisted of a total of 6,505 teachers employed at 92 campuses located across 13 Texas school districts. From the population of 6,505 teachers, a purposively selected sample included 1,430 teachers hired for new assignments within two academic years (2008–09 and 2009–10) at the 92 campuses. Mertens stated the following:

Although randomized probability samples are set forth as the ideal in the postpositivist paradigm, they are not commonly used in educational and psychological research. Thus, in practice, the postpositivist and constructivist paradigms are more similar than different in that both use nonrandom samples. (2005, p. 308)

A survey instrument was administered to all 1,430 teachers in the spring of 2010. Individual teacher response was used as the unit of analysis to examine indicators of information richness and position fit.

The population and sample used for this study were part of a larger employment research project conducted by the Center for Research, Evaluation, and Advancement of Teacher Education (CREATE) and the Texas Association of School Administrators (TASA). This larger employment study administered a separate survey instrument to three samples that included the survey to 1,430 newly hired teachers, another survey to 92 principals, and a last survey to 272 faculty members at university-based teacher preparation programs. The executives directors of CREATE and TASA sent invitations to participate in the larger study to superintendents of 27 school districts within the state. Invited districts were selected to assure geographic diversity, as well as to assure student economic and ethnic distributions were representative of the state at large. Of the 27 districts initially invited to participate, 13 accepted. Districts were classified per Texas Education Agency (TEA) Snapshot of School District Community Type 2008 criteria, available to the public on the TEA website, and include: five Major Urban; four Major Suburban; two Central City; one Other City Central Suburban; and one Independent Town. See Table 1 for more explicit attributes of each district type.

According to information obtained from the TEA Snapshot 2008 District Detail reports, also accessible on the TEA website, participating districts collectively included 863 campuses, enrolling approximately 591,669 students, or 12.5% of total statewide enrollment. Of 591,669 students in participating school districts, 19% were African-American, 19% were Anglo, and 57.9% were Hispanic.

Of the total 863 campuses in participating districts, 92 campuses participated in the study. To establish participating campuses, the researchers reviewed campus academic performance reports to identify the highest and lowest performing elementary, middle, and high school campuses in each of the 13 districts. Campus academic performance refers to the percentage of students passing all tests at all campus grade levels as reported in the TEA's Academic Excellence Indicator System (AEIS), available on TEA's website. The number of highest and lowest performing campuses was then increased proportionally within the largest of the participating districts. The 92 campuses purposively selected to participate included: 29 high schools, 33 middle schools, and 30 elementary schools.

Table 1TEA Snapshot of School District Community Type

Major Urban The largest school districts in the state that serve the six metropolitan areas of Houston, Dallas, San Antonio, Fort Worth, Austin, and El Paso. Major urban districts are the districts with the greatest membership in counties with populations of 735,000 or more, and more than 35 percent of the students are identified as economically disadvantaged. In some cases, other size threshold criteria may apply.

Major Suburban Other school districts in and around the major urban areas. Generally speaking, major suburban districts are contiguous to major urban districts. If the suburban district is not contiguous, it must have a student population that is at least 15 percent of the size of the district designated as major urban. In some cases, other size threshold criteria may apply.

Other Central City The major school districts in other large, but not major, Texas cities. Other central city districts are the largest districts in counties with populations between 100,000 and 734,999 and are not contiguous to any major urban districts. In some cases, other size threshold criteria may apply.

Other Central City Suburban Other school districts in and around the other large, but not major, Texas cities. Generally speaking, other central city suburban districts are contiguous to other central city districts. If the suburban district is not contiguous, it must have a student population that is at least 15 percent of the largest district enrollment in the county. Its enrollment is greater than 3 percent of the contiguous other central city district. In some cases, other size threshold criteria may apply.

Independent Town The largest school districts in counties with populations of 25,000 to 99,999. In some cases, other size threshold criteria may apply.

Procedures for Data Collection

At the onset of the larger CREATE and TASA employment study, leadership

established constituents of a research design team that included: (a) team co-chairs, (b)

project directors, (c) co-principal investigators, and (d) associate investigators. In

addition, the design team included designated liaisons from each of the 13 participating

Texas school districts as research coordinators. The author of this study participated in

this larger employment study that administered three separate surveys as a member of the

design team, in the role of associate investigator. The research design team submitted an Institutional Review Board (IRB) approval to the Tarleton State University Human Subjects Review Team, and approval was obtained. The IRB approval is included in Appendix C of this study.

This study utilizes one of the three survey instruments included in the larger employment study. The teacher survey instrument used for collecting data for this study as appears in Appendix B was a replication (with some amendments) of a survey questionnaire designed by and used with the permission of Susan Moore Johnson and her team for the Harvard Project on the Next Generation of Teachers called Survey of First-Year and Second-Year Teachers (2002). The original survey had four sections. The first section included general information, and the content of section one was included in the survey instrument with the exception of questions regarding charter schools (there were no charter school participants in the current sample of campuses). The second section included questions about the hiring process, and all content was included in the current survey instrument. The third section included questions about professional culture and was omitted from this survey instrument. The fourth section included questions about background information and was included in this survey instrument. The design team held three research meetings and used professional judgment to make sure the wording of the survey was consistent. The design team updated the question choices with district advances in technology.

The sample teachers participated in the electronic survey questionnaire in spring of 2010. Responses were returned to a secure database housed at CREATE. Identifying

data were stripped and replaced by arbitrary numerical codes in order to maintain anonymity. Written permission was obtained from CREATE to use the dataset from the teacher survey instrument for the purposes of this dissertation research study. The CREATE permission letter is included in Appendix D of this study.

The University of Texas Office of Research Support was then contacted to request an exemption from IRB review because this research study used existing data (teacher survey results). The exemption status was approved and is included in Appendix E of this study.

Instrument. In this study, the sample was administered an electronic survey instrument using Survey Monkey[™] software. The survey addressed aspects of employment at the campus from the teacher's viewpoint, including: (a) descriptions of campus-level selection procedures, (b) appraisal of campus-level selection procedures, and (c) appraisal of the degree of campus and job fit (position fit).

Teachers were asked to rate statements (using a 1–7 scored Likert scale) that indicate the degree to which they received an accurate picture of their job and campus from their recent hiring experience. The Likert scale offered seven rating choices with (1) representing *Strongly Disagreed* and moving along a continuum ending in (7) representing *Strongly Agreed*. The ten statements rated for an information-rich hiring process included survey instrument questions 12a–12j, and began with the stem: *From the hiring process, I got an accurate picture of*:

(a) what the <u>teachers</u> were like at the campus and whether I might enjoy working with them.

- (b) what the <u>students</u> were like at the campus and whether I might enjoy working with them.
- (c) the principal's leadership style.
- (d) the curriculum I would be teaching.
- (e) what my teaching assignment would be (i.e. subjects, grade levels, number of classes, other duties).
- (f) the support that the campus would provide to me as a new teacher.
- (g) how much autonomy I would have as a teacher at the campus.
- (h) the opportunities I might have to help make important campus-wide decisions.
- (i) the educational philosophy of the campus.
- (j) the non-classroom duties I would be required to perform.

Teachers were also asked to rate statements (using a 1–5 scored Likert scale) that indicate the degree to which they experienced a position fit. The Likert scale offered five rating choices with (1) representing *Very Poor Match* and moving along a continuum ending in (5) representing *Very Good Match*. The eleven statements rated for position fit included survey instrument questions 13a–13e, and 14a–14f. Survey questions 13a–13e began with the stem: *How closely would you say that your current teaching assignment matches:*

- (a) your subject matter knowledge and expertise.
- (b) your subject matter interests.
- (c) other skills and talents that you have (e.g. coaching sports, organizing

extracurricular activities, or advising students).

- (d) the grade level(s) that you would prefer to teach.
- (e) the type of student population you would prefer to teach.

Survey questions 14a–14e began with the stem: *How closely would you say that your campus matches:*

- (a) your own educational philosophy.
- (b) the amount of autonomy you would like to have as a teacher (i.e., over what and how much to teach).
- (c) your own views on student discipline.
- (d) the amount of collaboration or teamwork you would like with colleagues.
- (e) the amount of input (or influence) you would like to have on campus-wide decisions.
- (f) the amount of input (or influence) you would like to have on department or grade-level decisions.

Teachers were also asked a multiple-choice question in Survey Question 10: Which of the following best describes how you view your teaching job?

- (a) I will most likely remain a classroom teacher for the rest of my career.
- (b) I most likely will leave classroom teaching at some point, but I plan to stay in the field of education for the rest of my career.
- (c) I most likely will leave classroom teaching at some point, and I plan to work in another job(s) outside the field of education for the rest of my career.

Data Analysis

First, statements of null hypothesis were identified in terms of quantified measurements. The accepted level of significance for rejecting or retaining a null hypothesis was established at $\alpha = .05$. In order to identify the independent effect of position fit and information-rich hiring practices on the intentions of newly hired teachers to remain in the classroom, logistic regression was selected as the appropriate type of statistical analysis for this study, because there was only one dependent variable, and it involved binary data. According to Agresti (2007), the logistic regression model is a generalized linear model that is most popular for use with binary data; "The random component for the (success, failure) outcomes has a binomial distribution" (p. 71).

Next, SPSS software was used to conduct the logistic regression analysis, which was used to predict a categorical dependent variable on the basis of two continuous independent variables. The analysis estimated the odds that a teacher would intend to remain in the classroom by fitting data into a logit function. The dependent variable was a binary indicator of whether the teacher answered that his or her future intention was to remain in the classroom as represented by response to Teacher Survey Question number ten. The two independent variables were (a) degree of information richness of the hiring process as measured by responses to Teacher Survey Questions 12a–12j and (b) position fit of the teacher with their classroom assignment and campus assignment as measured by responses to Teacher Survey Questions 13a–13e and 14a–14f.

According to Agresti, an assumption for this type of model is that there are no issues of multicollinearity "correlations among predictors making it seem that no one variable is important when all the others are in the model" (p. 138). Multicollinearity was not an issue with this research as two separate analyses were conducted, one with information richness as the independent variable and the other with position fit as the independent variable.

In order to establish controls and reduce extraneous influences the individual teacher characteristics collected in the survey that might relate to a teacher's decision to remain in the classroom were controlled for. Specifically, the age, gender, and experience of the teacher were included. Because the relationship between age and leaving the teaching profession is non-linear, both the age and age-squared were included. Also, a binary variable indicating whether the teacher was female was included. Finally, a binary variable was included that indicated whether the teacher was a novice teacher—a teacher with two or fewer years of experience.

Also, the school level of the campus in which the teacher was employed was controlled for in case the school level was associated with the teacher's intentions to remain in the classroom. A binary variable indicating if the school was a middle school and another binary variable indicating if the school was a high school were included. The category of elementary schools was omitted so that it could be used as a reference category in the logistic regression analysis against which to compare middle and high school.

Finally, because school characteristics are strongly associated with teacher attrition, school fixed-effects were included. School fixed-effects control for the unobserved school characteristics that may be associated with a teacher's intentions to remain in the profession.

Descriptive statistics in terms of percentages were produced that show teachers' intentions to remain or leave the classroom. In addition, descriptive statistics were prepared in terms of percentages to show the degree to which respondents experienced an information-rich hiring process according to responses on the ten Likert scale statements, and to show the degree to which respondents experienced position fit according to responses on the 16 Likert scale statements.

Chapter Summary

In this chapter, a detailed description of the design, sample, and procedures used to conduct this study was provided. This quantitative study used an ex-post facto design and nonexperimental approach to examine relationships between an information-rich hiring process, position fit, and teachers' intentions to remain in the classroom. The purpose of this study was to examine how newly hired teachers at Texas campuses perceived the information richness of their hiring experiences and perceived the fit with their assigned positions in order to determine if predictive relationships exist. Chapter Four of this study reports the statistical findings of the data. Chapter Five presents further findings and conclusions and proposes areas for further research suggested by the results of this research study.

Chapter Four: Analysis of Data

This survey-based research study examined how newly hired teachers at Texas campuses (a) perceived the information richness of their hiring experiences and (b) perceived the fit with their classroom and campus assignments, for the purpose of determining associations between these two perceptions and the surveyed teachers' intentions to remain in the classroom. Two research questions guided this study:

- 1. Does an information-rich hiring process predict teachers' intentions to remain in the classroom?
- 2. Does position fit predict teachers' intentions to remain in the classroom?

Information was assessed by individual response of newly hired Texas teachers to a survey instrument developed as part of a larger employment research project conducted by CREATE and TASA. The sample included 1,430 newly hired teachers at 92 campuses located across 13 Texas school districts. Of the 1,430 newly hired teachers, 761 completed the survey for a return rate of 53.2%.

For the purpose of this study, I identified from the data set two independent variables—(a) information-rich hiring practices and (b) position fit. I designated the dependent variable as teachers' intentions to remain in the classroom.

The following two null hypotheses were tested using logistic regression analysis:

- 1. Ho: An information-rich hiring process does not predict teachers' intentions to remain in the classroom.
- 2. Ho: Position fit does not predict teachers' intentions to remain in the classroom.

Regression Analysis

The accepted level of significance for rejecting or retaining a null hypothesis was established at $\alpha = .05$. Results of the null hypothesis testing are included in the results section of the corresponding hypothesis.

In the regressions performed, gender, age, and school level were never statistically associated with teachers' intentions to remain in the profession. For information rich and position fit (see logistic regression analysis results for hypotheses 1 and 2) a significant statistical difference at 0.000 was observed in novice teachers' intentions to remain in the classroom. This result indicates an association between teachers who had two or less years of experience that indicated intention to remain in the classroom, and the degree to which they experienced an information-rich hiring experience and position fit with their classroom and campus assignments.

Results for hypothesis 1. *Perceived information-rich hiring practices predict teachers' intentions to remain in the classroom.* Hypothesis 1 was examined using logistic regression. When entered individually into the logistic regression equation, the average of information-rich practices was statistically significantly associated with a teacher's intentions to remain a classroom teacher. Specifically, for every one-point increase in the average of the information-rich practice statements, a teacher was 3.4% more likely to anticipate remaining in the profession (see Table 2). This result indicates that the more information rich the hiring experience for the newly hired teacher, the more likely the teacher was to anticipate remaining in the classroom.

Logistic Regression: Information-Rich Practices as Predictor of Teachers' Intentions to Remain in the Classroom

Variable	В	S.E.	Wald	Sig.	Exp(B)
Age	-0.009	0.081	0.011	0.915	0.991
Age Squared	0.001	0.001	0.900	0.343	1.001
Female	0.176	0.203	0.758	0.384	1.193
Novice	0.850	0.219	15.079	0.000	2.339
Sch Level: MS	0.084	1.598	0.003	0.958	1.088
Sch Level: HS	-0.246	1.273	0.038	0.846	0.782
Info Rich Practices Avg.	0.033	0.008	17.468	0.000	1.034

Note: B=coefficient for the constant; S.E.=standard error; Wald=Wald chi-square test; Sig.=statistical significance; Exp(B)=exponentiation of the B coefficient

Null Hypothesis 1 stated, "Information-rich hiring practices do not predict teachers' intentions to remain in the classroom." The threshold was set at .05; and the significance level for information-rich practices was 0.000. This null hypothesis was rejected, as the difference was statistically significant.

Results for hypothesis 2. *Perceived position fit predicts teachers' intentions to remain in the classroom.* Hypothesis 2 was examined using logistic regression. As with information-rich practices, entered individually into the logistic regression equation, the average of position fit was statistically significantly associated with a teacher's intentions to remain a classroom teacher. Specifically, for every one-point increase in the average position fit statements, a teacher was 1.2% more likely to anticipate remaining in the classroom (see Table 3). These results indicate that a higher degree of satisfaction with the position fit of his or her classroom and campus assignments was associated with intentions to remain in the classroom.

Variable B S.E. Wald Sig. Exp(B) 0.227 0.634 0.962 -0.039 0.081 Age Age Squared 0.001 0.001 1.674 0.196 1.001 Female 0.509 0.134 0.203 0.435 1.144 Novice 0.885 0.222 0.000 2.424 15.953 Sch Level: MS 0.445 1.593 0.078 0.780 1.560 Sch Level: HS 0.082 1.288 0.004 0.949 1.085 28.505 1.079 Position Fit Avg. 0.076 0.014 0.000

Logistic Regression: Position Fit as Predictor of Teachers' Intentions to Remain in the Classroom

Note: B=coefficient for the constant; S.E.=standard error; Wald=Wald chi-square test; Sig.=statistical significance; Exp(B)=exponentiation of the B coefficient

Null Hypothesis 2 stated, "Position fit does not predict teachers' intentions to remain in the classroom." The threshold was set at .05, and the significance level for position fit was 0.000. This null hypothesis was rejected, as the difference was statistically significant.

Descriptive Statistics

A total of 346 (48.2%) of respondents indicated intent to leave the classroom, although a percentage of those intending to leave, 282 (39.3%) would remain in the education field in another role. A slightly higher number of respondents 372 (51.8%) indicated intent to remain in the classroom (Table 4). These results indicate that those teachers intending to remain in the classroom may have been positively influenced by a satisfactory hiring process and by their classroom and campus assignments. Only a small percentage 64 (8.9%) gave indication that they would leave teaching to do something outside the field of education.

Future Intentions of Classroom Teachers

Future Intentions	Ν	%
Most likely will leave teaching	64	8.9
Will leave classroom but will stay in education	282	39.3
Most likely remain in classroom	372	51.8
Total	718	100.0

Teacher respondents were also asked to rate ten statements (using a 1–7 scored Likert scale) that indicated the degree to which they received an accurate picture of their job and campus from their recent hiring experience. The Likert scale offered seven rating choices with (1) representing *Strongly Disagreed* and moving along a continuum ending in (7) representing *Strongly Agreed*. Table 5 includes the rating results of information richness.

On one end of the spectrum, respondents most Strongly Agreed that they got an accurate perception of the education philosophy of the campus, the principal's leadership style, and what their teaching assignment would be. On the other end of the spectrum, the respondents most Strongly Disagreed that they got an accurate picture of what the teachers were like at the campus, the input they would have to make campus-wide decisions, and the non-classroom duties they would be required to perform. Out of all ten statements of information richness, input to make campus-wide decisions received the highest percentage, 219 (30.6%) of the Neutral rating.

Many statements received high over-all percentages along the *Agree* spectrum (Somewhat Agree, Agree, Strongly Agree) including getting an accurate picture of what

the other teachers were like (56.3%), what the students were like (61.1%), the curriculum they would be teaching (69.1%), and the support provided to them as a new teacher (56.8%).

Table 5

Degree of Information-Rich Hiring Practices

Statement "I got an accurate picture of"	N %	Strongly Disagree	Disagree	Somewhat Disagree	Neutral	Somewhat Agree	Agree	Strongly Agree
What teachers	N	50	76	45	142	121	177	105
were like at the campus	%	7.0	10.6	6.3	19.8	16.9	24.7	14.7
What students	N	44	60	45	129	165	171	102
were like at the campus	%	6.1	8.4	6.3	18.0	23.0	23.9	14.2
The principal's	N	47	43	37	87	146	200	156
leadership style	%	6.6	6.0	5.2	12.2	20.4	27.9	21.8
The curriculum	N	39	42	49	92	135	221	138
I would be teaching	%	5.4	5.9	6.8	12.8	18.9	30.9	19.3
What my teaching	N	29	16	43	54	122	228	224
assignment would be	%	4.1	2.2	6.0	7.5	17.0	31.8	31.3
How much support the	N	40	42	56	100	141	206	131
campus would provide me as a new teacher	%	5.6	5.9	7.8	14.0	19.7	28.8	18.3
How much autonomy I	N	42	45	46	168	154	184	77
would have as a teacher	%	5.9	6.3	6.4	23.5	21.5	25.7	10.8

Table 5 continued.

The input I would	N	53	58	85	219	144	108	49
have to make campus- wide decisions	%	7.4	8.1	11.9	30.6	20.1	15.1	6.8
The educational	N	30	33	39	99	147	216	152
philosophy of the campus	%	4.2	4.6	5.4	13.8	20.5	30.2	21.2
The non- classroom	N	52	66	72	165	145	136	80
duties I would be required to perform	%	7.3	9.2	10.1	23.0	20.3	19.0	11.2

Note. Table figures are based on the survey responses of the sample.

Teachers were also asked to rate eleven statements (using a 1–5 scored Likert scale) that indicated the degree to which they experienced position fit with their teaching and campus assignments. The Likert scale offered five rating choices with (1) representing *Very Poor Match* and moving along a continuum ending in (5) representing *Very Good Match*. Table 6 includes the rating results of position fit.

The first question asked the teacher to rate how closely their *teaching* assignment matched with five different statements. The majority of respondents indicated a positive match with their teaching assignment. Along the rating range of Good and Very Good, 90.7% felt their subject matter knowledge and expertise matched their assignment, 87.9% felt their subject matter interest matched their assignment, and 81.4% were teaching at a preferred grade level. Along the rating range of Very Poor to Poor, the percentages were much lower with 6.5% responding that they were not teaching the type of student

population that they preferred to teach, and only 0.8% indicating a poor match with their assignment and subject matter knowledge and expertise.

The second question asked the teacher to rate how closely their *campus* assignment matched with six different statements. Responses to the second question were more spread out along the rating spectrum. Along the rating range of Good and Very Good, 65.7% responded that there was a good match between their own education philosophy and that of the campus and, 72.2% felt there was a good match between the amount of autonomy they preferred and the amount they received at the campus. There was a wider range of responses regarding a match between respondents' personal views on student discipline and the campus view with 18.5% rating this statement in the Very Poor to Poor range and, 53.6% rating this statement in the Good to Very Good range. This wider distribution of ratings was also shown when the respondent was asked to rate the match between the amount of collaboration they preferred to have with others and the collaboration that actually took place on the campus, with 14.8% rating this statement in the Very Poor to Poor range and 61.9% rating the statement in the Good to Very Good range. Another statement involving the amount of input the teacher would like to have on campus-wide decisions showed 15.8% rating this statement in the Very Poor to Poor range and 51.6 % rating the statement in the Good to Very Good range.

Degree of Position Fit

Statement	N					
How closely would you say that your current <u>teaching assignment</u> matches the following	%	Very Poor	Poor	Moderate	Good	Very Good
Your subject matter knowledge	Ν	1	5	62	180	481
and expertise	%	0.1	0.7	8.5	24.7	66.0
Your subject matter interests	Ν	2	13	73	176	465
Tour subject matter merests	%	0.3	1.8	10.0	24.1	63.8
Other skills and talents you have	Ν	9	30	140	246	304
Other skins and talents you have	%	1.2	4.1	19.2	33.7	41.7
Grade level you profer to teach	Ν	8	19	109	223	370
Grade level you prefer to teach	%	1.1	2.6	15.0	30.6	50.8
Type of student population you	Ν	15	32	135	220	327
prefer to teach	%	2.1	4.4	18.5	30.2	44.9
Statement How closely would you say that your <u>campus</u> matches the following	N %	Very Poor	Poor	Moderate	Good	Very Good
Your own education philosophy	Ν	15	58	177	250	229
Tour own education philosophy	%	2.1	8.0	24.3	34.3	31.4
The amount of autonomy you	Ν	14	33	156	292	234
would like as a teacher	%	1.9	4.5	21.4	40.1	32.1
Your views on student discipline	Ν	51	84	203	226	165
Tour views on student discipline	%	7.0	11.5	27.8	31.0	22.6
The amount of collaboration you	Ν	28	80	170	225	226
would like with others	%	3.8	11.0	23.3	30.9	31.0
The amount of input you would	Ν	30	85	238	242	134
like on campus-wide decisions	%	4.1	11.7	32.6	33.2	18.4
Amount of input you would like	Ν	25	64	197	256	187
on dept/grade level decisions	%	3.4	8.8	27.0	35.1	25.7

Note. Table figures are based on the survey responses of the sample.

Chapter Five: Discussion, Conclusions, and Recommendations for Future Research

This chapter includes a summary of findings, discussion and conclusions, limitations, significance, recommendations for practice, and recommendations for further inquiry. The purpose of this study was to examine how newly hired teachers at Texas campuses (a) perceived the information richness of their hiring experiences, and (b) perceived the fit with their classroom and campus assignments, and to use this information to determine if there are any associations between these two perceptions and the teachers' intentions to remain in the classroom.

Information was assessed by individual teacher response to a survey instrument developed as part of a larger employment research project conducted by CREATE and TASA. The sample included 1,430 newly hired teachers at 92 campuses located across 13 Texas school districts. Of the 1,430 newly hired teachers, 761 completed the survey for a return rate of 53.2%.

Summary of Results

The following two research questions guided the research process:

1. Does an information-rich hiring process predict teachers' intentions to remain in the classroom?

Does position fit predict teachers' intentions to remain in the classroom?
 Results showed the following specific findings.

1. A significant statistical difference was observed in teachers' intentions to remain in the classroom when information-rich hiring practices was entered individually into the logistic regression equation. Teachers who reported having experienced higher levels of information-rich hiring were more likely to anticipate that they would remain in the classroom.

- 2. A significant statistical difference was observed in teachers' intentions to remain in the classroom when position fit was entered individually into the logistic regression equation. Teachers who reported having experienced higher levels of position fit with their current classroom assignment and campus were more likely to anticipate that they would remain in the classroom.
- 3. A significant statistical difference was observed in novice (1–2 years of experience) teachers' intentions to remain in the classroom when information rich was entered individually into the logistic regression equation. Novice teachers who reported having experienced higher levels of information-rich hiring were more likely to anticipate that they would remain in the classroom.
- 4. A significant statistical difference was observed in novice (1–2 years of experience) teachers' intentions to remain in the classroom when position fit was entered individually into the logistic regression equation. Novice teachers who reported having experienced higher levels of position fit with their current classroom assignment and campus were more likely to anticipate that they would remain in the classroom.

Discussion and Conclusions

The results of this quantitative study showed that (a) teachers who experience higher levels of information-rich hiring practices and (b) teachers who experience positive position fit with their teaching assignment and campus were more likely to anticipate remaining in the classroom. This is important because of the detrimental effects and steep costs incurred by schools and districts as a result of teacher turnover. The results support the results of other studies performed in the area of hiring practices, most notably that of Liu and Johnson (2006) who showed that better matches and closer position fit lead to improved schools, teacher satisfaction, and teacher retention.

An interesting finding in this study was that novice teachers who (a) experience higher levels of information-rich hiring practices and (b) experience positive position fit with their teaching assignment and campus were more likely to anticipate remaining in the classroom. Perhaps the novice teachers had recently completed their teacher preparation programs, and were fresh and eager to begin their nascent teaching careers. Enthusiasm for their new careers combined with positive hiring experiences and high levels of job satisfaction may have been key factors in their indications that they would remain in the classroom. It is also informative for principals and others involved in the teacher hiring process to know that if time and care are put forth into matching a novice teacher with a classroom and campus assignment, the likelihood of retaining those teachers may be increased. In addition, if time and effort were invested in matching novice teachers to assignments, employers would be helping to establish an environment that is conducive for effective teaching. When novice teachers are immediately assigned to hard-to-staff schools, without regard to their skills and interests, they face a hurdle to effective teaching from the onset due to potential mismatch.

102

The descriptive charts in Chapter Four of this study showed ratings that the teacher respondents assigned to different aspects of the hiring process and position fit. These results and the statements themselves could be used by principals to review their own hiring systems and ensure that communication with teacher candidates includes information about what their role on the campus would entail such as how they would be supported as a new teacher, the amount of autonomy they would have, the non-classroom duties they would be expected to perform, etc. The same could be done with statements involving position fit in the descriptive charts. Communicating with a teacher candidate about the potential match of their skills, interests, and expectations before the candidate is hired would be beneficial to both the employer and the candidate.

Limitations

This study had limitations that require acknowledgment, but are not serious enough to threaten the validity of the conclusions. The first limitation was that the data was not generalizable to all populations because the sample population was not random. However, in Chapter Three, details were provided that emphasize the total context of the sample population and how the sample was selected, thus giving readers the information necessary to determine the transferability of the findings for their own purposes.

The second limitation was that there is not a professional consensus on the definition of "effective" or "quality" teacher. But we do know that teachers are individuals who have unique attributes, skills, experiences, and work expectations that may or may not make them a good fit for a particular teaching assignment or campus. The results of this study show that even if definitions of the ideal teacher candidate may

differ, information-rich hiring processes and the position fit of teacher with his or her classroom assignment and campus influence a teacher's intentions to remain in the classroom.

The third limitation was reliance on self-reported data through the administration of the electronic survey instrument. Throughout the sample, examination of survey responses yielded evidence that information-rich practices and position fit were associated with teachers' intentions to remain in the classroom. The self-reported responses appeared to be consistent, reliable, and consistent with other study data. There was no reason to suspect dishonesty on part of the participants.

The fourth limitation was that *ex-post facto* methodology did not allow for control over the sample. Despite this limitation, the *ex-post facto* design allowed the time to focus research efforts on analyzing data already collected from a diverse and large number of teachers.

The fifth limitation was that this study was limited to after-the-fact data. Without the ability to manipulate treatment, cause and effect could not be assumed.

Significance and Recommendations for Practice

The results of this study showed links between teacher selection and assignment systems, hiring tools and methods, and teachers' intentions to remain in the classroom. These findings provide timely information to school district leadership as it prepares for the new policy trends anticipated with the reauthorization of the ESEA and as it works to increase teacher retention in order to improve campus stability and minimize teacher replacement costs.

In addition, these study results provide school leadership with information that can improve their human resource practices and support efforts to implement The Interstate School Licensure Consortium Educational Leadership Policy Standards that designates those in leadership positions with the responsibility of staffing and retaining quality teachers.

Furthermore, the findings of this study can contribute significantly to the knowledge base of university superintendent-, principal-, and teacher-preparation programs, thus preparing future school district leadership to improve teacher selection and assignment systems, as well as to prepare teacher candidates for successful navigation of current hiring environments, thus making informed decisions that will lead to job satisfaction and job stability.

Recommendations for Further Inquiry

This study investigated two research questions:

- 1. Does an information-rich hiring process predict teachers' intentions to remain in the classroom?
- 2. Does position fit predict teachers' intentions to remain in the classroom?

Statistically significant findings showed that (a) teachers who reported having experienced higher levels of information-rich hiring and (b) teachers who reported having experienced position fit were more likely to anticipate that they would remain in the classroom. Another statistically significant finding showed that novice teachers who reported having experienced higher levels of information-rich hiring and position fit were more likely to anticipate that they would remain in the classroom. As a follow up to the findings of this study, recommendations for further inquiry include:

1. Study of specific hiring practices to determine which tools generate information richness, as perceived by teacher candidates and district employers, to be conducted during the hiring phase, so as to collect immediate perceptions of those doing the hiring and applicants who are or are not hired. Isolating which hiring tools are most effective, as perceived by both employer and applicant, could further lead districts in improving their hiring practices.

2. Study of schools and districts that hold notably higher rates of quality teacher retention, to determine school/district characteristics and practices that yield positive job satisfaction and decreased attrition. Information could assist districts and schools in making minor changes to culture (etc.) in order to improve teacher satisfaction and retention.

3. Another study of the relationship between information-rich hiring practices, position fit, and teachers' intention to remain in the classroom, to determine the specific relationship among the three. This study further confirmed that information-rich hiring practices and position fit improve teacher intention to remain in the classroom, but it did not isolate the direct relationship among these three variables. I hypothesize that in this relationship, information-rich hiring practices support position fit which in turn supports teacher retention.

4. A study of "best practices" for teacher hiring that could be used as a foundation for the professional development of principals, and others involved in the campus-level hiring decisions.

This study and future studies will ultimately benefit students by supporting the hiring and retention of quality/effective teachers throughout the continuous evolvement of education policy environments and expectations.

Appendix A: InTASC Core Teaching Standards



InTASC Model Core

Teaching

Standards April 2011

Standard #1: Learner Development

The teacher understands how learners grow and develop, recognizing that patterns of learning and development vary individually within and across the cognitive, linguistic, social, emotional, and physical areas, and designs and implements developmentally appropriate and challenging learning experiences. PERFORMANCES

1(a) The teacher regularly assesses individual and group performance in order to design and modify instruction to meet learners' needs in each area of development (cognitive, linguistic, social, emotional, and physical) and scaffolds the next level of development.

1(b) The teacher creates developmentally appropriate instruction that takes into account individual learners' strengths, interests, and needs and that enables each learner to advance and accelerate his/her learning.

1(c) The teacher collaborates with families, communities, colleagues, and other professionals to promote learner growth and development. <u>ESSENTIAL KNOWLEDGE</u>

1(d) The teacher understands how learning occurs--how learners construct knowledge, acquire skills, and develop disciplined thinking processes--and knows how to use instructional strategies that promote student learning.

1(e) The teacher understands that each learner's cognitive, linguistic, social, emotional, and physical development influences learning and knows how to make instructional decisions that build on learners' strengths and needs.

1(f) The teacher identifies readiness for learning, and understands how development in any one area may affect performance in others.

1(g) The teacher understands the role of language and culture in learning and knows how to modify instruction to make language comprehensible and instruction relevant, accessible, and challenging.

CRITICAL DISPOSITIONS

1(h) The teacher respects learners' differing strengths and needs and is committed to using this information to further each learner's development.

1(i) The teacher is committed to using learners' strengths as a basis for growth, and their misconceptions as opportunities for learning.

1(j) The teacher takes responsibility for promoting learners' growth and development.

1(k) The teacher values the input and contributions of families, colleagues, and other professionals in understanding and supporting each learner's development.

Standard #2: Learning Differences

The teacher uses understanding of individual differences and diverse cultures and communities to ensure inclusive learning environments that enable each learner to meet high standards.

PERFORMANCES

2(a) The teacher designs, adapts, and delivers instruction to address each student's diverse learning strengths and needs and creates opportunities for students to demonstrate their learning in different ways.

2(b) The teacher makes appropriate and timely provisions (e.g., pacing for individual rates of growth, task demands, communication, assessment, and response modes) for individual students with particular learning differences or needs.

2(c) The teacher designs instruction to build on learners' prior knowledge and experiences, allowing learners to accelerate as they demonstrate their understandings.

2(d) The teacher brings multiple perspectives to the discussion of content, including attention to learners' personal, family, and community experiences and cultural norms.

2(e) The teacher incorporates tools of language development into planning and instruction, including strategies for making content accessible to English language learners and for evaluating and supporting their development of English proficiency.

2(f) The teacher accesses resources, supports, and specialized assistance and services to meet particular learning differences or needs.

ESSENTIAL KNOWLEDGE

2(g) The teacher understands and identifies differences in approaches to learning and performance and knows how to design instruction that uses each learner's strengths to promote growth.

2(h) The teacher understands students with exceptional needs, including those associated with disabilities and giftedness, and knows how to use strategies and resources to address these needs.

2(i) The teacher knows about second language acquisition processes and knows how to incorporate instructional strategies and resources to support language acquisition.

2(j) The teacher understands that learners bring assets for learning based on their individual experiences, abilities, talents, prior learning, and peer and social group interactions, as well as language, culture, family, and community values.

2(k) The teacher knows how to access information about the values of diverse cultures and communities and how to incorporate learners' experiences, cultures, and community resources into instruction.

CRITICAL DISPOSITIONS

2(1) The teacher believes that all learners can achieve at high levels and persists in helping each learner reach his/her full potential.

2(m) The teacher respects learners as individuals with differing personal and family backgrounds and various skills, abilities, perspectives, talents, and interests.

2(n) The teacher makes learners feel valued and helps them learn to value each other.

2(o) The teacher values diverse languages and dialects and seeks to integrate them into his/her instructional practice to engage students in learning.

Standard #3: Learning Environments

The teacher works with others to create environments that support individual and collaborative learning, and that encourage positive social interaction, active engagement in learning, and self-motivation.

PERFORMANCES

3(a) The teacher collaborates with learners, families, and colleagues to build a safe, positive learning climate of openness, mutual respect, support, and inquiry.

3(b) The teacher develops learning experiences that engage learners in collaborative and self-directed learning and that extend learner interaction with ideas and people locally and globally.

3(c) The teacher collaborates with learners and colleagues to develop shared values and expectations for respectful interactions, rigorous academic discussions, and individual and group responsibility for quality work.

3(d) The teacher manages the learning environment to actively and equitably engage learners by organizing, allocating, and coordinating the resources of time, space, and learners' attention.

3(e) The teacher uses a variety of methods to engage learners in evaluating the learning environment and collaborates with learners to make appropriate adjustments.

3(f) The teacher communicates verbally and nonverbally in ways that demonstrate respect for and responsiveness to the cultural backgrounds and differing perspectives learners bring to the learning environment.

3(g) The teacher promotes responsible learner use of interactive technologies to extend the possibilities for learning locally and globally.

3(h) The teacher intentionally builds learner capacity to collaborate in face-to-face and virtual environments through applying effective interpersonal communication skills.

ESSENTIAL KNOWLEDGE

3(i) The teacher understands the relationship between motivation and engagement and knows how to design learning experiences using strategies that build learner self-direction and ownership of learning.

3(j) The teacher knows how to help learners work productively and cooperatively with each other to achieve learning goals.

3(k) The teacher knows how to collaborate with learners to establish and monitor elements of a safe and productive learning environment including norms, expectations, routines, and organizational structures.

3(1) The teacher understands how learner diversity can affect communication and knows how to communicate effectively in differing environments.

3(m) The teacher knows how to use technologies and how to guide learners to apply them in appropriate, safe, and effective ways.

CRITICAL DISPOSITIONS

3(n) The teacher is committed to working with learners, colleagues, families, and communities to establish positive and supportive learning environments.

3(o) The teacher values the role of learners in promoting each other's learning and recognizes the importance of peer relationships in establishing a climate of learning.

3(p) The teacher is committed to supporting learners as they participate in decisionmaking, engage in exploration and invention, work collaboratively and independently, and engage in purposeful learning.

3(q) The teacher seeks to foster respectful communication among all members of the learning community.

3(r) The teacher is a thoughtful and responsive listener and observer.

Standard #4: Content Knowledge

The teacher understands the central concepts, tools of inquiry, and structures of the discipline(s) he or she teaches and creates learning experiences that make these aspects of the discipline accessible and meaningful for learners to assure mastery of the content.

PERFORMANCES

4(a) The teacher effectively uses multiple representations and explanations that capture key ideas in the discipline, guide learners through learning progressions, and promote each learner's achievement of content standards.

4(b) The teacher engages students in learning experiences in the discipline(s) that encourage learners to understand, question, and analyze ideas from diverse perspectives so that they master the content.

4(c) The teacher engages learners in applying methods of inquiry and standards of evidence used in the discipline.

4(d) The teacher stimulates learner reflection on prior content knowledge, links new concepts to familiar concepts, and makes connections to learners' experiences.

4(e) The teacher recognizes learner misconceptions in a discipline that interfere with learning, and creates experiences to build accurate conceptual understanding.

4(f) The teacher evaluates and modifies instructional resources and curriculum materials for their comprehensiveness, accuracy for representing particular concepts in the discipline, and appropriateness for his/her learners.

4(g) The teacher uses supplementary resources and technologies effectively to ensure accessibility and relevance for all learners.

4(h) The teacher creates opportunities for students to learn, practice, and master academic language in their content.

4(i) The teacher accesses school and/or district-based resources to evaluate the learner's content knowledge in their primary language.

ESSENTIAL KNOWLEDGE

4(j) The teacher understands major concepts, assumptions, debates, processes of inquiry, and ways of knowing that are central to the discipline(s) s/he teaches.

4(k) The teacher understands common misconceptions in learning the discipline and how to guide learners to accurate conceptual understanding.

4(1) The teacher knows and uses the academic language of the discipline and knows how to make it accessible to learners.

4(m) The teacher knows how to integrate culturally relevant content to build on learners' background knowledge.

4(n) The teacher has a deep knowledge of student content standards and learning progressions in the discipline(s) s/he teaches.

CRITICAL DISPOSITIONS

4(o) The teacher realizes that content knowledge is not a fixed body of facts but is complex, culturally situated, and ever evolving. S/he keeps abreast of new ideas and understandings in the field.

4(p) The teacher appreciates multiple perspectives within the discipline and facilitates learners' critical analysis of these perspectives.

4(q) The teacher recognizes the potential of bias in his/her representation of the discipline and seeks to appropriately address problems of bias.

4 (r) The teacher is committed to work toward each learner's mastery of disciplinary content and skills.

Standard #5: Application of Content

The teacher understands how to connect concepts and use differing perspectives to engage learners in critical thinking, creativity, and collaborative problem solving related to authentic local and global issues.

PERFORMANCES

5(a) The teacher develops and implements projects that guide learners in analyzing the complexities of an issue or question using perspectives from varied disciplines and cross-disciplinary skills (e.g., a water quality study that draws upon biology and chemistry to look at factual information and social studies to examine policy implications).

5(b) The teacher engages learners in applying content knowledge to real world problems through the lens of interdisciplinary themes (e.g., financial literacy, environmental literacy).

5(c) The teacher facilitates learners' use of current tools and resources to maximize content learning in varied contexts.

5(d) The teacher engages learners in questioning and challenging assumptions and approaches in order to foster innovation and problem solving in local and global contexts.

5(e) The teacher develops learners' communication skills in disciplinary and interdisciplinary contexts by creating meaningful opportunities to employ a variety of forms of communication that address varied audiences and purposes.

5(f) The teacher engages learners in generating and evaluating new ideas and novel approaches, seeking inventive solutions to problems, and developing original work.

5(g) The teacher facilitates learners' ability to develop diverse social and cultural perspectives that expand their understanding of local and global issues and create novel approaches to solving problems.

5(h) The teacher develops and implements supports for learner literacy development across content areas.

ESSENTIAL KNOWLEDGE

5(i) The teacher understands the ways of knowing in his/her discipline, how it relates to other disciplinary approaches to inquiry, and the strengths and limitations of each approach in addressing problems, issues, and concerns.

5(j) The teacher understands how current interdisciplinary themes (e.g., civic literacy, health literacy, global awareness) connect to the core subjects and knows how to weave those themes into meaningful learning experiences.

5(k) The teacher understands the demands of accessing and managing information as well as how to evaluate issues of ethics and quality related to information and its use.

5(1) The teacher understands how to use digital and interactive technologies for efficiently and effectively achieving specific learning goals.

5(m) The teacher understands critical thinking processes and knows how to help learners develop high level questioning skills to promote their independent learning.

5(n) The teacher understands communication modes and skills as vehicles for learning (e.g., information gathering and processing) across disciplines as well as vehicles for expressing learning.

5(o) The teacher understands creative thinking processes and how to engage learners in producing original work.

5(p) The teacher knows where and how to access resources to build global awareness and understanding, and how to integrate them into the curriculum.

CRITICAL DISPOSITIONS

5(q) The teacher is constantly exploring how to use disciplinary knowledge as a lens to address local and global issues.

5(r) The teacher values knowledge outside his/her own content area and how such knowledge enhances student learning.

5(s) The teacher values flexible learning environments that encourage learner exploration, discovery, and expression across content areas.

Standard #6: Assessment

The teacher understands and uses multiple methods of assessment to engage learners in their own growth, to monitor learner progress, and to guide the teacher's and learner's decision making.

PERFORMANCES

6(a) The teacher balances the use of formative and summative assessment as appropriate to support, verify, and document learning.

6(b) The teacher designs assessments that match learning objectives with assessment methods and minimizes sources of bias that can distort assessment results.

6(c) The teacher works independently and collaboratively to examine test and other performance data to understand each learner's progress and to guide planning.

6(d) The teacher engages learners in understanding and identifying quality work and provides them with effective descriptive feedback to guide their progress toward that work.

6(e) The teacher engages learners in multiple ways of demonstrating knowledge and skill as part of the assessment process.

6(f) The teacher models and structures processes that guide learners in examining their own thinking and learning as well as the performance of others.

6(g) The teacher effectively uses multiple and appropriate types of assessment data to identify each student's learning needs and to develop differentiated learning experiences.

6(h) The teacher prepares all learners for the demands of particular assessment formats and makes appropriate modifications in assessments or testing conditions especially for learners with disabilities and language learning needs.

6(i) The teacher continually seeks appropriate ways to employ technology to support assessment practice both to engage learners more fully and to assess and address learner needs.

ESSENTIAL KNOWLEDGE

6(j) The teacher understands the differences between formative and summative applications of assessment and knows how and when to use each.

6(k) The teacher understands the range of types and multiple purposes of assessment and how to design, adapt, or select appropriate assessments to address specific learning goals and individual differences, and to minimize sources of bias.

6(1) The teacher knows how to analyze assessment data to understand patterns and gaps in learning, to guide planning and instruction, and to provide meaningful feedback to all learners.

6(m) The teacher knows when and how to engage learners in analyzing their own assessment results and in helping to set goals for their own learning.

6(n) The teacher understands the positive impact of effective descriptive feedback for learners and knows a variety of strategies for communicating this feedback.

6(o) The teacher knows when and how to evaluate and report learner progress against standards.

6(p) The teacher understands how to prepare learners for assessments and how to make accommodations in assessments and testing conditions, especially for learners with disabilities and language learning needs.

CRITICAL DISPOSITIONS

6(q) The teacher is committed to engaging learners actively in assessment processes and to developing each learner's capacity to review and communicate about their own progress and learning.

6(r) The teacher takes responsibility for aligning instruction and assessment with learning goals.

6(s) The teacher is committed to providing timely and effective descriptive feedback to learners on their progress.

6(t) The teacher is committed to using multiple types of assessment processes to support, verify, and document learning.

6(u) The teacher is committed to making accommodations in assessments and testing conditions especially for learners with disabilities and language learning needs.

6(v) The teacher is committed to the ethical use of various assessments and assessment data to identify learner strengths and needs to promote learner growth.

Standard #7: Planning for Instruction

The teacher plans instruction that supports every student in meeting rigorous learning goals by drawing upon knowledge of content areas, curriculum, crossdisciplinary skills, and pedagogy, as well as knowledge of learners and the community context.

PERFORMANCES

7(a) The teacher individually and collaboratively selects and creates learning experiences that are appropriate for curriculum goals and content standards, and are relevant to learners.

7(b) The teacher plans how to achieve each student's learning goals, choosing appropriate strategies and accommodations, resources, and materials to differentiate instruction for individuals and groups of learners.

7(c) The teacher develops appropriate sequencing of learning experiences and provides multiple ways to demonstrate knowledge and skill.

7(d) The teacher plans for instruction based on formative and summative assessment data, prior learner knowledge, and learner interest.

7(e) The teacher plans collaboratively with professionals who have specialized expertise (e.g., special educators, related service providers, language learning specialists, librarians, media specialists) to design and jointly deliver as appropriate effective learning experiences to meet unique learning needs.

7(f) The teacher evaluates plans in relation to short- and long-range goals and systematically adjusts plans to meet each student's learning needs and enhance learning.

ESSENTIAL KNOWLEDGE

7(g) The teacher understands content and content standards and how these are organized in the curriculum.

7(h) The teacher understands how integrating cross-disciplinary skills in instruction engages learners purposefully in applying content knowledge.

7(i) The teacher understands learning theory, human development, cultural diversity, and individual differences and how these impact ongoing planning.

7(j) The teacher understands the strengths and needs of individual learners and how to plan instruction that is responsive to these strengths and needs.

7(k) The teacher knows a range of evidence-based instructional strategies, resources, and technological tools and how to use them effectively to plan instruction that meets diverse learning needs.

7(l) The teacher knows when and how to adjust plans based on assessment information and learner responses.

7(m) The teacher knows when and how to access resources and collaborate with others to support student learning (e.g., special educators, related service providers, language learner specialists, librarians, media specialists, community organizations).

CRITICAL DISPOSITIONS

7(n) The teacher respects learners' diverse strengths and needs and is committed to using this information to plan effective instruction.

7(o) The teacher values planning as a collegial activity that takes into consideration the input of learners, colleagues, families, and the larger community.

7(p) The teacher takes professional responsibility to use short- and long-term planning as a means of assuring student learning.

7(q) The teacher believes that plans must always be open to adjustment and revision based on learner needs and changing circumstances.

Standard #8: Instructional Strategies

The teacher understands and uses a variety of instructional strategies to encourage learners to develop deep understanding of content areas and their connections, and to build skills to apply knowledge in meaningful ways.

PERFORMANCES

8(a) The teacher uses appropriate strategies and resources to adapt instruction to the needs of individuals and groups of learners.

8(b) The teacher continuously monitors student learning, engages learners in assessing their progress, and adjusts instruction in response to student learning needs.

8(c) The teacher collaborates with learners to design and implement relevant learning experiences, identify their strengths, and access family and community resources to develop their areas of interest.

8(d) The teacher varies his/her role in the instructional process (e.g., instructor, facilitator, coach, audience) in relation to the content and purposes of instruction and the needs of learners.

8(e) The teacher provides multiple models and representations of concepts and skills with opportunities for learners to demonstrate their knowledge through a variety of products and performances.

8(f) The teacher engages all learners in developing higher order questioning skills and metacognitive processes.

8(g) The teacher engages learners in using a range of learning skills and technology tools to access, interpret, evaluate, and apply information.

8(h) The teacher uses a variety of instructional strategies to support and expand learners' communication through speaking, listening, reading, writing, and other modes.

8(i) The teacher asks questions to stimulate discussion that serves different purposes (e.g., probing for learner understanding, helping learners articulate their ideas and thinking processes, stimulating curiosity, and helping learners to question).

ESSENTIAL KNOWLEDGE

8(j) The teacher understands the cognitive processes associated with various kinds of learning (e.g., critical and creative thinking, problem framing and problem solving, invention, memorization and recall) and how these processes can be stimulated.

8(k) The teacher knows how to apply a range of developmentally, culturally, and linguistically appropriate instructional strategies to achieve learning goals.

8(1) The teacher knows when and how to use appropriate strategies to differentiate instruction and engage all learners in complex thinking and meaningful tasks.

8(m) The teacher understands how multiple forms of communication (oral, written, nonverbal, digital, visual) convey ideas, foster self expression, and build relationships.

8(n) The teacher knows how to use a wide variety of resources, including human and technological, to engage students in learning.

8(o) The teacher understands how content and skill development can be supported by media and technology and knows how to evaluate these resources for quality, accuracy, and effectiveness.

CRITICAL DISPOSITIONS

8(p) The teacher is committed to deepening awareness and understanding the strengths and needs of diverse learners when planning and adjusting instruction.

8(q) The teacher values the variety of ways people communicate and encourages learners to develop and use multiple forms of communication.

8(r) The teacher is committed to exploring how the use of new and emerging technologies can support and promote student learning.

8(s) The teacher values flexibility and reciprocity in the teaching process as necessary for adapting instruction to learner responses, ideas, and needs.

Standard #9: Professional Learning and Ethical Practice

The teacher engages in ongoing professional learning and uses evidence to continually evaluate his/her practice, particularly the effects of his/her choices and actions on others (learners, families, other professionals, and the community), and adapts practice to meet the needs of each learner.

PERFORMANCES

9(a) The teacher engages in ongoing learning opportunities to develop knowledge and skills in order to provide all learners with engaging curriculum and learning experiences based on local and state standards.

9(b) The teacher engages in meaningful and appropriate professional learning experiences aligned with his/her own needs and the needs of the learners, school, and system.

9(c) Independently and in collaboration with colleagues, the teacher uses a variety of data (e.g., systematic observation, information about learners, research) to evaluate the outcomes of teaching and learning and to adapt planning and practice.

9(d) The teacher actively seeks professional, community, and technological resources, within and outside the school, as supports for analysis, reflection, and problem-solving.

9(e) The teacher reflects on his/her personal biases and accesses resources to deepen his/her own understanding of cultural, ethnic, gender, and learning differences to build stronger relationships and create more relevant learning experiences.

9(f) The teacher advocates, models, and teaches safe, legal, and ethical use of information and technology including appropriate documentation of sources and respect for others in the use of social media.

ESSENTIAL KNOWLEDGE

9(g) The teacher understands and knows how to use a variety of self-assessment and problem-solving strategies to analyze and reflect on his/her practice and to plan for adaptations/adjustments.

9(h) The teacher knows how to use learner data to analyze practice and differentiate instruction accordingly.

9(i) The teacher understands how personal identity, worldview, and prior experience affect perceptions and expectations, and recognizes how they may bias behaviors and interactions with others.

9(j) The teacher understands laws related to learners' rights and teacher responsibilities (e.g., for educational equity, appropriate education for learners with disabilities, confidentiality, privacy, appropriate treatment of learners, reporting in situations related to possible child abuse).

9(k) The teacher knows how to build and implement a plan for professional growth directly aligned with his/her needs as a growing professional using feedback from teacher evaluations and observations, data on learner performance, and school- and system-wide priorities.

CRITICAL DISPOSITIONS

9(1) The teacher takes responsibility for student learning and uses ongoing analysis and reflection to improve planning and practice.

9(m) The teacher is committed to deepening understanding of his/her own frames of reference (e.g., culture, gender, language, abilities, ways of knowing), the potential biases in these frames, and their impact on expectations for and relationships with learners and their families.

9(n) The teacher sees him/herself as a learner, continuously seeking opportunities to draw upon current education policy and research as sources of analysis and reflection to improve practice.

9(o) The teacher understands the expectations of the profession including codes of ethics, professional standards of practice, and relevant law and policy.

Standard #10: Leadership and Collaboration

The teacher seeks appropriate leadership roles and opportunities to take responsibility for student learning, to collaborate with learners, families, colleagues, other school professionals, and community members to ensure learner growth, and to advance the profession.

PERFORMANCES

10(a) The teacher takes an active role on the instructional team, giving and receiving feedback on practice, examining learner work, analyzing data from multiple sources, and sharing responsibility for decision making and accountability for each student's learning.

10(b) The teacher works with other school professionals to plan and jointly facilitate learning on how to meet diverse needs of learners.

10(c) The teacher engages collaboratively in the school-wide effort to build a shared vision and supportive culture, identify common goals, and monitor and evaluate progress toward those goals.

10(d) The teacher works collaboratively with learners and their families to establish mutual expectations and ongoing communication to support learner development and achievement.

10(e) Working with school colleagues, the teacher builds ongoing connections with community resources to enhance student learning and well being.

10(f) The teacher engages in professional learning, contributes to the knowledge and skill of others, and works collaboratively to advance professional practice.

10(g) The teacher uses technological tools and a variety of communication strategies to build local and global learning communities that engage learners, families, and colleagues.

10(h) The teacher uses and generates meaningful research on education issues and policies.

10(i) The teacher seeks appropriate opportunities to model effective practice for colleagues, to lead professional learning activities, and to serve in other leadership roles.

10(j) The teacher advocates to meet the needs of learners, to strengthen the learning environment, and to enact system change.

10(k) The teacher takes on leadership roles at the school, district, state, and/or national level and advocates for learners, the school, the community, and the profession.

ESSENTIAL KNOWLEDGE

10(1) The teacher understands schools as organizations within a historical, cultural, political, and social context and knows how to work with others across the system to support learners.

10(m) The teacher understands that alignment of family, school, and community spheres of influence enhances student learning and that discontinuity in these spheres of influence interferes with learning.

10(n) The teacher knows how to work with other adults and has developed skills in collaborative interaction appropriate for both face-to-face and virtual contexts.

10(o) The teacher knows how to contribute to a common culture that supports high expectations for student learning.

CRITICAL DISPOSITIONS

10(p) The teacher actively shares responsibility for shaping and supporting the mission of his/her school as one of advocacy for learners and accountability for their success.

10(q) The teacher respects families' beliefs, norms, and expectations and seeks to work collaboratively with learners and families in setting and meeting challenging goals.

10(r) The teacher takes initiative to grow and develop with colleagues through interactions that enhance practice and support student learning.

10(s) The teacher takes responsibility for contributing to and advancing the profession.

10(t) The teacher embraces the challenge of continuous improvement and change.

Appendix B: Teacher Survey

Introduction

Your school district has authorized your campus to participate in a study of teacher hiring practices in Texas sponsored by The Texas Public Schools Research Network. The purpose of this study is to describe the employment tools, procedures, and processes the district used when hiring, selecting and assigning you to your present classroom teaching duties.

As a recently employed teacher, you have unique information about how teachers in the district and at your campus are hired. Please share your professional insights with us by taking the time to complete a brief survey. By clicking the "next" button below, you agree to answer the questions in this survey. **All of your responses will remain completely confidential.** Data is collected by CREATE, an independent research organization, so that <u>NO</u> district personnel will see your responses. All responses will be aggregated, and no individually identifiable information will ever be reported.

Instructions for completing the survey:

• Please set aside 15-20 minutes to complete the survey.

• You must complete the survey in one sitting. If you exit the survey before completing it, you will not be able to re-enter the survey, and your answers will not be saved.

• When a question has been answered and the "next button" is selected, you will not be able to return to that question again. Using the back button on the web browser to return to a previous page will cause Survey Monkey to malfunction.

• Please provide responses to ALL questions, even if you have to make your best guess when you are not sure. If you choose "other" as a response to any of the questions, try to specify what you mean by "other" in the space provided.

• Once you've completed the survey, click the "Submit" button on the thank you page to exit the survey. You will be re-directed to the CREATE website.

Thank you in advance for your time and attention to this survey. Should you have questions, problems, or need additional information, please contact Sherri Lowrey by email at <u>slowrey@createtx.org</u> or by telephone at 936-273-7661.

General Information Q1

The following set of eleven questions are general information questions about your campus and campus assignment.

1. Is this your first or second year as a teacher at this campus? Please DO NOT count time spent as a student teacher or short-term substitute.

- First year
- Second year
- $\ensuremath{\mathbb{C}}$ $\ensuremath{\,\rm I}$ am beyond my second year of teaching at this campus.

General Information Q 1a

1.a. Did you teach at your current school last school year (2008-2009)?

C Yes

No

Q1b

1.b. If you were at a different campus last year, which of the following best describes why you are now teaching at a new campus?

 \bigcirc I was unable to stay in my old position because I did not have enough seniority (i.e., a teacher with more seniority applied for my position and I got "bumped").

 \bigcirc I was unable to stay in my old school, but for reasons other than being "bumped" by a teacher with more seniority.

 $\ensuremath{\overline{\bigcirc}}$ I chose to move to a different school.

General Information Q2

2. How would you classify your main assignment at your current school?

- Full-time teacher
- O Part-time teacher

C Long-term substitute (i.e., your assignment requires that you fill the role of a regular teacher on a long-term basis, but you are still considered a substitute)

○ Student teacher, paraprofessional, or volunteer

General Information Q2a

2a. How much time do you work as a teacher at your current school?

- One-half (1/2) time or more
- $\ensuremath{\mathbb{C}}$ Less than one-half (1/2) time

General Information Q-3-4

3. Please check ALL of the grade levels that are at your current campus.

 K
 1
 2
 3
 4
 5
 6
 7
 8
 9
 10
 11
 12

 4. Please check all of the grade levels that YOU currently teach:

 K
 1
 2
 3
 4
 5
 6
 7
 8
 9
 10
 11
 12

Gene	ral Information Q 5-6
	What SUBJECTS or TYPES OF CLASSES do you currently teach? (Please eck all subjects that apply).
	General elementary (multi-subject)
	General middle school (multi-subject)
	Math
	Science
	Technology
	English/Language Arts
	Foreign Language
	Bilingual Education
	Special Education
	History/Social Studies
	Arts (music, drama, visual arts)
	Physical Education
	Other (please specify)
	Please mark the answer choice that you consider to be your PRIMARY nching assignment. (Please choose only one answer).
	General elementary (multi-subject)
0	General middle school (multi-subject)
0	Math
O	Science
O	Technology
O	English/Language Arts
Õ	Foreign Language
O	Bilingual Education
0	Special Education
0	History/Social Studies
0	Arts (music, drama, visual arts)

O Physical Education

General Information Q 7-8

7. If you taught last year, did your assignment change?

- O Yes
- No
- \bigcirc Not applicable because this is my first year of full-time teaching

8. Which of the following best describes the teacher preparation program you completed?

- \bigcirc Traditional undergraduate teacher preparation program
- $\ensuremath{\mathbb{C}}$ University post baccalaureate teacher preparation program
- University alternative certification program
- Private alternative certification program
- School district certification program
- Service center certification program
- \bigcirc No formal teacher preparation
- Other (please specify)

General Information Q 9-10

9. Which of the following choices best describes the life or career stage from which you entered teaching?

- $\mathbb C$ Teaching is my first career/job after college (not including short-term or temporary work).
- $\ensuremath{\mathbb{C}}$ I switched to teaching from full-time child-raising.
- \bigcirc I switched to teaching from another field of work.
- $\ensuremath{\mathbb{C}}$ $\ensuremath{\,\rm I}$ switched to teaching from another permanent job within the field of education.

10. Which of the following best describes how you view your teaching job?

 \bigcirc I most likely will remain a classroom teacher for the rest of my career.

 \bigcirc I most likely will leave classroom teaching at some point, but I plan to stay in the field of education for the rest of my career.

O I most likely will leave classroom teaching at some point, and I plan to work in another job(s) outside the field of education for the rest of my career.

General Information Q 11-12

11. So far, how satisfied are you with TEACHING?

- O Very Dissatisfied
- O Dissatisfied
- Somewhat Dissatisfied
- Neutral
- Somewhat Satisfied
- Satisfied
- O Very Satisfied

12. So far, how satisfied are you with YOUR SCHOOL AS A PLACE TO TEACH?

- O Very Dissatisfied
- O Dissatisfied
- Somewhat Dissatisfied
- O Neutral
- Somewhat Satisfied
- Satisfied
- O Very Satisfied

General Information Q13

13. How big a role did each of the following factors play in your decision to enter teaching?

	No Role	Small Role	Moderate Role	Large Role	Critical role
(a) Wanted meaningful work	C	C	C	0	O
(b) Wanted to work with students	Õ	Õ	O	\odot	O
(c) Dissatisfied with previous career	C	C	C	0	Ō
(d) Wanted to contribute to society	Õ	Õ	Ō	O	Ō
(e) Interested in pedagogy/teaching	C	C	C	0	Ō
(f) Interested in sharing love of subject matter with students	C	O	O	0	C
(g) Saw pay as attractive	C	C	C	0	O
(h) Found the daily and yearly schedule attractive	C	O	O	O	C

II. Hiring Process Q 1-1a

The next set of fifteen questions ask specifically about the hiring process. Please answer the questions in this section based on the teaching position that YOU CURRENTLY HOLD.

1. From which of the following sources did you learn of the opening for the job that you now have? (Please check all that apply).

- Contacted the campus directly (without prior knowledge of an opening)
- \square Contacted the district central office
- Contacted the campus principal or assistant principal
- College or university visit by school district recruiter
- □ College or university career placement office
- Current or former teacher at the campus where you now work
- Former co-worker (other than campus teacher)
- Friend or relative
- Broadcast media (t.v./radio)
- Internet posting
- 🔲 Job fair
- Newspaper advertisement
- Web posting

 \square Prior work experience at the campus (If you checked this answer, please answer the next question, Question 1a).

1a. If you checked "Prior work at the school" from the list above, please answer the following question:

In which of the following capacities did you work for your school prior to taking your current teaching position? (Please check all that apply).

2. Which of the following materials did you submit as part of your application for the position? (Please check all that apply).

Documentation of certification

- Undergraduate college transcripts
- $\begin{tabular}{|c|c|c|c|} \hline \end{tabular}$ Results of online prescreening instrument
- Professional references
- Sample lesson plan(s)
- Portfolio reflecting your preparation to teach
- □ Scores on teacher certification exam
- □ Writing sample or essay
- ☐ Graduate school transcripts
- ☐ Videotape of you teaching a sample lesson
- 🗌 Resume
- Other (please specify)

3. Which of the following BEST describes how you were hired? (Please choose only one answer).

 \bigcirc Offered a job by district central office, then assigned to a specific campus by the district

O Offered a job by the district office, then interviewed with campuses in the district to find a specific teaching position

C Screened by district central office (i.e., online or brief initial interview or conversation resulting in no guarantee of a job), then interviewed with and offered a job by a specific campus

O Screened by district, interviewed at campus and after principal recommendation, offered a job by district central office

 $\ensuremath{\mathbb{C}}$ Applied directly to a specific campus and was offered a position by that campus

4. Did you interview for your position before you were offered the job? Interviews can be either formal or informal, but they consist of more than chance conversations.

0	Yes
---	-----

🔿 No

Hiring Process redir Q4a-b

4a. Approximately how many different interviews did you have for this position? (Please enter a whole number).

4b. Did any of these interviews involve your being interviewed at the <u>campus level</u> by a group of individuals?

⑦ Yes

🔿 No

Hiring Process redir Q4c-d

4c-d. Who participated in the interview(s) and how many people from the following groups were present?

Please review each category below and indicate how many people from each group were present during the interview by typing the number in the box to the right of each category. IF a group was NOT represented during an interview, type a "0" in the box.

District or human resource personnel	
Superintendent	
Assistant Superintendent	
Campus principal	
Campus assistant principal or dean	
Campus department chair	
Campus program coordinator or other administrator	
Campus teacher	
Campus student	
Campus parent	

Hiring Process Q 5 5. In applying for your job, did someone observe you teach a lesson? O Yes 🔿 No

Hiring Process redir Q5a

5a. If so, who observed you teach the lesson? (Please check all that apply).

- District or human resource personnel
- Superintendent
- Campus principal
- □ Campus assistant principal or dean
- 🔲 Campus department chair
- Campus program coordinator or other administrator
- Campus Teacher
- Campus student
- 🗌 Campus parent

Other (please specify)

6. In applying for your job, did you visit and observe any classes while the campus was in session?

O Yes

🔿 No

Hiring Process redir Q 6a

6a. How many classes did you visit? (Please enter a whole number).

7. In applying for your job, did you observe or sit in on any faculty or team meetings (for instance, department meetings, grade-level meetings, cluster meetings, or full-faculty meetings)?

O Yes

🖸 No

Hiring Process redir Q7a

7a. How many meetings (either in full or in part) did you observe? (Please enter a number).

8. Approximately how many weeks elapsed between the time you were hired and your actual teaching responsibilities began? (Please enter a whole number).

9. Were you hired after the school year had already started?

YesNo

10. Did you have more than one job offer at the time that you decided to accept your current job?

O Yes

🖸 No

10 a-c. You indicated that you had more than one job offer at the time you accepted your current position. (Please provide a whole number for each of the following questions).
<form></form>

11. Approximately how many districts did you apply to (i.e., you sent a resume or letter to the district, or completed an application)?

151

In this question, we are interested in whether you were able to get an accurate picture (or preview) of what your campus would be like before you took your job. In the questions below, the "hiring process" refers to the interviews, observations, and informal conversations you participated in, or the written materials you submitted or received while applying for the job.

***** 12. FROM THE HIRING PROCESS, I GOT AN ACCURATE PICTURE OF:

	Strongly Disagree	Disagree	Somewhat Disagree	Neutral	Somewhat Agree	Agree	Strongly Agree
(a) What the <u>teachers</u> were like at the campus and whether I might enjoy working with them	0	C	0	C	0	C	C
(b) What the <u>students</u> were like at the campus and whether I might enjoy working with them	C	C	C	Õ	C	O	C
(c) The principal's leadership style	Ō	\odot	C	O	C	C	Ο
(d) The curriculum I would be teaching	0	O	O	Ô	C	O	O
(e) What my teaching assignment would be (i.e., subjects, grade levels, number of classes, other duties)	O	C	O	C	C	O	C
(f) The support that the campus would provide to me as a new teacher	C	C	O	C	O	O	C
(g) How much autonomy I would have as a teacher at the campus	C	C	C	C	O	C	O
(h) The opportunities I might have to help make important campus-wide decisions	C	O	C	C	O	C	O
(i) The educational philosophy of the campus	C	O	O	C	О	O	О
(j) The non-classroom duties I would be required to perform	O	O	C	O	O	O	O

***** 13. How closely would you say that your current TEACHING ASSIGNMENT matches the following:

	Very Poor Match	Poor Match	Moderate Match	Good Match	Very Good Match
 (a) Your subject matter knowledge and expertise 	C	О	C	C	C
(b) Your subject matter interests	Õ	0	O	Õ	\odot
(c) Other skills and talents that you have (e.g., coaching sports, organizing extracurricular activities, or advising students)	С	C	C	C	С
(d) The grade level(s)that you would preferto teach	C	O	C	C	C
(e) The type of student population you would prefer to teach	O	0	O	C	O

***** 14. How closely would you say that YOUR CAMPUS matches the following:

	Very Poor Match	Poor Match	Moderate Match	Good Match	Very Good Match
(a) Your own educational philosophy	C	0	C	O	O
(b) The amount of autonomy you would like to have as a teacher (i.e., over what and how much to teach)	C	O	O	C	O
(c) Your own views on student discipline	C	0	C	O	O
(d) The amount of collaboration or teamwork you would like with colleagues	C	C	C	C	O
(e) The amount of input (or influence) you would like to have on campus-wide decisions	C	C	C	С	С
(f) The amount of input (or influence) you would like to have on department or grade-level decisions	O	O	O	C	O

* 15. IF TODAY you were choosing among several job offers, how important would each of the following factors be in deciding which TEACHING JOB to take?

	Not Important	Slightly Important	Moderately Important	Very Important	Extremely Important
(a) Salary and benefits	O	\odot	\odot	\odot	\odot
(b) The principal and administrators at the campus and their leadership style	O	O	O	O	©
(c) Teachers at the campus and what you think they would be like as colleagues	O	C	O	O	C
(d) Student population the campus serves	Õ	O	O	O	O
(e) Grade level of students you would be teaching	0	O	O	0	O
(f) Overall teaching load (i.e., number of courses, number of students	O	O	C	C	O
(g) The specific content assignment (i.e., subjects, courses)	0	C	C	C	C
 (h) Amount of non- teaching responsibilities you would have 	O	C	C	C	C
(i) Opportunities for professional development and growth	O	C	C	C	C
(j) Campus' curriculum	Õ	O	\odot	\odot	\odot
(k) Campus' resources and facilities	C	Ô	O	O	0
(I) Length of the daily commute from home to campus	O	C	C	C	C

3rd reponse Jump from Q1

Because you indicated that you had more teaching experience than two years, you are excused from participanting in the survey. Please answer the following demographic questions to exit the survey.

Response jump from Q2

Because you indicated that you do not work as a full or over 50% time teacher, you are excused from completing this survey. Please answer the following demographic questions to exit the survey.

2.	Are you female or male?
0	Female
0	Male
3. '	What is your race or ethnicity?
0	American Indian or Alaska Native
0	Black or African American
0	Asian or Pacific Islander
C	Hispanic or Latino
C	White
O	Other (please specify)
	What is the highest degree or level of schooling that you have complet Bachelor's degree (e.g., BA, AB, BS)
0	
0	Bachelor's degree (e.g., BA, AB, BS)
0	Master's degree (e.g., MA, MS, MEd, EdM, MSW, MAT, MBA, MEng)

End

Thank you for completing the Selection Study Teacher Questionnaire.

We realize you are a busy professional with many competing time demands. We appreciate your willingness to share your professional experience and insight with us.

The data gathered through the survey will help CREATE and The Texas Public Schools Research Network better understand the teacher selection process in your district. We want to reemphasize that all responses will remain confidential, and no individually identifying information will <u>ever</u> be reported. We will send a copy of the final report to all respondents.

<u>Please click on the submit button</u> and the answers you provided will be counted. You will exit the survey and automatically be directed to the CREATE website.

Appendix C: IRB Approval for CREATE/TASA Employment Study

Tarleton State University

From:Williamson, Ms. NonaSent:Friday, January 15, 2010 3:04 PMTo:Johnson, Dr. DaleSubject:IRB application

Your IRB application "Teacher Selection, Assignment and Classroom Effectiveness" has been approved "Exempt". Thank you for submitting your application and we wish you success in your research.

Your IRB No. 2010-011310-10004

Thank you,

Nona Williamson Tarleton State University Administrative Assistant IV Sponsored Projects 254 968-9463 254 968-9509 Fax

Appendix D: CREATE Permission Letter



Center for Research, Evaluation & Advancement of Teacher Education

December 7, 2010

To: The Office of Research Support, The University of Texas at Austin From: William Reaves, Executive Director Emeritus, CREATE

RE: Permission to use CREATE dataset

To Whom It May Concern:

The Center for Research, Evaluation, and Advancement of Teacher Education (CREATE) gives permission for Julia C. McCreary to use the dataset from the teacher survey instrument administered in spring 2010 for the purposes of her dissertation research.

The data from the teacher survey instrument is part of a larger employment study conducted by CREATE and the Texas Association of School Administrators (TASA). Dr. Dale Johnson, professor at Tarleton State University is acting as principal investigator for the employment study. Dr. Johnson received IRB approval from Tarleton State University for the research associated with all components of the larger study.

Please feel free to contact me at the CREATE office with any questions 936-273-7661 or <u>wreaves@createtx.org</u>.

Sincerely, William Reaves Executive Director Emeritus

.

3232 College Park Drive, Suite 303, The Woodlands, TX 77384 • 936.273.7661 DIRECT • 936.273.7592 FAX

Appendix E: Exemption Status Approval



OFFICE OF RESEARCH SUPPORT

THE UNIVERSITY OF TEXAS AT AUSTIN P.O. Box 7426, Austin, Texas 78713 (512) 471-8871 -FAX (512 471-8873) North Office Building A, Suite 5.200 (Mail code A3200)

FWA # 00002030

Date: 12/14/10

PI(s): Julia C McCreary

Department&MailCode:

Title: Teachers' Perceptions of the Hiring Process in Texas Public Information Richness, Position Fit, and Student Achievement

IRB EXEMPT DETERMINATION: IRB Protocol # 2010-11-0080

Dear: Julia C McCreary

Recognition of Exempt status based on 45CFR 46.101(b).

Qualifying Period: 12/14/2010 - 12/13/2013 Expires 12 a.m. [midnight] of this date. A continuing review report must be submitted in three years if the research is ongoing. (4) Research involving the collection or study of existing data, documents, records, pathological specimens, or diagnostic specimens, if these sources are publicly available or if the information is recorded by the investigator in such a manner that subjects cannot be identified, directly or through identifiers linked to the subjects.

Responsibilities of the Principal Investigator:

Research that is determined to be Exempt from Institutional Review Board (IRB) review is not exempt from ensuring protection of human subjects. The following criteria to

protect human subjects must be met. The Principal Investigator (PI):

1. Assures that all investigators and co-principal investigators are trained in the ethical principles, relevant Federal Regulations and institutional policies governing human subject research.

2. Will provide subjects with pertinent information (e.g. risks and benefits, contact information for investigators and IRB Chair) and assures that human subjects will voluntarily consent to participate in the research when appropriate (e.g. surveys, interviews).

3. Assures the subjects will be selected equitably, so that the risks and benefits of the research are justly distributed.

4. Assures that the IRB will be immediately informed of any information or unanticipated problems that may increase the risk to the subjects and cause the category of review to be reclassified to Expedited or Full Board Review.

5. Assures that the IRB will be immediately informed of any complaints from subjects regarding their risks and benefits.

6. Assures that confidentiality and privacy of the subjects and the research data will be maintained appropriately to ensure minimal risk to subjects.

7. Will report, by amendment, any changes in the research study.

These criteria are specified in the PI Assurance Statement that must be signed before determination of Exempt status will be granted. The PI's signature acknowledges that he/she understands and accepts these conditions. Refer to the Office of Research Support (ORS) website, www.utexas.edu/irb for specific information on training, voluntary informed consent, privacy, and how to notify the IRB of unanticipated problems.

1. **Closure:** Upon completion of the research study, a Closure Report must be submitted to the ORS.

2. Unanticipated Problems: Any unanticipated problems or complaints must be reported to the IRB/ORS immediately. For a description of unanticipated problems, please refer to the ORS webpage:

http://www.utexas.edu/research/rsc/humansubjects/policies/section7.html#7.3

3. **Informed Consent:** The informed consent procedures laid out within your research proposal must be followed.

4. Continuing Review: If the study will continue beyond the three year qualifying

period, a continuing review report must be filed.

5. Amendments: Amendments do not need to be filed with the ORS if the amendments do not change the risk level of the study (for example: increasing sample size, adding or removing co- Principal Investigators, adding or removing research sites, or minor modifications to the research protocol). Changes altering the level of risk to subjects must be requested by submitting an amendment application and revised proposal to the ORS prior to those changes being implemented. For a description of the types of modifications that require an amendment application, refer to the ORS webpage: http://www.utexas.edu/research/rsc/humansubjects/policies/section6.html#635b, or call 471- 8871.

If you have any questions call or contact the ORS (Mail Code A3200) or via e-mail at orsc@uts.cc.utexas.edu. Sincerely,

John LAwa

Jody L. Jensen, Ph.D. Professor Chair, Institutional Review Board

References

Achinstein, B., Ogawa, R. T., & Speiglman, A. (2004). Are we creating separate and unequal tracks of teachers? The effects of state policy, local conditions, and teacher characteristics on new teacher socialization. *American Educational Research Journal*, 41(3), 557–603. Retrieved from JSTOR database.

Agresti, A. (2007). *An introduction to categorical data analysis*. (2nd ed.). New Jersey: John Wiley & Sons, Inc.

- Anderson, N. M. (2010, April). A cliffhanger: How America's schools continue to feel the impact of the economic downturn. Retrieved from the American Association of School Administrators website: http://www.aasa.org
- Ballou, D. (1996). Do public schools hire the best applicants? *Quarterly Journal of Economics*, 111(1), 97–133. Retrieved from JSTOR database.
- Ballou, D., & Podgursky, M. (1995). Recruiting smarter teachers. *Journal of Human Resources*, 30(2), 326-338. Retrieved from JSTOR database.
- Ballou, D., & Podgursky, M. (1998). Teacher recruitment and retention in public and private schools. *Journal of Policy Analysis and Management*, 17(3), 393-417. Retrieved from Wiley InterScience database.
- Behrstock, E., & Clifford, M. (2009, February). Leading Gen Y teachers: Emerging strategies for school leaders. (Research and Policy Brief). Retrieved from the National Comprehensive Center for Teacher Quality website: http://www.tqsource.org
- Björk, L. G. & J. Blase. (2009). The micropolitics of school district decentralization. *Educational Assessment, Evaluation, and Accountability 21*(3), 195–208.
- Bolton, D. L., (1969). The effects of various formats on teacher selection decisions. *American Educational Research Journal, (6)*3, 329–347. Retrieved from JSTOR database.
- Boote, D. N., & Beile, P. (2005). Scholars before researchers: on the centrality of the dissertation literature review in research preparation. *Educational Researcher*, 34(6), 3–15.
- Bowman, J. S. (2005). *The use of fit in teacher selection: Do principals and superintendents think alike*? Paper presented at the Academy of Human Resource Development International Conference. Retrieved from http://www.eric.ed.gov/

- Boyd, D., Grossman, P., Lankford, H., Loeb, S., & Wyckoff, J. (2005). How changes in entry requirements alter the teacher workforce and affect student achievement. (NBER Working Paper 11844). Retrieved from the National Bureau of Economic Research Working Paper Series Database.
- Boyd, D., Lankford, H., Loeb, S., & Wyckoff, J. (2005). The draw of home: How teachers' preferences for proximity disadvantage urban schools. *Journal of Policy Analysis and Management*, 24(1), 113–132.
- Boyd, D., Lankford, H., Loeb, S., & Wyckoff, J. (2006). Analyzing the determinants of the matching of public school teachers to jobs: Estimating compensating differentials in imperfect labor markets. (A revision of NBER working paper 9878). Retrieved from the Teacher Policy Research website: http://www.teacherpolicyresearch.org
- Bretz, Jr., R. D., & Judge, T. A. (1994). The role of human resource systems in job applicant decision processes. *Journal of Management*, 20(3), 531–551.
- Cable, D. M., & Judge, T. A. (1996). Person-organization fit, job choice decisions, and organizational entry. Organizational Behavior and Human Decision Processes, 67(3), 294–311.
- Castetter, W. B., & Young, I. P., (2000). *The human resource function in educational administration*. (7th Ed.). Upper Saddle River, NJ: Merrill.
- Chatman, J. A., (1991). People and organizations: Selection and socialization in public accounting firms. *Administrative Science Quarterly*, *36*(3), 459–484.
- Claycomb, C. (2000). High-quality urban school teachers: What they need to enter and to remain in hard-to-staff schools. *The State Education Standard, Winter*, 17–20. Retrieved from the National Association of School Boards website: http://www.nasbe.org
- Clement, M. C. (2008). *Recruiting and hiring effective teachers: A behavior-based approach*. Alexandria, VA: Educational Research Service.
- Clotfelter, C. T., Ladd, H. F., & Vigdor, J. L. (2005). Who teaches whom? Race and the distribution of novice teachers. *Economics of Education Review*, 24(4), 377–392.
- Clotfelter, C. T., Ladd, H. F., & Vigdor, J. L. (2007). How and why do teacher credentials matter for student achievement? (NBER Working Paper No. 8792-750). Retrieved from http://www.nber.org/papers.html
- Coleman, J. S. *Equality of educational opportunity*. Washington, DC: U.S. Department of Health, Education, and Welfare, Office of Education.

- Cohn, M. M. (1992). How teachers perceive teaching: Change over two decades, 1964– 1984. In A. Lieberman (Ed.), *The changing contexts of teaching* (pp. 110–137). Chicago: The University of Chicago Press.
- Cooper, B. S., Fusarelli, E. & Randall, V. (2004). *Better polices, better schools: theories and applications*. Boston: Allyn and Bacon.
- Corcoran, S. P. (2009). Human capital policy and the quality of the teacher workforce. In Goldhaber, D., & Hannaway, J. (Eds.), *Creating a new teaching profession* (pp. 29–52). Washington, DC: Urban Institute Press.
- Council of Chief State School Officers. (2005). Next steps: Moving toward performancebased licensing in teaching. Interstate New Teacher Assessment and Support Consortium INTASC. Retrieved from http://www.ccsso.org/Projects/ interstate_new_teacher_assessment_and_support_consortium/780.cfm
- Council of Chief State School Officers. (2008). *Educational Leadership Policy Standards: ISLLC 2008.* Retrieved from http://www.ccsso.org/projects /education leadership initiatives/ISLLC Standards/
- Council of Chief State School Officers, Interstate Teacher Assessment and Support Consortium. (2011, April). *InTASC model core teaching standards: A resource for state dialogue*. Retrieved from http://www.ccsso.org/Documents/2011/InTASC_Model_Core_Teaching_Standar ds_2011.pdf
- DeArmond, M. M., Shaw, K. L., & Wright, P. M. (2009). Zooming in and zooming out. Rethinking school district human resource management. In Goldhaber, D., & Hannaway, J. (Eds.), *Creating a new teaching profession* (pp. 53–79). Washington DC: Urban Institute Press.
- Darling-Hammond, L. (2001). The challenge of staffing our schools. *Educational Leadership*, 58(8), 12–17. Retrieved from the Professional Development database.
- Darling-Hammond, L., & Prince, C. D. (2007, July). *Strengthening teacher quality in high-need schools–Policy and practice*. Retrieved from the Council of Chief State School Officers website: http://www.ccsso.org/
- Delli, D. A., & Vera, E. M. (2003). Psychological and contextual influences on the teacher selection interview: A model for future research. *Journal of Personnel Evaluation in Education*, 17(2), 137–155.
- Dugonni, B. L., & Ilgen, D. R. (1981). Realistic job previews and the adjustment of new employees. *The Academy of Management Journal*, 24(3), 579–591.

- Ferguson, R. E., (1991). Racial patterns in how school and teacher quality affect achievement and earnings. *Challenge: A Journal of Research on African American Men*, 1. Retrieved from Academic Search Complete database.
- Galiani, S., Gertler, P., & Schargrodsky, E. (2008). School decentralization: Helping the good get better, but leaving the poor behind. *Journal of Public Economics* 92(10–11), 2106–2120.
- Gooden, M. A. (2002). Stewardship and critical leadership: Sufficient for leadership in urban schools? *Education and Urban Society*, *35*, 133–143.
- Gooden, M. A. (2005). The role of an African American principal in an urban information technology high school. *Educational Administration Quarterly*, *41*, 630–650.
- Goodlad, J. I. (1984). A place called school. New York: McGraw Hill.
- Guarino, C. M., Santibañez, L., & Daley, G. A. (2006). Teacher recruitment and retention: A review of the recent empirical literature. *Review of Educational Research*, 76, 173–208.
- Hannaway, J., & Stanislawski, M. (2005). Flip-flops in school reform: An evolutionary theory of decentralization. In, F. M. Hess (Ed.), *Urban school reform: Lessons* from San Diego (pp. 53–70). Cambridge, MA: Harvard Education Press.
- Hanson, M. E. (2003). *Educational administration and organizational behavior*. (5th ed.). Boston: Allyn and Bacon.
- Hanushek, E. A., & Rivkin, S. G. (2004). How to improve the supply of high quality teachers. (Paper Prepared for Brookings Papers on Education Policy), 7–25. Washington, DC: Brookings Institution Press.
- Hanushek, E. A., & Rivkin, S. G. (2010, March). Constrained job matching: Does teacher job search harm disadvantaged urban schools? (Working paper 15816).
 Retrieved from the National Bureau of Economic Research website: http://www.nber.org/papers/w15816
- Harlen, W., & Schlapp, U. (1998). Literature reviews. *SCRE Spotlights*, 71, 1–5. Edinburg, Scotland: Scottish Council for Research in Education.
- Harris, D. N., Rutledge, S. A., Ingle, W. K., & Thompson, C. C. (2007). Mix and match: What principals look for when hiring teachers and what this means for teacher quality policies. Retrieved from the Teacher Quality Research website: http://www.teacherqualityresearch.org

- Hays, K. T., & Behrstock, E. (2009, June). *Teacher recruitment: Strategies for widening the teaching pool in a shrinking economy*. (A Policy to Practice Brief). Retrieved from the National Comprehensive Center for Teacher Quality website: http://www.tqsource.org
- Heck, R. H. (2004). *Studying educational and social policy. Theoretical Concepts and research methods.* Mahwah, NJ: Lawrence Erlbaum Associates.
- Hess, F. M. (2009). The human capital challenge. In Goldhaber, D., & Hannaway, J. (Eds.), *Creating a new teaching profession*, (pp. 115–136). Washington DC: Urban Institute Press.
- Hill, P. T. (2009). Consequences of instructional technology for human resource needs in education. In Goldhaber, D., & Hannaway, J. (Eds.), *Creating a new teaching profession*, (pp. 137–163). Washington DC: Urban Institute Press.
- House Bill 1, 82nd Regular Session, ch. 1355, General and Special Laws of Texas, 2011.
- Ingersoll, R. M. (2001). Teacher turnover and teacher shortages: An organizational analysis. *American Educational Research Journal*, *38*(3), 499–534. Retrieved from JSTOR database.
- Jacob, B. A. (2007). The challenges of staffing urban schools with effective teachers. *The Future of Children, 17*(1), 129–153. Retrieved from the Project Muse Database: http://muse.jhu.edu/
- Johnson, S. M., & Birkeland, S. E. (2003). Pursuing a "sense of success": New teachers explain their career decisions. *American Educational Research Journal*, 40(3), 581-618. Retrieved from JSTOR database.
- Judge, T. A., Higgins, C. A., & Cable D. M. (2000). The employment interview: A review of recent research and recommendations for future research. *Human Resource Management Review*, 10(4), 383-486. Retrieved from Elsevier Science Inc. database.
- Kerlinger, F. N., & Lee, H. B. (2000). *Foundations of Behavioral Research* (4th ed.). Northridge, CO: Wadsworth Thomson Learning.
- Kristoff, A. L. (1996, spring). Person-organization fit: An integrative review of its conceptualizations, measurement, and implications. *Personnel Psychology*, 1–49. Retrieved from EBCO database.
- Lankford, H., Loeb, S., & Wyckoff, J. (2002). Teacher sorting and the plight of urban schools. A descriptive analysis. *Educational Evaluation and Policy Analysis*, 24(1), 37–62. Retrieved from JSTOR database.

- Lee, T. W., & Mowday, R. T. (1987). Voluntarily leaving an organization: An empirical investigation of Steer's and Mowday's model of turnover. Academy of Management Journal, 30, 721–743. Retrieved from JSTOR database.
- Leedy, P., & Ormrod, J. (2005). *Practical research: Planning and design*. New York: Macmillan.
- Levin, J., & Quinn, M. (2003). *Missed opportunities: How we keep high-quality teachers out of urban classrooms*. Retrieved from The New Teacher Project website: http://www.tntp.org
- Levin, J., Mulhern, J., & Schunk, J. (2005). Unintended consequences: The case for reforming the staffing rules in urban teachers union contracts. Retrieved from The New Teacher Project website: http://www.tntp.org
- Liu, E. (2002, April). *New teachers' experiences of hiring in New Jersey*. Paper presented at the annual meeting of the American Educational Research Association, New Orleans, LA. Retrieved from http://www.gse.harvard.edu/~ngt/papers.htm#hiring
- Liu, E., & Johnson, S. M. (2006). New teachers' experiences of hiring: Late, rushed, and information poor. *Educational Administration Quarterly*, 43(3), 324-360.
- Lortie, D. C. (1975). *School-teacher: A sociological study*. Chicago: The University of Chicago Press.
- McGraner, K. L. (2009, June). *Key issue: Recruiting science, technology, engineering, and mathematics (STEM) teachers.* Retrieved from the National Comprehensive Center for Teacher Quality website: http://www.tqsource.org
- Mertens, D. M. (2005). *Research and evaluation in education and psychology*. *Integrating diversity with quantitative, qualitative, and mixed methods*. (2nd ed.). Thousand Oaks, CA: SAGE Publications.
- Meek, A. (1998). America's teachers: Much to celebrate. *Educational Leadership*, 55, 12–17. Retrieved from <u>http://www.eric.ed.gov/</u>
- Metzer, S. A., & Wu, M. J. (2008). Commercial teacher selection instruments: The validity of selecting teachers through beliefs, attitudes and values. *Review of Educational Research*, 78(4), 921–940.
- Miller, R. & Chait, R. (2008). *Teacher turnover, tenure policies, and the distribution of teacher quality. Can high-poverty schools catch a break?* Washington, DC. Retrieved from www.americanprogress.org

- Murnane, R. J., Singer, J. D., Willett, J. B., Kemple, J. J., & Olsen, R. J. (1991). *Who will teach? Policies that matter*. Cambridge, MA: Harvard University Press.
- National Commission on Teaching and America's Future (1996). *What matters most: Teaching for America's future. Report of the National Commission on Teaching and America's Future.* Woodbridge, VA. Retrieved from <u>http://www.eric.ed.gov/</u>
- National Commission on Teaching and America's Future (2007). *The High Cost of Teacher Turnover*. Washington, DC. Retrieved from <u>www.nctaf.org</u>
- No Child Left Behind Act of 2001, Public Law No: 107-110. (2002)
- No Child Left Behind Act of 2001: Qualifications for Teachers and Professionals, 20 U.S.C. §6319, (2002).
- Obama, B. (2010, July 13). *Obama's Remarks to the American Federation of Teachers*. Speech given to the 80th Convention of the American Federation of Teachers via satellite. Retrieved from http://www.realclearpolitics.com/
- O'Reilly, C. A., Chatman, J., & Caldwell, D. F. (1991). People and organizational culture: A profile comparison approach to assessing person-organization fit. *The Academy of Management Journal*, *34*(3), 487-516. Retrieved from JSTOR database.
- Papa, F., Jr., & Baxter, I. (2008). Hiring teachers in New York's public schools: Can the principal make a difference? *Leadership and Policy in Schools*, 7(1), 87-117. doi:10.1080/15700760701655524
- Perrachione, B. A., Rosser, V. J., & Peterson, G. J. (2008, fall). Why do they stay? Elementary teachers' perceptions of job satisfaction and retention. *The Professional Educator*, 32(2), 1–17. Retrieved from http://www.eric.ed.gov/PDFS/EJ862759.pdf
- Peterson, K. (2002). Effective teacher hiring: A guide to getting the best. Alexandria, VA: Association for Curriculum and Professional Development.
- Rockoff, J. E., Jacob, B. A., Kane, T. J., & Staiger, D. O. (2008, November). Can you recognize an effective teacher when you recruit one? (Working paper 14485). Retrieved from the National Bureau of Economic Research website: http://www.nber.org/papers/w14485
- Rutledge, S. A., Harris, D. N., Thompson, C., & Ingle, K. W. (2007). Certify, blink, hire: An examination of the process and tools of teacher screening and selection. Retrieved from the Teacher Quality Research website: http://www.teacherqualityresearch.org

- Sachs, S. K. (2004). Evaluation of teacher attributes as predictors of success in urban schools. *Journal of Teacher Education*, 55(2), 177–187.
- Sanders, W. L., & Horn, S. P. (1998) Research findings from the Tennessee value-added assessment system (TVAAS) database: Implications for teacher evaluation. *Journal of Personnel Evaluation in Education*, 12(3), 247–256.
- Scafidi, B., Sjoquist, D. L., & Stinebrickner, T. R. (2007). Race poverty and teacher mobility. *Economics of Education Review*, 26, 145–159.
- Schwartz, H. (2010). The endurance of centralized government systems in an age of school district decentralization. (Report prepared for the Ford Foundation Project on Choice District Governance). Retrieved from the National Center for the Study of Privatization in Education organization website: ncspe.org/publications
- Seyforth, J. T. (2002). *Human resources management for effective schools* (3rd ed.). Boston: Allyn and Bacon.
- Shaffer, J. (2008). Gen Y talent: How to attract and retain the young and the restless. (White Paper). Redwood shores, CA: Saba; and Washington, DC: Human Capital Institute. Retrieved from http://www.saba.com
- Smith, R. E. (2009) *Human resources administration: A school-based perspective* (4th ed.). Larchmont, NY: Eye on Education, Inc.
- Sorenson, R. D., & Goldsmith, L. (2009). *The Principal's Guide to Managing School Personnel*. Thousand Oaks, CA: Corwin Press.
- Strauss, R. P., Bowes, L. R., Marks, M. S., & Plesko, M. R. (2000). Improving teacher preparation and selection: Lessons from the Pennsylvania experience. *Economics* of Education Review, 19(4), 387–415. Retrieved from Elsevier database.

Texas Administrative Code, Title 19 Education. Part 7: State Board for Educator Certification, Chapters 230–233.

Texas Education Code: Campus Planning and Site-Based Decision-Making, §11.253 (1995a).

Texas Education Code: Principals, §11.202 (1995b).

The New Teacher Project. (2003). *Missed Opportunities: How we keep high quality teachers out of urban classrooms*. Retrieved from <u>http://tntp.org/assets/documents/MissedOpportunities.pdf?files/MissedOpportuniti</u> es.pdf

- The New Teacher Project. (2007). *Hiring, assignment, and transfer in Chicago public schools*. Retrieved from http://www.tntp.org
- The New Teacher Project. (December, 2010). *Certifying Effective Teachers: The New Teacher Project's Approach*. Retrieved from the University of Indianapolis, 2010 Education Transformation Conference website: http://cell.uindy.edu/2010educationconference/resources/Menezes-CertifyingEffectiveTeachers.pdf
- Trimble, S. (2001). The teacher applicant pool: What top administrators seek. *Principal Leadership*, *1*(7), 45–54. Retrieved from WilsonWeb database.
- U.S. Department of Education (2010). A blueprint for reform. The reauthorization of the Elementary and Secondary Education Act. Retrieved from http://www2.ed.gov/policy/elsec/leg/blueprint/index.html
- U.S. Equal Employment Opportunity Commission. (2010). Federal equal employment opportunity laws. Retrieved from http://www.eeoc.gov/laws/statutes/index.cfm
- Wayne, A. J., & Youngs, P. (2003). Teacher characteristics and student achievement gains: A review. *Review of Educational Research*, 73, 89–111.
- Weisberg, D., Sexton, S., Mulhern, J., & Keeling, D. (2009). *The widget effect: Our national failure to acknowledge and act on differences in teacher effectiveness*. Retrieved from The New Teacher Project website: http://www.tntp.org
- Wise, A. E., Darling-Hammond, L., Barnett, B., Berliner, D., Haller, E., Praskac, A., & Schlechty, P. (1987). *Effective Teacher Selection From Recruitment to Retention*. Retrieved from the RAND website: http://www.rand.org/pubs/reports/
- Winter, P. A., & Melloy, S. H. (2005). Teacher recruitment in a school reform state: Factors that influence applicant attraction to teaching vacancies. *Educational Administration Quarterly*, *41*(2), 349–372.
- Wright, S. P., Horn, S. P., & Sanders, W. L. (1997). Teacher and classroom context effects on student achievement: Implications for teacher evaluation. *Journal of Personnel Evaluation in Education*, 11, 57–67. Retrieved from SpringerLink Database.
- Young, I. P., & Miller-Smith, K. (2006). Effects of a state mandated policy (site-based councils) and of potential role of incumbents on teacher screening section in high and low performing schools. *Education Policy Analysis Archives*, 14(7), 1–24. Retrieved from epaa.asu.edu

Young, I. P. (2008). *The human resource function in educational administration*. (9th ed.). Upper Saddle River, NJ: Pearson Education Inc.